
UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 10-Q

(MARK ONE)

[X] QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

FOR THE QUARTERLY PERIOD ENDED JUNE 30, 2005

0R

[] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

FOR THE TRANSITION PERIOD FROM ______ TO _____

COMMISSION FILE NUMBER: 000-30111

LEXICON GENETICS INCORPORATED (EXACT NAME OF REGISTRANT AS SPECIFIED IN ITS CHARTER)

DELAWARE (STATE OR OTHER JURISDICTION OF INCORPORATION OR ORGANIZATION) 76-0474169 (I.R.S. EMPLOYER IDENTIFICATION NUMBER)

8800 TECHNOLOGY FOREST PLACE THE WOODLANDS, TEXAS 77381 (ADDRESS OF PRINCIPAL EXECUTIVE OFFICES AND ZIP CODE)

(281) 863-3000 (REGISTRANT'S TELEPHONE NUMBER, INCLUDING AREA CODE)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports) and (2) has been subject to such filing requirements for the past 90 days.

Yes X No

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act.

Yes X No

As of July 26, 2005, 63,735,954 shares of the registrant's common stock, par value \$0.001 per share, were outstanding.

LEXICON GENETICS INCORPORATED

TABLE OF CONTENTS

	PAGE
FACTORS AFFECTING FORWARD-LOOKING STATEMENTS	2
PART I - FINANCIAL INFORMATION	
Item 1. Financial Statements Consolidated Balance Sheets - June 30, 2005 (unaudited) and December 31, 2004 Consolidated Statements of Operations (unaudited) - Three and Six Months Ended	3
June 30, 2005 and 2004	
June 30, 2005 and 2004	5 6
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	10
Item 3. Quantitative and Qualitative Disclosures About Market Risk	20
Item 4. Controls and Procedures	20
PART II - OTHER INFORMATION	
Item 4. Submission of Matters to a Vote of Security Holders Item 6. Exhibits and Reports on Form 8-K	21 21
SIGNATURES	23

The Lexicon name and logo, LexVision(R) and OmniBank(R) are registered trademarks and Genome5000(TM) and e-Biology(TM) are trademarks of Lexicon Genetics Incorporated.

FACTORS AFFECTING FORWARD LOOKING STATEMENTS

This quarterly report on Form 10-Q contains forward-looking statements. These statements relate to future events or our future financial performance. We have attempted to identify forward-looking statements by terminology including "anticipate," "believe," "can," "continue," "could," "estimate," "expect," "intend," "may," "plan," "potential," "predict," "should" or "will" or the negative of these terms or other comparable terminology. These statements are only predictions and involve known and unknown risks, uncertainties and other factors, including the risks outlined under "Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations - Risk Factors," that may cause our or our industry's actual results, levels of activity, performance or achievements to be materially different from any future results, levels or activity, performance or achievements.

Although we believe that the expectations reflected in the forward-looking statements are reasonable, we cannot guarantee future results, levels of activity, performance or achievements. We are not under any duty to update any of the forward-looking statements after the date of this quarterly report on Form 10-Q to conform these statements to actual results, unless required by law.

ITEM 1. FINANCIAL STATEMENTS

LEXICON GENETICS INCORPORATED

CONSOLIDATED BALANCE SHEETS (IN THOUSANDS, EXCEPT PAR VALUE)

AS OF AS OF JUNE 30, DECEMBER 31, 2005 2004 --------- ASSETS (UNAUDITED) ----- Current assets: Cash and cash equivalents \$ 12,490 \$ 14,612 Short-term investments, including restricted investments of \$430 60,302 72,946 Accounts receivable, net of allowance for doubtful accounts of \$75 .. 1,789 5,345 Other receivables -- 1,052 Prepaid expenses and other current assets Total current assets Property and equipment, net of accumulated depreciation of \$44,094 and \$41,892, respectively 86,051 84,573 Goodwill 25,798 25,798 Intangible assets, net of amortization of \$4,760 and \$4,160, respectively 1,240 1,840 Other assets 946 1,021 ----- Total assets \$ 192,149 \$ 211,980 ======= ====== LIABILITIES AND STOCKHOLDERS' EQUITY Current liabilities: Accounts payable \$ 2,802 \$ 7,574 Accrued liabilities Current portion of deferred revenue 23,818 19,500 Current portion of long-term debt 4,721 4,691 ----- Total current liabilities revenue, net of current portion 26,344 18,092 Long-term debt 32,568 32,940 Other long-term liabilities 699 644 ----------- Total liabilities Commitments and contingencies Stockholders' equity: Preferred stock, \$.01 par value; 5,000 shares authorized; no shares issued and outstanding --Common stock, \$.001 par value; 120,000 shares authorized; 63,730 and 63,491 shares issued and outstanding 64 63 Additional paid-in capital Deferred stock compensation deficit (289,223) (261,115) Accumulated other comprehensive loss Total stockholders' equity - ----- Total liabilities and stockholders' equity

The accompanying notes are an integral part of these consolidated financial statements.

CONSOLIDATED STATEMENTS OF OPERATIONS (IN THOUSANDS, EXCEPT PER SHARE AMOUNTS) (UNAUDITED)

THREE MONTHS ENDED JUNE 30, SIX MONTHS ENDED JUNE 30, --------------- 2005 2004 2005 2004 ------- ----- ----- ------Revenues: Collaborative research \$ 13,771 \$ 8,211 \$ 22,654 \$ 16,505 Subscription and license fees 127 2,567 5,169 6,115 - Total revenues 13,898 10,778 27,823 22,620 Operating expenses: Research and development 23,667 22,580 46,427 44,981 General and administrative 4,750 4,642 9,182 9,686 ------ ------ Total operating expenses 28,417 27,222 55,609 54,667 ------- ----- Loss from operations (14,519) (16,444) (27,786) (32,047) Interest income 506 361 997 793 Interest expense (827) (705) (1,632) (996) Other income, net (2) -- 313 (4) -------- ---- Net loss \$(14,842) \$(16,788) \$(28,108) \$(32,254) ====== ======== ============= Net loss per common share, basic and diluted \$ (0.23) \$ (0.26) \$ (0.44) \$ (0.51)Shares used in computing net loss per common share, basic and diluted 63,636 63,369 63,581 63,217

The accompanying notes are an integral part of these consolidated financial statements.

LEXICON GENETICS INCORPORATED

CONSOLIDATED STATEMENTS OF CASH FLOWS (IN THOUSANDS) (UNALIDITED)

(UNAUDITED) SIX MONTHS ENDED JUNE 30, ----- 2005 2004 ----- Cash flows from operating activities: Net loss \$ (28,108) \$ (32,254) Adjustments to reconcile net loss to net cash used in operating activities: Depreciation 5,207 5,430 Amortization of intangible assets, other than goodwill 600 600 Amortization of deferred stock compensation (20) 828 Loss on disposal of property and equipment 10 -- Changes in operating assets and liabilities: Decrease in accounts receivable 4,608 4,393 Decrease in prepaid expenses and other current assets 1,260 368 (Increase) decrease in other assets 75 (898) Decrease in accounts payable and other liabilities (4,367) (1,333) Increase (decrease) in deferred revenue 12,570 (10,914) -------- Net cash used in operating activities (8,165) (33,780) Cash flows from investing activities: Purchases of property and equipment (6,780) (3,579) Proceeds from disposal of property and equipment 85 15 Decrease in restricted cash -- 14,372 Purchases of investments (118,354) Maturities of investments ----- Net cash provided by investing activities 5,928 30,888 Cash flows from financing activities: Proceeds from issuance of common stock 457 1,511 Proceeds from debt borrowings -- 34,000 Repayment of debt borrowings (342) (52,392) Repayment of other long-term liabilities -- (2,466) ------Net cash provided by (used in) financing activities 115 (19,347) ------ Net decrease in cash and cash equivalents (2,122) (22,239) Cash and cash equivalents at beginning of period 14,612 35,856 ---------- Cash and cash equivalents at end of period \$ 12,490 \$ 13,617 ======== ======= Supplemental disclosure of cash flow information: Cash paid for interest \$ 1,395 \$ 567 Supplemental disclosure of non-cash investing and financing activities: Unrealized loss on investments \$ (21) \$ -- Reversal of deferred stock compensation, in connection with stock options\$ 35 \$ 19 Retirement of property and equipment \$ 3,100 \$ 283

The accompanying notes are an integral part of these consolidated financial statements.

LEXICON GENETICS INCORPORATED

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

1. BASIS OF PRESENTATION

The accompanying unaudited consolidated financial statements of Lexicon Genetics Incorporated (Lexicon or the Company) have been prepared in accordance with generally accepted accounting principles for interim financial information and pursuant to the rules and regulations of the Securities and Exchange Commission (SEC). Accordingly, they do not include all of the information and footnotes required by generally accepted accounting principles for complete financial statements.

In the opinion of management, all adjustments (consisting of normal recurring adjustments) considered necessary for a fair presentation have been included. Operating results for the six-month period ended June 30, 2005 are not necessarily indicative of the results that may be expected for the year ended December 31, 2005.

The accompanying consolidated financial statements include the accounts of Lexicon and its subsidiaries. Intercompany transactions and balances are eliminated in consolidation.

For further information, refer to the financial statements and footnotes thereto included in Lexicon's annual report on Form 10-K for the year ended December 31, 2004, as filed with the SEC.

2. RECLASSIFICATION

As of June 30, 2004 and December 31, 2003, Lexicon reclassified auction rate securities of \$53.5 million and \$46.1 million, respectively, from cash equivalents to short-term investments and, as of December 31, 2003, Lexicon reclassified \$42.6 million from restricted cash to short-term investments. The accompanying consolidated statement of cash flows for the six months ended June 30, 2004 has been adjusted to reflect these reclassifications.

3. COMPREHENSIVE LOSS

Comprehensive loss is comprised of net loss and unrealized gains and losses on short-term investments, which are considered available-for-sale securities. Comprehensive loss for the three months ended June 30, 2005 was \$14.8 million, which includes a net loss of \$14.8 million and a \$12,000 unrealized gain on short-term investments. Comprehensive loss for the six months ended June 30, 2005 was \$28.1 million, which includes a net loss of \$28.1 million and a \$21,000 unrealized loss on short-term investments.

4. NET LOSS PER SHARE

Net loss per share is computed using the weighted average number of shares of common stock outstanding during the applicable period. Shares associated with stock options and warrants are not included because they are antidilutive. There are no differences between basic and diluted net loss per share for all periods presented.

5. STOCK-BASED COMPENSATION

Lexicon's stock-based compensation plans are accounted for under the recognition and measurement provisions of Accounting Principles Board (APB) Opinion No. 25, "Accounting for Stock Issued to Employees, and Related Interpretations." Under the intrinsic value method described in APB Opinion No. 25, no compensation expense is recognized if the exercise price of the employee stock option equals the market price of the underlying stock on the date of grant. Lexicon recognized stock-based compensation expense of \$0.8 million for the six months ended June 30, 2004, primarily relating to option grants made prior to Lexicon's April 2000 initial public offering. All deferred stock compensation relating to these options was fully amortized as of January 31, 2004 when these options became fully vested.

The following table illustrates the effect on net loss and net loss per share if the fair value recognition provisions of Financial Accounting Standards Board (FASB) Statement of Financial Accounting Standards (SFAS) No. 123 "Accounting for Stock Based Compensation," had been applied to all outstanding and unvested awards in each period:

THREE MONTHS ENDED SIX MONTHS ENDED JUNE 30, JUNE 30,						
- 2005 2004 2005 2004 Net loss, as reported:						
<pre>\$(14,842) \$(16,788) \$(28,108) \$(32,254) Add: Stock-based employee compensation expense included in reported net loss (9) (20) 827 Deduct: Total stock-based employee compensation expense determined under fair value based method for all awards</pre>						
Pro forma net loss						
\$(17,978) \$(20,654) \$(34,385) \$(40,175) ====================================						
\$ (0.23) \$ (0.26) \$ (0.44) \$ (0.51) ======= Pro forma						
\$ (0.28) \$ (0.33) \$ (0.54) \$ (0.64) ======== ===========================						

6. DEBT OBLIGATIONS

Genentech Loan: On December 31, 2002, Lexicon borrowed \$4.0 million under a note agreement with Genentech, Inc. The proceeds of the loan are to be used to fund research efforts under the alliance agreement with Genentech. The note matures on December 31, 2005, but the Company may prepay it at any time. The Company may repay the note, at its option, in cash, in shares of common stock valued at the then-current market price, or in a combination of cash and shares, subject to certain limitations. The note accrues interest at an annual rate of 8%, compounded quarterly.

Mortgage Loan: In April 2004, Lexicon purchased its facilities in The Woodlands, Texas that were previously subject to a synthetic lease. The Company repaid the \$54.8 million funded under the synthetic lease with proceeds from a \$34.0 million third-party mortgage financing and \$20.8 million in cash. The mortgage loan has a ten-year term with a 20-year amortization and bears interest at a fixed rate of 8.23%. As a result of the refinancing, all restrictions on the cash and investments that had secured the obligations under the synthetic lease were eliminated.

7. COMMITMENTS AND CONTINGENCIES

In May 2002, Lexicon's subsidiary Lexicon Pharmaceuticals (New Jersey), Inc. leased a 76,000 square-foot laboratory and office space in Hopewell, New Jersey under an agreement which expires in June 2013. The lease provides for an escalating yearly rent payment of \$1.3 million in the first year,

\$2.1 million in years two and three, \$2.2 million in years four to six, \$2.3 million in years seven to nine and \$2.4 million in years ten and eleven. Lexicon is the guarantor of the obligations of its subsidiary under the lease. The Company is required to maintain restricted investments to collateralize the Hopewell lease. As of June 30, 2005, the Company had \$430,000 in restricted investments to collateralize a standby letter of credit for this lease.

8. NEW COLLABORATION AGREEMENT

Lexicon formed a collaboration with N.V. Organon (Organon) in May 2005 to jointly discover, develop and commercialize novel biotherapeutics. In the collaboration, Lexicon will create and analyze mouse knockouts for up to 300 genes selected by the parties that encode secreted proteins or potential antibody targets, including two of Lexicon's existing drug discovery programs. The parties will jointly select targets for further research and development and will equally share costs and responsibility for research, preclinical and clinical activities. The parties will jointly determine the manner in which collaboration products will be commercialized and will equally benefit from product revenue. If fewer than five development candidates are designated under the collaboration, Lexicon's share of costs and product revenue will be proportionally reduced. Lexicon will receive a milestone payment for each development candidate in excess of five. Either party may decline to participate in further research or development efforts with respect to a collaboration product, in which case such party will receive royalty payments on sales of such collaboration product rather than sharing in revenue. Organon will have principal responsibility for manufacturing biotherapeutic products resulting from the collaboration for use in clinical trials and for worldwide sales.

Lexicon received an upfront payment of \$22.5 million from Organon in exchange for access to Lexicon's drug target discovery capabilities and the exclusive right to co-develop biotherapeutic products that modulate the 300 genes selected for the collaboration. This upfront payment will be recognized as revenue over the four-year target function discovery portion of the alliance. Organon will also provide Lexicon with annual research funding totaling up to \$50 million for its 50% share of the collaboration's costs during this same period.

9. SUBSEQUENT EVENT

In July 2005, Lexicon was awarded \$35 million from the Texas Enterprise Fund for the creation of a knockout mouse embryonic stem cell library containing 350,000 cell lines using Lexicon's proprietary gene trapping technology. Lexicon will create the library for the Texas Institute for Genomic Medicine (TIGM), a newly formed non-profit institute whose founding members are Texas A&M University, the Texas A&M University System Health Science Center and Lexicon. TIGM researchers may also access specific cells from Lexicon's current gene trap library of 270,000 mouse embryonic stem cell lines and will have certain rights to utilize Lexicon's patented gene targeting technologies. In addition, Lexicon will equip TIGM with the bioinformatics software required for the management and analysis of data relating to the library. The Texas Enterprise Fund has also awarded \$15 million to the Texas A&M University System for the creation of facilities and infrastructure to house the library.

Under the terms of the award, Lexicon is responsible for the creation of a specified number of jobs, reaching an aggregate of 1,616 new jobs in Texas by December 31, 2015. Lexicon will obtain credits based on funding received by TIGM and certain related parties from sources other than the State of Texas that it may offset against its potential liability for any job creation shortfalls. Lexicon will also obtain credits against future jobs commitment liabilities for any surplus jobs it creates. Subject to these credits, if Lexicon fails to create the specified number of jobs, the state may require Lexicon to repay \$2,415 for each job Lexicon fails short. Lexicon's maximum aggregate exposure for such payments, if Lexicon fails to create any new jobs, is approximately \$14.4 million, without giving effect to any credits

to which Lexicon may be entitled. The Texas A&M University System, together with TIGM, has independent job creation obligations and is obligated for an additional period to maintain an aggregate of 5,000 jobs, inclusive of those Lexicon creates.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

OVERVIEW

We are a biopharmaceutical company focused on discovering and developing breakthrough treatments for human disease. We are using gene knockout technology to systematically discover the physiological functions of genes in living mammals, or in vivo. We generate our gene function discoveries using knockout mice - mice whose DNA has been altered to disrupt, or "knock out," the function of the altered gene. Our patented gene trapping and gene targeting technologies enable us to rapidly generate these knockout mice by altering the DNA of genes in a special variety of mouse cells, called embryonic stem cells, which can be cloned and used to generate mice with the altered gene. We employ an integrated platform of advanced medical technologies to systematically discover and validate which genes, when knocked out, result in a favorable medical profile with pharmaceutical utility. We then pursue those genes and the proteins they encode as potential targets for therapeutic intervention in our drug discovery programs.

We employ internal resources and drug discovery alliances to discover potential small molecule, antibody and protein drugs for in vivo-validated drug targets that we consider to have high pharmaceutical value. We use our own sophisticated libraries of drug-like chemical compounds and an industrialized medicinal chemistry platform to identify small molecule drug candidates for our in vivo-validated drug targets. We have established alliances with Bristol-Myers Squibb Company to discover and develop novel small molecule drugs in the neuroscience field; with Genentech, Inc. to discover therapeutic proteins and antibody targets; with N.V. Organon to discover, develop and commercialize novel biotherapeutics; and with Takeda Pharmaceutical Company Limited to discover new drugs for the treatment of high blood pressure. In addition, we have established collaborations and license agreements with many other leading pharmaceutical and biotechnology companies under which we receive fees and, in some cases, are eligible to receive milestone and royalty payments, for access to some of our technologies and discoveries for use in their own drug discovery efforts.

We derive substantially all of our revenues from drug discovery alliances, target validation collaborations for the development and, in some cases, analysis of the physiological effects of genes altered in knockout mice and technology licenses. To date, we have generated a substantial portion of our revenues from a limited number of sources.

Our operating results and, in particular, our ability to generate additional revenues are dependent on many factors, including our success in establishing research collaborations and technology licenses, expirations of our research collaborations, the success rate of our discovery efforts leading to opportunities for new research collaborations and licenses, as well as milestone payments and royalties, the timing and willingness of collaborators to commercialize products which may result in royalties, and general and industry-specific economic conditions which may affect research and development expenditures. Our future revenues from collaborations and alliances are uncertain because our existing agreements have fixed terms or relate to specific projects of limited duration. Our future revenues from technology licenses are uncertain because they depend, in large part, on securing new agreements. Subject to limited exceptions, we do not intend to offer subscriptions to our databases or make our compound libraries available for purchase in the future. Our ability to secure future revenue-generating agreements will depend upon our ability to address the needs of our potential future collaborators and licensees, and to negotiate agreements that we believe are in our long-term best interests. We may determine that our interests are better served by retaining rights to our discoveries and advancing our therapeutic programs to a later stage, which could limit our near-term revenues. Because of these and other factors, our operating results have fluctuated in the past and are likely to do so in the future, and we

do not believe that period-to-period comparisons of our operating results are a good indication of our future performance.

Since our inception, we have incurred significant losses and, as of June 30, 2005, we had an accumulated deficit of \$289.2 million. Our losses have resulted principally from costs incurred in research and development, general and administrative costs associated with our operations, and non-cash stock-based compensation expenses associated with stock options granted to employees and consultants prior to our April 2000 initial public offering. Research and development expenses consist primarily of salaries and related personnel costs, material costs, facility costs, depreciation on property and equipment, legal expenses resulting from intellectual property prosecution and other expenses related to our drug discovery and Genome5000 programs, the development and analysis of knockout mice and our other target validation research efforts, and the development of compound libraries. General and administrative expenses consist primarily of salaries and related expenses for executive, finance and other administrative personnel, professional fees and other corporate expenses, including information technology, facilities costs and general legal activities. In connection with the expansion of our drug discovery programs and our target validation research efforts, we expect to incur increasing research and development and general and administrative costs. As a result, we will need to generate significantly higher revenues to achieve profitability.

CRITICAL ACCOUNTING POLICIES

Revenue Recognition

We recognize revenues when persuasive evidence of an arrangement exists, delivery has occurred or services have been rendered, the price is fixed and determinable, and collectibility is reasonably assured. Payments received in advance under these arrangements are recorded as deferred revenue until earned.

Upfront fees and annual research funding under our drug discovery alliances are recognized as revenue on a straight-line basis over the estimated period of service, generally the contractual research term, to the extent they are non-refundable. Milestone-based fees are recognized upon completion of specified milestones according to contract terms. Fees for access to our databases and other target validation resources are recognized ratably over the subscription or access period. Payments received under target validation collaborations and government grants are recognized as revenue as we perform our obligations related to such research to the extent such fees are non-refundable. Non-refundable technology license fees are recognized as revenue upon the grant of the license, when performance is complete and there is no continuing involvement.

Revenues recognized from multiple element contracts are allocated to each element of the arrangement based on the relative fair value of the elements. The determination of fair value of each element is based on objective evidence. When revenues for an element are specifically tied to a separate earnings process, revenue is recognized when the specific performance obligation associated with the element is completed. When revenues for an element are not specifically tied to a separate earnings process, they are recognized ratably over the term of the agreement.

A change in our revenue recognition policy or changes in the terms of contracts under which we recognize revenues could have an impact on the amount and timing of our recognition of revenues.

Research and Development Expenses

Research and development expenses consist of costs incurred for company-sponsored as well as collaborative research and development activities. These costs include direct and research-related overhead expenses and are expensed as incurred. Patent costs and technology license fees for technologies that are utilized in research and development and have no alternative future use are expensed when incurred.

Prior to preclinical development work, we are unable to segregate the costs related to research performed on drug candidates because the drug candidate is often not specifically identified until the later stages of our research. With the commencement of formal preclinical development in 2005, we will account on a program-by-program basis for the costs related to the development of the identified drug products.

Goodwill Impairment

Goodwill is not amortized, but is tested at least annually for impairment at the reporting unit level. We have determined that the reporting unit is the single operating segment disclosed in our current financial statements. Impairment is the condition that exists when the carrying amount of goodwill exceeds its implied fair value. The first step in the impairment process is to determine the fair value of the reporting unit and then compare it to the carrying value, including goodwill. We determined that the market capitalization approach is the most appropriate method of measuring fair value of the reporting unit. Under this approach, fair value is calculated as the average closing price of our common stock for the 30 days preceding the date that the annual impairment test is performed, multiplied by the number of outstanding shares on that date. A control premium, which is representative of premiums paid in the marketplace to acquire a controlling interest in a company, is then added to the market capitalization to determine the fair value of the reporting unit. If the fair value exceeds the carrying value, no further action is required and no impairment loss is recognized. Additional impairment assessments may be performed on an interim basis if we encounter events or changes in circumstances that would indicate that, more likely than not, the carrying value of goodwill has been impaired.

RESULTS OF OPERATIONS

Three Months Ended June 30, 2005 and 2004

Revenues. Total revenues and dollar and percentage changes as compared to the corresponding period in the prior year are as follows (dollar amounts are presented in millions):

 Collaborative research - Revenue from collaborative research increased 68% to \$13.8 million primarily due to our recognition of revenues under our biotherapeutics collaboration with Organon, which was entered into in May 2005, and our hypertension drug discovery alliance with Takeda, which was entered into in July 2004. This was offset in part by a decrease in revenues from the termination of our therapeutic protein discovery alliance with Incyte Corporation in June 2004.

 Subscription and license fees - Revenue from subscriptions and license fees decreased 95% to \$0.1 million primarily as a result of the termination in June 2004 and December 2004, respectively, of our LexVision(R) database subscription programs with Incyte and Bristol-Myers Squibb.

Research and Development Expenses. Research and development expenses and dollar and percentage changes as compared to the corresponding period in the prior year are as follows (dollar amounts are presented in millions):

Research and development expenses consist primarily of salaries and other personnel-related expenses, laboratory supplies, facility and equipment costs, third-party and other services. The change in the three months ended June 30, 2005 as compared to the corresponding period in 2004 resulted primarily from the following costs:

- o Personnel Personnel costs increased 8% to \$11.8 million primarily due to increased personnel to support the expansion of our drug discovery programs and merit-based pay increases for employees. Salaries, bonuses, employee benefits, payroll taxes, and recruiting and relocation costs are included in personnel costs.
- Laboratory supplies Laboratory supplies expense decreased 6% to \$3.2 million due primarily to fewer purchases of specialty reagents.
- Facilities and equipment Facilities and equipment costs increased 5% to \$5.3 million due primarily to higher utility costs.
- o Third-party services Costs associated with third-party services decreased 3% to \$2.0 million primarily due to the termination in June 2004 of our LifeSeq(R) Gold database subscription, offset in part by an increase in third-party contract research costs. Costs associated with third-party services include third-party contract research, subscriptions to third-party databases, technology licenses, and legal and patent fees.
- o Other Other costs increased by 21% to \$1.4 million primarily related to increased information technology and insurance costs.

General and Administrative Expenses. General and administrative expenses and dollar and percentage changes as compared to the corresponding period in the prior year are as follows (dollar amounts are presented in millions):

General and administrative expenses consist primarily of personnel costs to support our research activities, facility and equipment costs and professional fees, such as legal fees. The change in the three months ended June 30, 2005 as compared to the corresponding period in 2004 resulted primarily from the following costs:

- Personnel Personnel costs increased 3% to \$2.8 million.
 Salaries, bonuses, employee benefits, payroll taxes, recruiting and relocation costs are included in personnel costs.
- Facilities and equipment Facilities and equipment costs decreased 2% to \$0.7 million.
- Professional fees Professional fees decreased 3% to \$0.6 million.
- o Other Other costs increased 9% to \$0.7 million.

Interest Income. Interest income increased 40% to \$0.5 million in the three months ended June 30, 2005 from \$0.4 million in the corresponding period in 2004 due to higher interest rates, offset in part by lower average cash and investment balances.

Interest Expense. Interest expense increased 17% to \$0.8 million in the three months ended June 30, 2005 from \$0.7 million in the corresponding period in 2004. The increase was attributable to interest expense on the \$34.0 million mortgage loan on our facilities in The Woodlands, Texas, which was entered into in April 2004.

Net Loss and Net Loss Per Common Share. Net loss decreased 12% to \$14.8 million in the three months ended June 30, 2005 from \$16.8 million in the corresponding period in 2004. Net loss per common share decreased to \$0.23 in the three months ended June 30, 2005 from \$0.26 in the corresponding period in 2004.

Our quarterly operating results have fluctuated in the past and are likely to do so in the future, and we believe that quarter-to-quarter comparisons of our operating results are not a good indication of our future performance.

Six Months Ended June 30, 2005 and 2004

Revenues. Total revenues and dollar and percentage changes as compared to the corresponding period in the prior year are as follows (dollar amounts are presented in millions):

- O Collaborative research Revenue from collaborative research increased 37% to \$22.7 million primarily due to our recognition of revenues under our biotherapeutics collaboration with Organon, which was entered into in May 2005, and our hypertension drug discovery alliance with Takeda, which was entered into in July 2004. This was offset in part by a decrease in revenues from the termination of our therapeutic protein discovery alliance with Incyte in June 2004.
- o Subscription and license fees Revenue from subscriptions and license fees decreased 15% to \$5.2 million primarily as a result of the termination in June 2004 and December 2004, respectively, of our LexVision(R) database subscription programs with Incyte and Bristol-Myers Squibb. The reduction was offset in part by technology license fees received from Deltagen, Inc. in connection with the settlement of Lexicon's claim in Deltagen's bankruptcy proceedings.

Research and Development Expenses. Research and development expenses and dollar and percentage changes as compared to the corresponding period in the prior year are as follows (dollar amounts are presented in millions):

Research and development expenses consist primarily of salaries and other personnel-related expenses, laboratory supplies, facility and equipment costs, third-party and other services. The change in the six months ended June 30, 2005 as compared to the corresponding period in 2004 resulted primarily from the following costs:

- Personnel Personnel costs increased 10% to \$23.5 million primarily due to increased personnel to support the expansion of our drug discovery programs and merit-based pay increases for employees.
- Laboratory supplies Laboratory supplies expense decreased 7% to \$6.4 million due primarily to the bulk purchase of certain supplies in the prior year period.
- Facilities and equipment Facilities and equipment costs increased 3% to \$10.4 million.
- Third-party services Costs associated with third-party services decreased 10% to \$3.4 million primarily due to the termination in June 2004 of our LifeSeq Gold database subscription, offset in part by an increase in third-party contract research costs.
- o Other Other costs increased by 16% to \$2.8 million primarily related to increased information technology costs.

General and Administrative Expenses. General and administrative expenses and dollar and percentage changes as compared to the corresponding period in the prior year are as follows (dollar amounts are presented in millions):

General and administrative expenses consist primarily of personnel costs to support our research activities, facility and equipment costs and professional fees, such as legal fees. The change in the six months ended June 30, 2005 as compared to the corresponding period in 2004 resulted primarily from the following costs:

- o Personnel Personnel costs decreased 2% to \$5.5 million.
- Facilities and equipment Facilities and equipment costs decreased 4% to \$1.5 million.
- Professional fees Professional fees increased 7% to \$1.0 million primarily due to increased consulting fees.

o Other - Other costs remained unchanged at \$1.2 million in both of the six-month periods ended June 30, 2005 and 2004.

Interest Income. Interest income increased 26% to \$1.0 million in the six months ended June 30, 2005 from \$0.8 million in the corresponding period in 2004 primarily due to higher interest rates, offset by lower average cash and investment balances.

Interest Expense. Interest expense increased to \$1.6 million in the six months ended June 30, 2005 from \$1.0 million in the corresponding period in 2004. The increase was attributable to interest expense on the \$34.0 million mortgage loan on our facilities in The Woodlands, Texas, which was entered into in April 2004.

Net Loss and Net Loss Per Common Share. Net loss decreased 13% to \$28.1 million in the six months ended June 30, 2005 from \$32.3 million in the corresponding period in 2004. Net loss per common share decreased to \$0.44 in the six months ended June 30, 2005 from \$0.51 in the corresponding period in 2004. Net loss includes stock-based compensation expense of \$0.8 million in the six months ended June 30, 2004.

LIQUIDITY AND CAPITAL RESOURCES

We have financed our operations from inception primarily through sales of common and preferred stock, contract and milestone payments to us under our drug discovery alliance, target validation, database subscription and license agreements, equipment financing arrangements and leasing arrangements. From our inception through June 30, 2005, we had received net proceeds of \$295.3 million from issuances of common and preferred stock, including \$203.2 million of net proceeds from the initial public offering of our common stock in April 2000 and \$50.1 million from our July 2003 common stock offering. In addition, from our inception through June 30, 2005, we received \$269.0 million in cash payments from drug discovery alliances, target validation collaborations, database subscription and technology license fees, sales of compound libraries and reagents, and government grants, of which \$220.9 million had been recognized as revenues through June 30, 2005.

As of June 30, 2005, we had \$72.8 million in cash, cash equivalents and short-term investments (including \$0.4 million of restricted investments), as compared to \$87.6 million (including \$0.4 million of restricted investments) as of December 31, 2004. We used cash of \$8.2 million in operations in the six months ended June 30, 2005. This consisted primarily of the net loss for the period of \$28.1 million offset by non-cash charges of \$5.2 million related to depreciation expense and \$0.6 million related to amortization of intangible assets other than goodwill; a \$12.6 million increase in deferred revenue; and changes in other operating assets and liabilities of \$1.5 million. Investing activities provided cash of \$5.9 million in the six months ended June 30, 2005, primarily due to net maturities of short-term investments of \$12.6 million. This was offset by purchases of property and equipment of \$6.8 million. Financing activities provided cash of \$0.1 million.

In April 2004, we purchased our facilities in The Woodlands, Texas from the lessor under our previous synthetic lease agreement. In connection with such purchase, we repaid the \$54.8 million funded under the synthetic lease with proceeds from a \$34.0 million third-party mortgage financing and \$20.8 million in cash. The mortgage loan has a ten-year term with a 20-year amortization and bears interest at a fixed rate of 8.23%. As a result of the refinancing, all restrictions on the cash and investments that had secured our obligations under the synthetic lease were eliminated.

In May 2002, our subsidiary Lexicon Pharmaceuticals (New Jersey), Inc. signed a ten-year lease for a 76,000 square-foot facility in Hopewell, New Jersey. The term of the lease extends until June 30,

2013. The lease provides for an escalating yearly base rent payment of \$1.3 million in the first year, \$2.1 million in years two and three, \$2.2 million in years four to six, \$2.3 million in years seven to nine and \$2.4 million in years ten and eleven. We are the guarantor of the obligations of our subsidiary under the lease.

In December 2002, we borrowed \$4.0 million under a note agreement with Genentech. The proceeds of the loan are to be used to fund research efforts under our alliance with Genentech for the discovery of therapeutic proteins and antibody targets. The note matures on or before December 31, 2005, but we may prepay it at any time. We may repay the note, at our option, in cash, in shares of our common stock valued at the then-current market value, or in a combination of cash and shares, subject to certain limitations. The note accrues interest at an annual rate of 8%, compounded quarterly.

In May 2005, we formed a collaboration with N.V. Organon to jointly discover, develop and commercialize novel biotherapeutics. We and Organon will equally share costs and responsibility for research, preclinical and clinical activities and equally benefit from product revenue. If fewer than five development candidates are designated under the collaboration, our share of costs and product revenue will be proportionally reduced. We will receive a milestone payment for each development candidate in excess of five. Either party may decline to participate in further research or development efforts with respect to a collaboration product, in which case such party will receive royalty payments on sales of such collaboration product rather than sharing in revenue. We received an upfront payment of \$22.5 million and Organon will also provide us with annual research funding totaling up to \$50 million for its 50% share of the collaboration's costs during the four-year target function discovery portion of the alliance.

Subsequent to June 30, 2005, we were awarded \$35 million from the Texas Enterprise Fund for the creation of a knockout mouse embryonic stem cell library containing 350,000 cell lines. We will create the library for the Texas Institute for Genomic Medicine, a newly formed non-profit institute whose founding members are Texas A&M University, the Texas A&M University System Health Science Center and us. Under the terms of the award, we are responsible for the creation of a specified number of jobs, reaching an aggregate of 1,616 new jobs in Texas by December 31, 2015. We will obtain credits based on funding received by the institute and certain related parties from sources other than the State of Texas that we may offset against our potential liability for any job creation shortfalls. We will also obtain credits against future jobs commitment liabilities for any surplus jobs we create. Subject to these credits, if we fail to create the specified number of jobs, the state may require us to repay \$2,415 for each job we fall short. Our maximum aggregate exposure for such payments, if we fail to create any new jobs, is approximately \$14.4 million, without giving effect to any credits to which we may be entitled.

Our future capital requirements will be substantial and will depend on many factors, including our ability to obtain alliance, collaboration and technology license agreements, the amount and timing of payments under such agreements, the level and timing of our research and development expenditures, market acceptance of our products, the resources we devote to developing and supporting our products and other factors. Our capital requirements will also be affected by any expenditures we make in connection with license agreements and acquisitions of and investments in complementary technologies and businesses. We expect to devote substantial capital resources to continue our research and development efforts, to expand our support and product development activities, and for other general corporate activities. We believe that our current unrestricted cash and investment balances and revenues we expect to derive from drug discovery alliances, target validation collaborations and technology licenses will be sufficient to fund our operations at least through the next two years. During or after this period, if cash generated by operations is insufficient to satisfy our liquidity requirements, we will need to sell additional equity or debt securities or obtain additional credit arrangements. Additional financing may not be available on terms acceptable to us or at all. The sale of additional equity or convertible debt securities may result in additional dilution to our stockholders.

DISCLOSURE ABOUT MARKET RISK

We are exposed to limited market and credit risk on our cash equivalents, which have maturities of three months or less at the time of purchase. We maintain a short-term investment portfolio which consists of U.S. government agency debt obligations, investment grade commercial paper, corporate debt securities and certificates of deposit that mature within twelve months and auction rate securities that mature greater than twelve months from the time of purchase, which we believe are subject to limited market and credit risk. We currently do not hedge interest rate exposure or hold any derivative financial instruments in our investment portfolio.

We have operated primarily in the United States and substantially all sales to date have been made in U.S. dollars. Accordingly, we have not had any material exposure to foreign currency rate fluctuations.

RISK FACTORS

Our business is subject to certain risks and uncertainties, including those referenced below:

Risks Related to Our Company and Business

- we have a history of net losses, and we expect to continue to incur net losses and may not achieve or maintain profitability
- o we will need additional capital in the future and, if it is not available, we will have to curtail or cease operations
- o any sale of additional equity securities in the future may be dilutive to our stockholders
- o we are an early-stage company, and we may not successfully develop or commercialize any therapeutics or drug targets that we have identified
- o we face substantial competition in the discovery of the DNA sequences of genes and their functions and in our drug discovery and product development efforts
- o we rely heavily on our collaborators to develop and commercialize pharmaceutical products based on genes that we identify as promising candidates for development as drug targets, and our collaborators' efforts may fail to yield pharmaceutical products on a timely basis, if at all
- we rely on several key collaborators for a significant portion of our revenues, the loss of any of which would negatively impact our business to the extent such losses are not offset by additional collaborators
- cancellations by or conflicts with our collaborators could harm our business
- we may be unsuccessful in developing and commercializing pharmaceutical products on our own
- we lack the capability to manufacture materials for preclinical studies, clinical trials or commercial sales and will rely on third parties to manufacture our potential products, which may harm or delay our product development and commercialization efforts
- o we may engage in future acquisitions, which may be expensive and time consuming and from which we may not realize anticipated benefits

- o if we lose our key personnel or are unable to attract and retain additional personnel, we may be unable to pursue collaborations or develop our own products
- any contamination among our knockout mouse population could negatively affect the reliability of our scientific research or cause us to incur significant remedial costs
- o because all of our target validation operations are located at a single facility, the occurrence of a disaster could significantly disrupt our business
- o our operating results have been and likely will continue to fluctuate, and we believe that period-to-period comparisons of our operating results are not a good indication of our future performance

Risks Related to Our Industry

- o our ability to patent our inventions is uncertain because patent laws and their interpretation are highly uncertain and subject to change
- o our patent applications may not result in enforceable patent rights and, as a result, the protection afforded to our scientific discoveries may be insufficient
- o if other companies and institutions obtain patents relating to our drug target or product candidate discoveries, we may be unable to obtain patents for our inventions based upon those discoveries and may be blocked from using or developing some of our technologies and products
- o issued or pending patents may not fully protect our discoveries, and our competitors may be able to commercialize technologies or products similar to those covered by our issued or pending patents
- o we may be involved in patent litigation and other disputes regarding intellectual property rights and may require licenses from third parties for our discovery and development and planned commercialization activities, and we may not prevail in any such litigation or other dispute or be able to obtain required licenses
- o we use intellectual property that we license from third parties, and if we do not comply with these licenses, we could lose our rights under them
- o we have not sought patent protection outside of the United States for some of our inventions, and some of our licensed patents only provide coverage in the United States, and as a result, our international competitors could be granted foreign patent protection with respect to our discoveries
- o we may be unable to protect our trade secrets
- o our efforts to discover, evaluate and validate potential targets for drug intervention and our drug discovery programs are subject to evolving data and other risks inherent in the drug discovery process
- o our industry is subject to extensive and uncertain government regulatory requirements, which could significantly hinder our ability, or the ability of our collaborators, to obtain, in a timely manner or at all, government approval of products based on genes that we identify, or to commercialize such products

- o if our potential products receive regulatory approval, we or our collaborators will remain subject to extensive and rigorous ongoing regulation
- o the uncertainty of pharmaceutical pricing and reimbursement may decrease the commercial potential of any products that we or our collaborators may develop and affect our ability to raise capital
- we use hazardous chemicals and radioactive and biological materials in our business; any disputes relating to improper handling, storage or disposal of these materials could be time consuming and costly
- o we may be sued for product liability
- o public perception of ethical and social issues may limit or discourage the use of our technologies, which could reduce our revenues

For additional discussion of the risks and uncertainties that affect our business, see "Item 1. Business - Risk Factors" included in our annual report on Form 10-K for the year ended December 31, 2004, as filed with the Securities and Exchange Commission.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

See "Disclosure about Market Risk" under "Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations" for quantitative and qualitative disclosures about market risk.

ITEM 4. CONTROLS AND PROCEDURES

Our chief executive officer and chief financial officer have concluded that our disclosure controls and procedures (as defined in rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934 (the "Exchange Act")) are sufficiently effective to ensure that the information required to be disclosed by us in the reports we file under the Exchange Act is gathered, analyzed and disclosed with adequate timeliness, accuracy and completeness, based on an evaluation of such controls and procedures as of the end of the period covered by this report.

Subsequent to our evaluation, there were no significant changes in internal controls or other factors that could significantly affect internal controls, including any corrective actions with regard to significant deficiencies and material weaknesses.

PART II OTHER INFORMATION

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

Our annual meeting of stockholders was held on April 27, 2005 to consider and vote on the following proposals:

(1) The following individuals were nominated and elected as Class II directors, with the following numbers of shares voted for and withheld for such directors:

NAME OF	
NAME OF	
DIRECTOR	
FOR	
WITHHELD -	
Samuel L.	
Barker,	
Ph.D.	
46,745,034	
3,638,782	
Patricia	
Μ.	
Cloherty	
46,726,024	
3,657,792	
(2)	The following additional matters were considered and approved,
(2)	The fortowing additional matters were considered and approved,
	with the following numbers of shares voted for, voted against
	and abstaining with respect to such matters:
MATTER FOR	
MATTER FOR	
AGAINST	
ABSTAIN	
-	
Ratification	
and	
approval of	
an	
amendment	
to our 2000	
Non-	
Employee	
Directors'	
Stock	
Option Plan	
increasing	
increasing	
increasing the	
increasing the 25,965,492	
increasing the 25,965,492 11,723,453	
increasing the 25,965,492 11,723,453	
increasing the 25,965,492 11,723,453 118,392	
increasing the 25,965,492 11,723,453 118,392 number of	
increasing the 25,965,492 11,723,453 118,392 number of shares of	
increasing the 25,965,492 11,723,453 118,392 number of shares of common	
increasing the 25,965,492 11,723,453 118,392 number of shares of	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares</pre>	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares Ratification	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares Ratification and	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares Ratification	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares Ratification and	
increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares Ratification and approval of the	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares Ratification and approval of the appointment</pre>	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares Ratification and approval of the appointment of Ernst &</pre>	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares Ratification and approval of the appointment of Ernst & 47,979,031</pre>	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares Ratification and approval of the appointment of Ernst & 47,979,031</pre>	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares Ratification and approval of the appointment of Ernst & 47,979,031 2,337,460</pre>	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares to 10,000 shares Ratification and approval of the appointment of Ernst & 47,979,031 2,337,460 67,325</pre>	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares to 10,000 shares Ratification and approval of the appointment of Ernst & 47,979,031 2,337,460 67,325 Young LLP</pre>	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares Ratification and approval of the appointment of Ernst & 47,979,031 2,337,460 67,325 Young LLP as our</pre>	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares Ratification and approval of the appointment of Ernst & 47,979,031 2,337,460 67,325 Young LLP as our</pre>	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares to 10,000 shares Ratification and approval of the appointment of Ernst & 47,979,031 2,337,460 67,325 Young LLP as our independent</pre>	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares to 10,000 shares Ratification and approval of the appointment of Ernst & 47,979,031 2,337,460 67,325 Young LLP as our independent auditors</pre>	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares to 10,000 shares Ratification and approval of the appointment of Ernst & 47,979,031 2,337,460 67,325 Young LLP as our independent auditors for the</pre>	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares to 10,000 shares Ratification and approval of the appointment of Ernst & 47,979,031 2,337,460 67,325 Young LLP as our independent auditors</pre>	
<pre>increasing the 25,965,492 11,723,453 118,392 number of shares of common stock underlying each annual option grant from 6,000 shares to 10,000 shares to 10,000 shares Ratification and approval of the appointment of Ernst & 47,979,031 2,337,460 67,325 Young LLP as our independent auditors for the</pre>	

December 31, 2005

ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K

(a)	Exhibits
EXHIBIT NO. DESCRIPTION	
+10.1	
Collaboration and License	
Agreement, dated May	
16, 2005,	
with N.V. Organon and	
(only with	
respect to Section 9.4	
thereof)	
Intervet Inc. 31.1	
Certification of CEO	
Pursuant to	
Section 302 of the	
Sarbanes-	
Oxley Act of 2002 31.2	
Certification	
of CFO Pursuant to	
Section 302	
of the Sarbanes-	
Oxley Act of 2002 32.1	
Certification	
of CEO and CFO Pursuant	
to Section	
906 of the Sarbanes-	
Oxley Act of 2002	
2002	

- + Confidential treatment has been requested for a portion of this exhibit. The confidential portions of this exhibit have been omitted and filed separately with the Securities and Exchange Commission
 - (b) Reports on Form 8-K:

On April 1, 2005, we filed a Current Report on Form 8-K dated March 28, 2005 related to our entry into a Consulting Agreement, dated March 28, 2005, with C. Thomas Caskey, M.D., a member of our board of directors.

On April 28, 2005, we filed a Current Report on Form 8-K dated April 27, 2005 related to the ratification and approval by our stockholders of an amendment to our 2000 Non-Employee Directors' Stock Option Plan. Additionally, the Form 8-K related to our issuance of a press release reporting our financial results for the quarter ended March 31, 2005, which press release included our consolidated balance sheet data and consolidated statements of operations data for the period.

On May 17, 2005, we filed a Current Report on Form 8-K dated May 16, 2005 related to our entry into a Collaboration and License Agreement, dated May 16, 2005, with N.V. Organon and (only with respect to Section 9.4 thereof) Intervet Inc.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

LEXICON GENETICS INCORPORATED

Date:	July 29,	2005	By:	/s/ Arthur T. Sands
				Arthur T. Sands, M.D., Ph.D. President and Chief Executive Officer
Date:	July 29,	2005	By:	/s/ Julia P. Gregory
			Julia P. Gregory Executive Vice President, Corporate Development and Chief Financial Officer	

DESCRIPTION ----------+10.1 --Collaboration and License Agreement, dated May 16, 2005, with N.V. Organon and (only with respect to Section 9.4 thereof) Interveť Inc. 31.1 --Certification of CEO Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 31.2 --Certification of CFO Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 32.1 --Certification of CEO and CFO Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

EXHIBIT NO.

+ Confidential treatment has been requested for a portion of this exhibit. The confidential portions of this exhibit have been omitted and filed separately with the Securities and Exchange Commission Confidential materials omitted and filed separately with the Securities and Exchange Commission. Asterisks denote omissions.

COLLABORATION AND LICENSE AGREEMENT

THIS COLLABORATION AND LICENSE AGREEMENT (this "Agreement") is dated as of May 16, 2005 (the "Effective Date") and is made by and between LEXICON GENETICS INCORPORATED, a corporation organized under the laws of the State of Delaware, United States of America ("Lexicon"), N.V. ORGANON, a registered company organized under the laws of the Netherlands ("Organon"), and (only with respect to Section 9.4) INTERVET Inc., a corporation organized under the laws of the State of Delaware, United States of America ("Intervet"). Lexicon and Organon are sometimes referred to herein individually as a "party" and collectively as the "parties."

RECITALS

WHEREAS, Lexicon and Organon are each in the business of discovering, developing and commercializing products for the prevention or treatment of human diseases and conditions;

WHEREAS, Lexicon has technology for and expertise in the identification and validation of gene and protein targets for use in the discovery of such products, including potential antibody and protein therapeutics, as well as expertise in the research and development of such products;

WHEREAS, Organon has expertise in the research, development and manufacture of such products; and

WHEREAS, Lexicon and Organon are interested in collaborating in the discovery and development of antibody and protein therapeutics for selected gene and protein targets;

NOW, THEREFORE, in consideration of the premises and of the covenants herein contained, the parties hereto mutually agree as follows:

ARTICLE 1. DEFINITIONS

For purposes of this Agreement, the terms defined in this Article 1 shall have the respective meanings specified below:

1.1 "Acquiring Party" has the meaning specified in Section 17.3 hereof.

1.2 "Adverse Drug Reaction" means any untoward medical occurrence in a patient or subject who is administered a Collaboration Product or Opt-out Product, whether or not considered related to the Collaboration Product or Opt-out Product, as applicable, including any undesirable sign (including abnormal laboratory findings of clinical concern), symptom or disease temporally associated with the use of such Collaboration Product or Opt-out Product.

1.3 "Affiliate" means any corporation, company, partnership, joint venture or firm that controls, is controlled by or is under common control with a party to this Agreement. For purposes hereof, "control" means (a) in the case of a corporate entity, direct or indirect ownership of more than fifty percent (50%) of the stock or shares entitled to vote for the election of directors; and (b) in the case of a non-corporate entity, direct or indirect ownership of more than fifty percent (50%) of the equity interests with the power to direct the management and policies of such non-corporate entity.

1.4 "Affiliated Subcontractor" has the meaning specified in Section 2.2.4 hereof.

1.5 "Allowable Costs" means the sum of the following (without any item being accounted for more than once): (a) Target Function Discovery Program Costs; (b) Biotherapeutics Research Program Costs; (c) Development Costs; (d) Product Supply Costs; and (e) reasonable out-of-pocket expenses associated with Patent Prosecution of Program Patent Rights. For clarity, Allowable Costs shall not include costs and expenses attributable to the research, development, manufacture or supply of Opt-out Products from and after the effectiveness of the relevant Opt Out.

1.6 "Antibody" means a composition comprising a whole antibody, or any fragment thereof.

1.7 "Antisense Compound" means a composition comprising an oligonucleotide or oligonucleotide analog, whether single or double-stranded, that hybridizes to a selected mRNA or otherwise interferes with translation or transcription of such mRNA.

1.8 "Background Materials" means Lexicon Background Materials and Organon Background Materials.

1.9 "Background Technology" means Lexicon Background Technology and Organon Background Technology.

1.10 "Back-up Product" means a Collaboration Product selected for Development by the Joint Management Committee as a possible [**] a previously-designated Collaboration Product for the same Program Target; provided, that a Collaboration Product may be deemed to have been selected as a [**] a previously-designated Collaboration Product for the same Program Target only if both Collaboration Products [**]. For clarity, a Collaboration Product selected as a possible [**] a previously-designated Collaboration Product shall be a Back-up Product if it [**].

1.11 "BLA" means a Biologics Licensing Application (as defined in the FDC Act) filed with the FDA and any other equivalent marketing authorization application or other license, registration or other application seeking approval from a Regulatory Authority to market a Collaboration Product in the Collaboration Field in any country or region within the Territory.

1.12 "Biotherapeutics Research" means the conduct of activities relating to the generation of Antibodies, Antisense Compounds and Proteins relating to Program Targets that have not become Opt-out Targets and the identification, characterization, selection, optimization and research of Program Antibodies, Program Antisense Compounds and Program Proteins prior to their designation as Collaboration Products. Biotherapeutics Research may include, without limitation, (a) the expression of Proteins comprising Program Targets, (b) the generation of Program Antibodies that bind to Program Targets, (c) the generation of Antisense Compounds that hybridize with mRNA encoded by Program Targets or otherwise interfere with translation or transcription of such mRNA, (d) the development of assays for Program Antibodies, Program Antisense Compounds and Program Proteins to, inter alia, confirm the activity of such Program Antibodies, Program Antisense Compounds or Program Proteins, and (e) the optimization of such Program Antibodies, Program Antisense Compounds or Program Proteins, as the case may be, in each case with the objective of identifying Program Antibodies, Program Antisense Compounds and Program Proteins that are suitable for Development and meet the criteria required for designation as Collaboration Products.

1.13 "Biotherapeutics Research Plan" means the plan to be developed by the Joint Research Committee and approved by the Joint Management Committee for each Contract Year in accordance with Section 5.2 hereof.

1.14 "Biotherapeutics Research Program" means the conduct of Biotherapeutics Research activities in accordance with an applicable Biotherapeutics Research Plan.

1.15 "Biotherapeutics Research Program Costs" means costs and expenses that are incurred after the Effective Date by either party and Affiliated Subcontractors in performing Biotherapeutics Research activities in accordance with an applicable Biotherapeutics Research Plan and associated budget approved by the Joint Management Committee, including:

> (a) the costs of internal scientific, medical, technical and managerial personnel engaged in Biotherapeutics Research activities (together with all associated laboratory supplies, facilities and occupancy costs), which costs shall be determined based on FTE Costs, unless another basis is otherwise agreed upon by the parties in writing;

> (b) out-of-pocket expenditures directly related to such Biotherapeutics Research activities, including payments to contract research organizations, consultants and other subcontractors, subject to Section 2.2.4; and

(c) any other costs expressly provided for and actually incurred in accordance with such Biotherapeutics Research Plan.

1.16 "Change in Control" means, with respect to either party, the occurrence of either (a) a Third Party becoming the "beneficial owner" (as defined in the rules and regulations promulgated under the Securities Exchange Act of 1934), directly or indirectly, of securities of such party representing fifty percent (50%) or more of the combined voting power of such party's then-outstanding voting securities or (b) the sale by such party, in one or more related transactions, of all or substantially all of such party's property and assets to any Third Party.

1.17 "cGMP Requirements" means the FDA's current good manufacturing practice requirements as promulgated under the FDC Act at 21 C.F.R. (parts 210 and 211), and as further defined by FDA guidance documents, as amended from time to time.

1.18 "Collaboration" means the program described in this Agreement in which Lexicon and Organon will collaborate to identify, characterize and carry out the Biotherapeutics Research, Development and Manufacturing of Antibodies (and, if appropriate, Antisense Compounds) that act through and Proteins that are encoded by Program Targets for use in the Collaboration Field.

1.19 "Collaboration Committee" means the Joint Management Committee, Joint Research Committee or Joint Development Committee.

1.20 "Collaboration Field" means the diagnosis, prevention, control and treatment in humans of any disease or condition.

1.21 "Collaboration Product" means a Program Antibody, Program Antisense Compound or Program Protein that has been selected for Development by the Joint Management Committee in accordance with Section 3.4 hereof and that has not become an Opt-out Product.

1.22 "Collaboration Term" has the meaning specified in Section 2.1.2 hereof.

1.23 "Confidential Information" means any proprietary information and data received by a party or its Affiliates (the "Receiving Party") from the other party or its Affiliates (the "Disclosing Party") in connection with this Agreement (including, without limitation, any research, testing, clinical,

regulatory, marketing or other scientific or business information, plans, or data pertaining to any Collaboration Product or Opt-out Product of the Disclosing Party). Notwithstanding the foregoing, Confidential Information shall not include any part of such information or data that:

> (a) is or becomes part of the public domain other than by unauthorized acts of the Receiving Party or its Affiliates;

(b) can be shown by written documents to have been already in the possession of the Receiving Party or its Affiliates prior to disclosure under this Agreement, provided such Confidential Information was not obtained directly or indirectly from the Disclosing Party under an obligation of confidentiality;

(c) can be shown by written documents to have been disclosed to the Receiving Party or its Affiliates by a Third Party, provided such Confidential Information was not obtained directly or indirectly from the Disclosing Party under an obligation of confidentiality; or

(d) can be shown by written documents to have been independently developed by the Receiving Party or its Affiliates without use of, or access to, Confidential Information of the Disclosing Party.

Specific Confidential Information of a Disclosing Party shall not be deemed to come under the foregoing exceptions merely because it is embraced by more general information that is or becomes part of the public domain, or is known by, disclosed to or independently developed by the Receiving Party.

1.24 "Continuing Party" has the meaning specified in Section 10.1 hereof.

1.25 "Contract Year" means (a) with respect to the first Contract Year, the period beginning on the Effective Date and ending on December 31, 2005 (the "First Contract Year"), and (b) with respect to each subsequent Contract Year, the twelve (12) month period beginning on the day following the end of the First Contract Year and each succeeding twelve (12) month period thereafter. Each Contract Year (other than the First and last Contract Year, as applicable) shall be divided into four (4) "Contract Quarters" comprised of successive three (3) month periods. In the First Contract Year, the first Contract Quarter shall begin on the Effective Date and end on June 30, 2005.

1.26 "Control" or "Controlled" means, with respect to any (a) material, document, item of information, method, data or other know-how or (b) Patent Right or other intellectual property right, the possession (whether by ownership or license, other than by a license granted pursuant to this Agreement) by a party or its Affiliates of the ability to grant to the other party access, ownership, a license or a sublicense as provided herein under such item or right without violating the terms of any agreement or other arrangement with any Third Party as of the time such party would first be required hereunder to grant the other party such access, ownership, license or sublicense.

1.27 "Cost Sharing Ratio" means (a) with respect to Lexicon, fifty percent (50%), and (b) with respect to Organon, fifty percent (50%), subject, in each case to adjustment in accordance with Section 11.4.

1.28 "Cover," "Covered" or "Covering" means, with respect to a Patent Right, that, but for rights granted to a person or entity under such Patent Right, the practice by such person or entity of an invention claimed in such Patent Right would infringe a Valid Claim included in such Patent Right, or in the case of a Patent Right that is a patent application, would infringe a Valid Claim in such patent application if it were to issue as a patent.

1.29 "Development" or "Develop" means the conduct of all tests, clinical and other studies and other activities (including test method development, toxicology studies, statistical analysis and report writing, preclinical and other testing, packaging and regulatory affairs, product approval and registration activities) set forth in, or required to obtain the information set forth in, applicable Development Plan(s), including such tests, studies (including Post-Approval Studies) and other activities as may be required or recommended from time to time by any Regulatory Authority to obtain, maintain or expand Regulatory Approval of a Collaboration Product in the Collaboration Field, but excluding any (a) Post-Approval Studies that are not so required or recommended by the applicable Regulatory Authority and (b) such studies which are required for purposes of obtaining or maintaining a pricing or reimbursement approval.

1.30 "Development Costs" means the costs and expenses that are incurred by either party or their Affiliated Subcontractors in the Development of a Collaboration Product in accordance with an applicable Development Plan and associated budget approved by the Joint Management Committee, including (without duplication):

> (a) the costs of internal scientific, medical, technical and managerial personnel engaged in Development activities with respect to such Collaboration Product (together with all associated laboratory supplies, facilities and occupancy costs), in each case to the extent not accounted for in other provisions of this definition (e.g., in Manufacturing Development Costs under clause (c) below or in Product Supply Costs under clause (e) below), which costs shall be determined based on FTE Costs, unless another basis is otherwise agreed upon by the parties in writing;

(b) out-of-pocket expenditures directly related to the Development of such Collaboration Product, including (i) payments to investigators, contract research organizations, consultants and other subcontractors for preclinical studies, pharmacodynamic and pharmacokinetic studies, toxicology studies, data management, statistical design, programming and analysis, clinical studies, clinical trial management, document preparation and review, subject recruitment and reimbursement, insurance, contract negotiation and travel relating to such activities; (ii) payments to investigators, contract research organizations, consultants and other subcontractors in connection with the preparation, filing and submission of INDs, BLAs and other regulatory filings with Regulatory Authorities (including pharmacoeconomic studies and any other clinical studies reasonably necessary for Regulatory Approval by relevant Regulatory Authorities to sell such Collaboration Product in a given country); and (iii) filing, submission and similar fees payable to Regulatory Authorities in connection with the preparation, filing and submission of such INDs, BLAs and other regulatory filings; in each case, subject to Section 2.2.4:

(c) Manufacturing Development Costs relating to such Collaboration Product;

(d) out-of-pocket expenditures under any Third Party licenses related to the Development of such Collaboration Product entered into (i) prior to the Effective Date and disclosed to the other party prior to the Effective Date or (ii) in accordance with Section 2.8, in each case other than royalty and other amounts paid to Third Parties in connection with the commercialization of such Collaboration Product or otherwise payable with respect to activities attributable to a given country and occurring after Regulatory Approval of such Collaboration Product in such country;

(e) the costs and expenses of clinical supplies and related charges directly related to the Development of such Collaboration Product as set forth in the Development Plan, including: (i) the Product Supply Costs of clinical supplies of such Collaboration Product; (ii) costs and expenses incurred to purchase or package comparator or combination drugs or devices; and (iii) costs and expenses of disposal of clinical samples; subject, as applicable, to Section 2.2.4; and

(f) any other costs expressly provided for and actually incurred in accordance with such Development Plan.

1.31 "Development Plan" means the plan(s) to be developed by the Joint Development Committee and approved by the Joint Management Committee for each Contract Year [**] with respect to the Development of a Collaboration Product in accordance with Section 6.3 hereof.

1.32 "Development Program" means the conduct of Development activities with respect to a Collaboration Product in accordance with an applicable Development Plan.

1.33 "Diligent Efforts" means the carrying out of obligations or tasks by a party (or, as applicable, its Affiliates) in a sustained manner using good faith commercially reasonable and diligent efforts, which efforts shall be consistent with the exercise of prudent scientific and business judgment in accordance with the efforts such party devotes to products or research, development or marketing projects of similar scientific and commercial potential. Diligent Efforts requires that the party or its applicable Affiliates: (a) promptly assign responsibility for such obligations to specific employees who are held accountable for progress and monitor such progress on an on-going basis, (b) set and consistently seek to achieve specific and meaningful objectives for carrying out such obligations, and (c) consistently make and implement decisions and allocate resources designed to advance progress with respect to such objectives.

1.34 "Disclosing Party" has the meaning specified in Section 1.23 hereof.

1.35 "Effective Date" means the date specified in the initial paragraph of this Agreement.

1.36 "EMEA" means the European Agency for the Evaluation of Medicinal Products, or any successor thereto.

1.37 "Event of Default" means an event described in Section 16.2 hereof.

 $1.38\ "FDA"$ means the United States Food and Drug Administration, or any successor thereto.

1.39 "FDC Act" means the United States Food, Drug and Cosmetic Act (or any successor thereto), as amended, and the rules and regulations promulgated thereunder.

1.40 "First Commercial Sale" means the first sale for use or consumption by the general public of a Collaboration Product or Opt-out Product in a country after Regulatory Approval has been obtained in such country. For clarity, First Commercial Sale shall not include the sale of any Collaboration Product or Opt-out Product for use in clinical trials or for compassionate use prior to the approval of a BLA.

1.41 "FTE" means a full-time-equivalent person-year of scientific, technical or managerial work on or directly related to Target Function Discovery Program, Biotherapeutics Research or Development activities, as applicable.

1.42 "FTE Costs" means the amounts (which amounts include salaries, fringe benefits, overtime and all other costs of employing FTEs, including overhead such as laboratory supplies, facilities

and occupancy costs) determined by multiplying (a) the number of FTEs allocated by a party or its Affiliated Subcontractor(s) during the relevant time period, subject to any limitations set forth in the applicable Biotherapeutics Research Plan or Development Plan or otherwise established by the Joint Management Committee, by (b) the applicable FTE Rate(s).

1.43 "FTE Rate" means the agreed-upon cost per FTE by functional area, to be adjusted annually (beginning in January 2006) for inflation using the latest available U.S. Producer Price Index for Total Manufacturing Industries, unadjusted (PCUOMFG#) as a simple percentage. Such adjustments shall be the responsibility of the Joint Management Committee. The initial FTE Rate is [**] on a per annum basis for the following areas: Biotherapeutics Research, preclinical Development, clinical Development, regulatory affairs, technical development, process development, technical transfer, project management and drug safety and quality assurance. The Joint Management Committee shall approve common FTE Rates for any new functional areas that come within the scope of the Collaboration. Notwithstanding the foregoing, if [**], then, [**].

1.44 "GAAP" means United States or international generally accepted accounting principles, as they exist from time to time, consistently applied.

1.45 "IND" means an Investigational New Drug application filed with the FDA or a similar application for the clinical testing of a Collaboration Product in human subjects filed with a foreign Regulatory Authority.

1.46 "Joint Development Committee" has the meaning specified in Section 3.1.3.

1.47 "Joint Development Project Team" has the meaning specified in Section 3.1.3.

1.48 "Joint Management Committee" has the meaning specified in Section 3.1.1 hereof.

1.49 "Joint Marketing/Development Collaborator" has the meaning specified in Section 8.3 hereof.

1.50 "Joint Marketing/Development Collaborator Revenue" means all license fees, royalties, milestone payments and other income or items of value (including, without limitation, any premium received on an equity investment in Lexicon, Organon or their respective Affiliates, as the case may be, by such Joint Marketing/Development Collaborator) received from a Joint Marketing/Development Collaborator in respect of a Collaboration Product (or an Opt-out Product for which the parties entered into a definitive agreement with such Joint Marketing/Development Collaborator prior to the effectiveness of such Opt Out, but only with respect to countries within the Territory covered by such definitive agreement), less any amounts specifically incurred in connection with acquiring such revenue (e.g., attorneys' fees to establish underlying agreements with a Joint Marketing/Development Collaborator or any potential Joint Marketing/Development Collaborator) and less any reasonable amounts of indemnity actually paid by either party under any agreements with a Joint Marketing/Development Collaborator.

1.51 "Joint Research Committee" has the meaning specified in Section 3.1.2 hereof.

1.52 "Joint Research Project Team" has the meaning specified in Section 3.1.2 hereof.

1.53 "[**]" means [**].

1.54 "Laws" means all laws, statutes, rules, regulations, ordinances and other pronouncements having the effect of law of any federal, national, multinational, state, provincial, county, city or other political subdivision, domestic or foreign.

1.55 "Level 1 Phenotypic Analysis" means the analyses of the phenotypes of Mutant Mice described in Exhibit 1.55.

1.56 "Level 2 Phenotypic Analysis" means any one or more of the analyses of the phenotypes of Mutant Mice described in Exhibit 1.56.

1.57 "Lexicon Background Materials" means any Antibodies, Antisense Compounds, Proteins, Mutant Mice, reagents, assays or other materials that are (a) necessary or useful for the conduct of the Collaboration, (b) Controlled by Lexicon, (c) utilized in the Collaboration (but only to the extent so utilized) and (d) either in Lexicon's or any of its Affiliates' possession as of the Effective Date or are discovered or acquired by Lexicon or any of its Affiliates during the Collaboration Term but outside of the conduct of the Collaboration. Lexicon Background Materials excludes Program Targets, Program Antibodies, Program Antisense Compounds and Program Proteins.

1.58 "Lexicon Background Technology" means any inventions, information, methods, know-how, trade secrets or data that (a) are necessary or useful for the performance of the Collaboration, (b) are Controlled by Lexicon, (c) are utilized in the Collaboration (but only to the extent so utilized) and (d) either are in Lexicon's or any of its Affiliates' possession as of the Effective Date or are discovered or acquired by Lexicon or any of its Affiliates during the Collaboration Term but outside of the conduct of the Collaboration. Lexicon Background Technology includes any inventions, information, method, know-how, trade secrets or data, other than Program Technology, that are first identified or discovered in the conduct of the Target Function Discovery Program. Lexicon Background Technology also includes any inventions, information, methods, know-how, trade secrets or data (i) relating to research and development methods and processes first identified or discovered by Lexicon or its Affiliated Subcontractor(s) in the course of performing Biotherapeutics Research or Development activities under the Collaboration or (ii) relating to manufacturing and analytical methods and processes first identified or discovered by Lexicon or its Affiliated Subcontractor(s) in the course of Manufacture of a Collaboration Product, in each case that are not Program Technology.

1.59 "Lexicon Product" means any Collaboration Product as to which Organon has Opted Out in accordance with Section 10.1 and, if applicable [**], any Program Antibodies, Program Antisense Compounds or Program Proteins that have not been designated for Development relating to the same Opt-out Target.

1.60 "Lexicon Opt-out Target" means any Program Target as to which Organon has Opted Out in accordance with Section 10.1.

1.61 "Major Market" means the United States, the European Union (under the centralized process or any other process), Germany, the United Kingdom, France, Italy, Spain or Japan.

1.62 "Manufacturing" or "Manufacture" means all activities set forth in the applicable Manufacturing Plan associated with the production, processing, filling, finishing, packaging, labeling, shipping and storage of Collaboration Products in the Collaboration Field, including stability testing, formulation, manufacturing process development, process validation, manufacturing scale-up, preclinical, clinical and commercial manufacture and analytical development and quality assurance and quality control activities.

1.63 "Manufacturing Development Costs" means, with respect to the Development of a Collaboration Product as set forth in the applicable Manufacturing Plan and associated budget approved by the Joint Management Committee, (a) the reasonable internal costs of the applicable party or its Affiliated Subcontractor(s), which costs shall be determined based on FTE Costs, unless another basis is otherwise agreed upon by the parties in writing, plus reasonable out-of-pocket expenditures and (b) the actual costs billed to such party or parties or to their Affiliated Subcontractors by Third Parties, subject to Section 2.2.4, each in accordance with the budget set forth in the applicable Manufacturing Plan, incurred in process development, process validation, process improvement, formulation development, facility and plant validation (which are product-specific), manufacturing scale-up and recovery costs, the development of standard operating procedures, batch records, and quality assurance and quality control methods and procedures, and the production of qualification lots, all costs incurred in obtaining and maintaining approval specifically for the manufacture of such Collaboration Product for commercial sale, and the costs for preparing, submitting, reviewing or developing data or information for the purpose of a drug master file or for submission to a Regulatory Authority to obtain or retain such approvals.

1.64 "Manufacturing Plan" has the meaning specified in Section 7.2 hereof.

1.65 "Mutant Mouse" means mouse cell or mouse containing a selected mutation in the murine ortholog of a Program Target that is made or produced by Lexicon. A "line of Mutant Mice" means Mutant Mice having the same selected mutation.

1.66 "Net Sales" means, with respect to a Collaboration Product or Opt-out Product, as the case may be, the gross amount invoiced for sales of such Collaboration Product or Opt-out Product by Lexicon and its Affiliates, Organon and its Affiliates, Joint Marketing/Development Collaborators and their Affiliates, and (sub)licensees of the Continuing Party and their Affiliates, in each case as applicable, to customers which are not Affiliates (or which are Affiliates but are end users of such Collaboration Product or Opt-out Product), less:

(a) trade, quantity and cash discounts actually allowed;

(b) discounts, refunds, rebates, chargebacks, retroactive price adjustments, billing errors and any other allowances (including, without limitation, government-mandated and managed health care-negotiated rebates) actually granted which effectively reduce the net selling price;

(c) product returns credits and allowances actually granted;

(d) any tax imposed on the production, sale, delivery or use of the product (excluding federal, state or local taxes based on income);

(e) freight, postage, shipping, customs duties, excises, tariffs, surcharges, other governmental charges (excluding federal, state or local taxes based on income) and insurance charges actually allowed or paid for delivery of the product;

(f) payments or rebates paid with respect to such Collaboration Product or Opt-out Product, as applicable, in connection with state or federal Medicare, Medicaid or similar programs in the United States or in connection with similar programs in other countries in which there are sales; and

(g) adjustments for bad debts actually incurred.

Such amounts shall be determined from the books and records of the Lexicon and its Affiliates, Organon and its Affiliates, Joint Marketing/Development Collaborators and their Affiliates, and (sub)licensees of the Continuing Party and their Affiliates, as the case may be, maintained in accordance with GAAP.

In the event the Collaboration Product or Opt-out Product, as applicable, is sold as part of a Combination Product (as defined below), the Net Sales from the Combination Product, for the purposes of determining royalty payments, will be determined by multiplying the Net Sales of the Combination Product by the fraction, A/A+B where A is the average sale price of such Collaboration Product or Opt-out Product when sold separately in finished form and B is the average sale price of the other active compounds or ingredients in the Combination Product sold separately in finished form.

In the event that the average sale price of the Collaboration Product or Opt-out Product, as applicable, can be determined but the average sale price of the other active compounds or ingredients cannot be determined, Net Sales for purposes of determining royalty payments will be calculated by multiplying the Net Sales of the Combination Product by the fraction C/C+D where C is the selling party's average sales price of the Collaboration Product or Opt-out Product and D is the difference between the average selling price of the Combination Product and the average selling price of the Collaboration Product or Opt-out Product. If the average sale price of the other active compounds or ingredients can be determined but the average price of the Collaboration Product or Opt-out Product, as applicable, cannot be determined, Net Sales for purposes of determining royalty payments will be calculated by multiplying the Net Sales of the Combination Product by the following formula: one minus C/C+D where C is the average selling price of the other product(s) and D is the difference between the average selling price of the Combination Product and the average selling price of the other active compounds or ingredients.

In the event that the average sales price of both the Collaboration Product or Opt-out Product, as applicable, and the other active compounds or ingredients in the Combination Product cannot be determined, the Net Sales of the Collaboration Product or Opt-out Product shall be negotiated in good faith by the parties.

The Net Sales price for a Combination Product in a given country will be calculated once each Contract Year and such price will be used during all applicable royalty reporting periods for the entire Contract Year for such country, absent extraordinary conditions or events. When determining the average sale price of a Collaboration Product or Opt-out Product, as applicable, or the other active compounds or ingredients in the Combination Product, the average sale price will be calculated using data arising from the twelve (12) months preceding the calculation of the Net Sales price for the Combination Product. As used above, the term "Combination Product" means any Collaboration Product or Opt-out Product, as applicable, sold in conjunction with any other active component(s) (whether packaged together or in the same therapeutic formulation).

Free samples of Collaboration Product or Opt-out Product, as applicable, and the disposition of Collaboration Product or Opt-out Product for, or the use of Collaboration Product or Opt-out Product in, Phase 1 Clinical Trials, Phase 2 Clinical Trials, Phase 3 Clinical Trials or Post-Approval Studies in which such Collaboration Product or Opt-out Product is provided to patients without any payment shall not result in any Net Sales.

1.67 "Oncology/Immunology Field" means the diagnosis, prevention, control and treatment in humans of (a) cancer and (b) diseases and conditions of the immune system other than those associated with diabetes, obesity, cardiovascular disease, diseases and conditions of the eye, and diseases and conditions of the central and peripheral nervous system.

1.68 "Opt Out" has the meaning specified in Section 10.1 hereof.

1.69 "Opt-out Party" has the meaning specified in Section 10.1 hereof.

1.70 "Opt-out Product" means a Lexicon Product or an Organon Product.

1.71 "Opt-out Target" means a Lexicon Opt-out Target or an Organon Opt-out Target.

1.72 "Organon Background Materials" means any Antibodies, Antisense Compounds, Proteins, assays, reagents or other materials that are (a) necessary or useful for the conduct of the Collaboration, (b) Controlled by Organon, (c) utilized in the Collaboration (but only to the extent so utilized) and (d) either in Organon's or any of its Affiliates' possession as of the Effective Date or are discovered or acquired by Organon or any of its Affiliates during the Collaboration Term but outside of the conduct of the Collaboration. Organon Background Materials excludes Program Targets, Program Antibodies, Program Antisense Compounds and Program Proteins.

1.73 "Organon Background Technology" means any inventions, information, methods, know-how, trade secrets or data that (a) are necessary or useful for the performance of the Collaboration, (b) are Controlled by Organon, (c) are utilized in the Collaboration (but only to the extent so utilized) and (d) either are in Organon's or any of its Affiliates' possession as of the Effective Date or are discovered or acquired by Organon or any of its Affiliates during the Collaboration Term but outside of the conduct of the Collaboration. Organon Background Technology also includes any inventions, information, methods, know-how, trade secrets or data (i) relating to research and development methods and processes first identified or discovered by Organon or its Affiliated Subcontractor(s) in the course of performing Biotherapeutics Research or Development activities under the Collaboration or (ii) relating to manufacturing and analytical methods and processes first identified or discovered by Organon or its Affiliated Subcontractor(s) in the course of Manufacture of a Collaboration Product, in each case that are not Program Technology.

1.74 "Organon Product" means any Collaboration Product as to which Lexicon has Opted Out in accordance with Section 10.1 and, if applicable [**], any Program Antibodies, Program Antisense Compounds or Program Proteins that have not been designated for Development relating to the same Opt-out Target.

1.75 "Organon Opt-out Target" means any Program Target as to which Lexicon has Opted Out in accordance with Section 10.1.

1.76 "Patent Prosecution" has the meaning specified in Section 12.2.1 hereof.

1.77 "Patent Rights" means all existing patents and patent applications and all patent applications hereafter filed and patents hereafter issued, including, without limitation, any continuations, continuations-in-part, divisions, provisionals or any substitute applications, any patent issued with respect to any such patent applications, any reissue, reexamination, renewal or extension (including any supplemental protection certificate) of any such patent, and any confirmation patent or registration patent or patent of addition based on any such patent, and all foreign counterparts of any of the foregoing.

1.78 "Phase 1 Clinical Trial" means a human clinical trial in any country that is intended to initially evaluate the safety or pharmacological effect of a Collaboration Product in subjects or that would otherwise satisfy requirements of 21 CFR 312.21(a), or its foreign equivalent. For purposes of this Agreement, "commencement of a Phase 1 Clinical Trial" for a Collaboration Product means the first introduction of such Collaboration Product into a human patient in a Phase 1 Clinical Trial.

1.79 "Phase 2 Clinical Trial" means a human clinical trial in any country that is intended to initially evaluate the effectiveness of a Collaboration Product for a particular indication or indications in patients with the disease or indication under study or that would otherwise satisfy requirements of 21 CFR 312.21(b), or its foreign equivalent. For purposes of this Agreement, "commencement of a Phase 2 Clinical Trial" for a Collaboration Product means the first introduction of such Collaboration Product into a human patient in a Phase 2 Clinical Trial.

1.80 "Phase 2b Clinical Trial" means any Phase 2 Clinical Trial of a Collaboration Product that follows the completion of an initial Phase 2 Clinical Trial of such Collaboration Product, and that is a well-controlled clinical trial with the purpose of evaluating safety and efficacy in subjects who have the disease condition to be treated, diagnosed or prevented. For purposes of this Agreement, "commencement of a Phase 2b Clinical Trial" for a Collaboration Product means the first introduction of such Collaboration Product into a human patient in a Phase 2b Clinical Trial.

1.81 "Phase 3 Clinical Trial" means a pivotal human clinical trial in any country the results of which could be used to establish safety and efficacy of a Collaboration Product as a basis for a BLA or that would otherwise satisfy the requirements of 21 CFR 312.21(c) or its foreign equivalent. For purposes of this Agreement, "commencement of a Phase 3 Clinical Trial" for a Collaboration Product means the first introduction of such Collaboration Product into a human patient in a Phase 3 Clinical Trial.

1.82 "Plan" means a Biotherapeutics Research Plan, Development Plan or Manufacturing Plan, as the case may be.

1.83 "Post-Approval Study" means a clinical trial conducted after Regulatory Approval of the applicable Collaboration Product for the applicable indication has been obtained in the relevant country.

1.84 "Pre-existing Obligations" means the obligations of Lexicon or Organon, as the case may be, existing under agreements in effect prior to the Effective Date with respect to Background Materials and Background Technology, in each case as disclosed to the other party as applicable during the Collaboration Term.

1.85 "Product Supply Costs" means (a) to the extent that a Collaboration Product is sourced from a party or its Affiliated Subcontractor(s), the cost of Manufacture of such Collaboration Product, including (i) direct material and direct labor costs, (ii) manufacturing overhead fairly allocated to such Collaboration Product and (iii) [**], all calculated in accordance with GAAP and without regard to whether or not such costs result in usable products or materials, and (b) to the extent that a Collaboration Product is sourced from a Third Party, subject to Sections 2.2.4 and 7.1.2, the actual price paid to such Third Party for the manufacture, supply and packaging of such Collaboration Product. For purposes of the foregoing, (i) "direct material costs" means actual costs incurred in manufacturing or purchasing materials, including freight-in costs, sales and excise taxes imposed thereon and customs duty and charges levied by government authorities, and all costs of packaging components; (ii) "direct labor costs" means actual cost of employees engaged in direct manufacturing activities and quality control and quality assurance activities who are directly employed in manufacturing and packaging such Collaboration Product; and (iii) "manufacturing overhead" attributable to such Collaboration Product will include a reasonable allocation of indirect labor (not previously included in direct labor costs), a reasonable allocation of administrative costs, and a reasonable allocation of facilities costs, all in accordance with GAAP, but will not include corporate administrative overhead or plant start-up costs or costs associated with excess capacity. All allocations will be based on the assumption that such party's plant and equipment are utilized to their reasonable full capacity (except with respect to equipment that is specific to the Collaboration Product being Manufactured), and all costs and allocations shall be consistent with

the methods used for such costs and allocations for such party's internal purposes. More specifically, the components of Product Supply Costs shall comprise: (A) direct labor (fermentation, purification personnel); (B) direct materials; (C) facility costs (rent, property taxes, depreciation of leaseholds, utilities, spare parts, maintenance contracts); (D) manufacturing equipment depreciation; (E) allocations for information technology, document control, quality engineering, purchasing, warehouse management, microbiology (with such allocations to be based on estimated service levels, headcount or square footage occupancy, depending on the category); (F) indirect labor (manufacturing supervision); (G) manufacturing department overhead (uniforms, materials used in plant maintenance); (H) quality assurance/quality control; and (I) such other similar costs as may be reasonably included in such definition.

1.86 "Program Director" has the meaning specified in Section 3.2 hereof.

1.87 "Program Antibody" means an Antibody that: (a) (i) is selected by the Joint Research Committee for research, optimization or preclinical evaluation in the conduct of the Collaboration, (ii) is Controlled by a party, (iii) either is in a party's or any of its Affiliates' possession as of the Effective Date or is discovered or acquired by either or both parties or any of their respective Affiliates during the Collaboration Term but outside the conduct of the Collaboration, and (iv) binds to or otherwise modulates a Program Target; (b) is first [**] in the conduct of the Collaboration; or (c) is otherwise designated a Program Antibody by the Joint Management Committee; provided, however, that in no event shall [**] become a Program Antibody unless such designation is affirmatively agreed to by the Joint Management Committee after disclosure of the nature of such Pre-existing Obligation by the applicable party.

1.88 "Program Antisense Compound" means an Antisense Compound that: (a)(i) is selected by the Joint Research Committee for research, optimization or preclinical evaluation in the conduct of the Collaboration, (ii) is Controlled by a party, (iii) either is in a party's or any of its Affiliates' possession as of the Effective Date or is discovered or acquired by either or both parties or any of their respective Affiliates during the Collaboration Term but outside the conduct of the Collaboration, and (iv) binds to or otherwise modulates a Program Target; (b) is first [**] in the conduct of the Collaboration; or (c) is otherwise designated a Program Antisense Compound by the Joint Management Committee; provided, however, that in no event shall [**] become a Program Antisense Compound unless such designation is affirmatively agreed to by the Joint Management Committee after disclosure of the nature of such Pre-existing Obligation by the applicable party.

1.89 "Program Intellectual Property" means Program Patent Rights and any other proprietary rights in Program Material and Program Technology.

1.90 "Program Materials" means (a) any Program Antibodies, (b) any Program Antisense Compounds, (c) any Program Proteins, and (d) any materials other than Program Antibodies, Program Antisense Compounds or Program Proteins first identified or discovered in the conduct of the Collaboration.

1.91 "Program Patent Rights" means any Patent Rights that are Controlled by one or both parties and that Cover any Program Technology or Program Materials. For clarification, such Program Patent Rights include the entire scope of all of the claims contained in such Patent Rights.

1.92 "Program Protein" means a Protein that: (a) (i) is selected by the Joint Research Committee for research, optimization or preclinical evaluation in the conduct of the Collaboration, (ii) is Controlled by a party, (iii) either is in a party's or any of its Affiliates' possession as of the Effective Date or is discovered or acquired by either or both parties or any of their respective Affiliates during the Collaboration Term but outside the conduct of the Collaboration, and (iv) is encoded by a Program Target

or derived from (e.g., by means of amino acid substitutions, additions, deletions, and C- and N-terminal fusions) a Protein encoded by a Program Target; (b) is first [**] in the conduct of the Collaboration; or (c) is otherwise designated a Program Protein by the Joint Management Committee; provided, however, that in no event shall [**] become a Program Protein unless such designation is affirmatively agreed to by the Joint Management Committee after disclosure of the nature of such Pre-existing Obligation by the applicable party.

1.93 "Program Target" means one of the three hundred (300) human genes selected in accordance with Section 2.3 (in each case, identified by the full-length cDNA or amino acid sequence of the gene or, in the event the gene has more than one splice variant form, by the full-length cDNA or amino acid sequence of at least one splice variant form of such gene) and the products encoded by such gene, including, without limitation, (a) any [**] from such gene [**], (b) any [**] encoded by any such gene, and (c) any [**] encoded by any such gene.

1.94 "Program Technology" means any inventions, information, methods, know-how, trade secrets or data that (a) are Controlled by a party or jointly by the parties and (b)(i) relate to the use of Program Antibodies, Program Antisense Compounds or Program Proteins, (ii) relate to the use of a Program Target to identify Antibodies or Antisense Compounds acting through such Program Targets, and the use of such Antibodies or Antisense Compounds in the Collaboration Field or Veterinary Field, (iii) relate to the use of Protein(s) encoded by such Program Target in the Collaboration Field or Veterinary Field, or (iv) are first identified or discovered in the conduct of the Collaboration. For clarity, Program Technology excludes Program Materials. Notwithstanding the foregoing, Program Technology does not include any inventions, information, methods, know-how, trade secrets or data (A) relating to [**] or (B) relating to [**], in each case that do not [**].

1.95 "Protein" means a composition comprising a high molecular weight (i.e., weighing more than [**]), polymer compound composed of a variety of amino acids joined by peptide linkages, or any fragment thereof.

1.96 "Receiving Party" has the meaning specified in Section 1.23 hereof.

1.97 "Regulatory Approval" means any and all approvals (including any applicable governmental price and reimbursement approvals), licenses, registrations, or authorizations of any federal, national, multinational, state, provincial or local regulatory agency, department bureau or other governmental entity that are necessary for the Manufacture, use, storage, import, transport, promotion, marketing and sale of a Collaboration Product in the Collaboration Field in a country or group of countries.

1.98 "Regulatory Authority" means any governmental authority in a country or region that regulates the manufacture or sale of pharmaceutical products, including the FDA and the EMEA, and any successors thereto.

1.99 "Small Molecule Compound" means a composition comprising a chemical compound, whether synthetic or naturally-derived, having a molecular weight of less than [**]. For clarity, Small Molecule Compounds specifically exclude: (a) Antibodies, (b) Antisense Compounds and (c) Proteins.

1.100 "Specifications" means, with respect to any Collaboration Product, the applicable written specifications for Manufacturing, filling, packaging and warehousing such Collaboration Product in effect at a particular time and approved by the Joint Development Committee, including, but not limited to, specifications provided in any Regulatory Approval for such Collaboration Product.

1.101 "Target Function Discovery Program" has the meaning specified in Section 2.1.1 hereof.

1.102 "Target Function Discovery Program Costs" means the following amounts:

(a) [**] for each Program Target for which Lexicon or its Affiliated Subcontractor(s) [**] in the conduct of the Target Function Discovery Program;

(b) the costs of internal scientific, medical, technical and managerial personnel engaged in [**] (together with all associated laboratory supplies, facilities and occupancy costs), which costs shall be determined based on FTE Costs, unless another basis is otherwise agreed upon by the parties in writing; and

(c) out-of-pocket expenditures, if any, expressly approved by the Joint Management Committee and actually incurred after the Effective Date by Lexicon or its Affiliated Subcontractor(s) in conducting the Target Function Discovery Program, subject to Section 2.2.4.

1.103 "Target Function Discovery Program Term" has the meaning specified in Section 4.1.2 hereof.

1.104 "Territory" means all of the countries of the world.

1.105 "Therapeutic Area" means any one of the following areas with respect to which the Joint Research Committee may approve Level 2 Phenotypic Analysis of Mutant Mice relating to a Program Target: (a) metabolism and endocrinology, (b) cardiology, (c) neurology, (d) oncology, (e) immunology, (f) ophthalmology and (g) any other therapeutic area for which Lexicon subsequently develops Level 2 Phenotypic Analysis capabilities and focuses its own internal drug discovery efforts.

1.106 "Third Party" means any person or entity other than Lexicon, Organon and their respective Affiliates.

1.107 "Valid Claim" means a claim of an issued and unexpired patent which has not been held permanently revoked, unenforceable or invalid by a decision of a court or other governmental agency of competent jurisdiction, unappealable or unappealed within the time allowed for appeal and that is not admitted to be invalid or unenforceable through reissue, disclaimer or otherwise.

1.108 "Veterinary Field" means the diagnosis, prevention, control and treatment in animals other than humans of any disease or condition.

ARTICLE 2. COLLABORATION OVERVIEW

2.1 General.

2.1.1 Objectives. The parties intend to carry out their obligations and responsibilities under the Collaboration, consistent with the objectives set forth in and the resources allocated to such activities in the applicable Plan(s). It is intended that the Collaboration will be conducted as a unified collaborative effort with activities by the parties carried out primarily at each party's respective facilities, and this intent shall be reflected in the applicable Plan(s). It is further intended that each party shall contribute to Target Function Discovery Program Costs, Biotherapeutics Research Program Costs and Development Costs in proportion to the applicable Cost Sharing Ratio, and the Plans will be consistent with and provide for such proportional contribution. In support of the Collaboration, Lexicon will conduct efforts, using its technology for the generation and analysis of the phenotypes of Mutant Mice, to identify and validate Program Targets with potential utility in the Collaboration Field (the "Target Function Discovery Program").

2.1.2 Collaboration Term. The Collaboration shall commence on the Effective Date and continue until the earlier of (a) the time that one or the other party has Opted Out of all Program Targets or (b) the expiration or earlier termination of this Agreement pursuant to Article 16 hereof (the "Collaboration Term").

2.2 Conduct of Collaboration.

2.2.1 Efforts. The Joint Management Committee shall adopt project progression guidelines, including criteria for the designation of Collaboration Products for Development, the filing of INDs, the commencement of Phase 1 Clinical Trials, Phase 2 Clinical Trials and Phase 3 Clinical Trials, and the filing of BLAs. The parties shall conduct the Collaboration in good scientific manner in accordance with such project progression guidelines and in compliance with applicable Laws. Each party shall use Diligent Efforts to conduct the activities of the Collaboration that are assigned to it in the then-applicable Plan(s), and each shall devote sufficient resources to carry out such respective activities. While the parties acknowledge and agree that neither party guarantees the success of the Collaboration or any individual task undertaken thereunder, each party agrees that it will perform the activities assigned to it under the Collaboration in a professional manner in accordance with the highest industry standards.

2.2.2 Resources. Over the course of the Collaboration, tasks will be allocated between the parties in accordance with the following principles and objectives: (a) Lexicon will be solely responsible for conducting the Target Function Discovery Program; (b) except to the extent otherwise provided in an applicable Biotherapeutics Research Plan, Lexicon will be principally, but not exclusively, responsible for conducting Biotherapeutics Research activities involving the generation of research Program Antibodies, Program Antisense Compounds and Program Proteins and the conduct of in vivo research and proof of concept studies (although Organon may also be involved in such activities), Organon will be responsible for conducting Biotherapeutics Research activities involving [**] and the process development of Program Antibodies, Program Antisense Compounds and Program Proteins in preparation for Development, and the parties shall be allocated responsibility substantially pursuant to the applicable Cost Sharing Ratio; (c) each party's participation in Development activities (based on FTE utilization and out-of-pocket expenditures) will be substantially pursuant to the applicable Cost Sharing Ratio; and (d) each party's share of Target Function Discovery Program Costs, Biotherapeutics Research Program Costs and Development Costs will be substantially pursuant to the applicable Cost Sharing Ratio. Subject to and in accordance with the foregoing, particular tasks and responsibilities shall be assigned in a manner consistent with each party's respective capabilities, capacity and expertise. For purposes of this Agreement, "out-of-pocket expenditures" includes, but is not limited to, the cost of subcontractors related to the Collaboration, subject to Section 2.2.4, but specifically excludes the cost of laboratory supplies and facilities and occupancy costs (such as, for example, costs for laboratory space, equipment and utilities).

2.2.3 FTE Levels. The parties agree to commit to the Collaboration the personnel necessary to meet their respective responsibilities set forth in each Plan. The Plans shall set forth specific FTE levels for each Contract Year to be assigned to specific activities.

2.2.4 Subcontractors. In accordance with Section 2.2.2, the parties will endeavor to optimize the allocation of their resources for the conduct of the Collaboration. As necessary and

in furtherance of the Collaboration, however, (a) Lexicon may enter into agreements or subcontracts for Target Function Discovery Program activities in accordance with this Section 2.2.4 and (b) either party may enter into agreements or subcontracts for Biotherapeutics Research, Development or Manufacturing activities in accordance with this Section 2.2.4; provided that (i) none of the rights of the other party hereunder are diminished or otherwise adversely affected as a result of such subcontracting, (ii) such party obtains the written approval of the other party prior to engaging any subcontractor, which approval shall not be unreasonably withheld or delayed (for purposes of which it shall not be deemed unreasonable for a party to withhold consent when the withholding party is capable of conducting the activities proposed to be conducted by such subcontractor and is willing to do so on terms, including cost, time and quality, equivalent to those offered by such contractor), and (iii) the subcontractor undertakes in writing obligations of confidentiality and non-use regarding the other party's Confidential Information that are substantially the same as those undertaken by Organon and Lexicon pursuant to Article 13 hereof. In the event a party performs one or more of its obligations under the Collaboration through a subcontractor, then such party shall at all times be responsible for the performance of such subcontractor. The Joint Management Committee shall decide the allocation of the cost of any such agreement between the parties or if the cost is to be borne by one party and whether it can be allocated to offset other obligations set forth in the applicable Plan(s). Notwithstanding the foregoing provisions of this Section 2.2.4, either party may subcontract Target Function Discovery Program, Biotherapeutics Research, Development or Manufacturing activities to an Affiliate without the other party's prior consent (such party, an "Affiliated Subcontractor").

2.2.5 Reports. Lexicon shall submit [**] reports to the Joint Management Committee detailing its activities under the Target Function Discovery Program. Each party shall submit [**] reports to the Joint Management Committee, as may be required by the then-current Plan(s), detailing its activities under the Biotherapeutics Research Program and Development Program. The Joint Management Committee shall use such [**] reports to monitor the parties' respective contributions to the Collaboration. The Joint Management Committee may amend the Plan(s) as necessary to maintain substantial compliance over the course of the Collaboration in resources devoted and participation by the parties in accordance with the principles and objectives set forth in Section 2.2.2.

2.2.6 Adjustments. If either party believes that the parties are not devoting resources and participation to the Collaboration substantially in accordance with the principles and objectives set forth in Section 2.2.2, such party may submit the matter to the Joint Management Committee in writing, providing a reasonably detailed description of its reasons for such belief. Taking into account historical and prospective participation and resource devotion of the Parties during the current [**] and the immediately following [**], the Joint Management Committee shall take such steps as may be reasonably necessary to ensure substantial compliance in resources devoted and participation by the parties in the Collaboration with the principles and objectives set forth in Section 2.2.2.

2.3 Selection of Program Targets. Within [**] after the Effective Date, the Joint Management Committee shall select, from the list of available genes provided by Lexicon to Organon prior to the Effective Date, [**] Program Targets in addition to the two Program Targets separately designated as of the Effective Date. [**]. In the event that the Joint Management Committee is unable, within the relevant time periods specified above, to reach agreement with respect to the designation of the full number of Program Targets contemplated hereby, Organon shall have the right, during the period of [**] thereafter, to designate any remaining Program Targets from the relevant list of available genes.

2.4 Exclusivity. During the Collaboration Term, each party shall work exclusively with the other party under the terms of this Agreement with respect to Biotherapeutics Research, Development and commercialization of (a) Program Antibodies, Program Antisense Compounds and other Antibodies and Antisense Compounds acting through Program Targets and (b) Program Proteins and other Protein(s) encoded by Program Targets. For clarity, a Continuing Party and, subject to Section 10.2, an Opt-out Party shall have no further obligation under this Section 2.4 with respect to Opt-out Products and Opt-out Targets.

2.5 Collaboration Records.

2.5.1 All work conducted by each party in the course of the Collaboration shall be completely and accurately recorded, in reasonable detail and in good scientific manner, in separate laboratory notebooks. On reasonable notice, and at reasonable intervals, each party shall have the right to inspect and copy all such records of the other party reflecting Program Technology or work done under the Collaboration, to the extent reasonably required to carry out its respective obligations and to exercise its respective rights hereunder. Notwithstanding the definition of "Confidential Information," all such records shall constitute Confidential Information of the party owning such records.

2.5.2 In order to protect the parties' Patent Rights under U.S. law in any inventions conceived or reduced to practice during or as a result of the Collaboration, each party agrees to maintain a policy that requires its employees to record and maintain all data and information developed during the Collaboration in such a manner as to enable the parties to use such records to establish the earliest date of invention or diligence to reduction to practice. At a minimum, the policy shall require such individuals to record all inventions generated by them in standard laboratory notebooks or other suitable means that are dated and corroborated by non-inventors on a regular, contemporaneous basis.

2.6 Disclosure of Collaboration Results. Subject to restrictions imposed by a party's confidentiality obligations to any Third Party with respect to Background Materials or Background Technology, each party will disclose to the Joint Research Committee or Joint Development Committee, as applicable, all Program Technology that is discovered, invented or made by such party during the course of the Collaboration and that is useful in or relates to the Collaboration, including, without limitation, information regarding Program Targets, Program Antibodies, Program Antisense Compounds and Program Proteins and uses thereof and the results of all Biotherapeutics Research and Development studies and other activities. Such Program Technology will be promptly disclosed to the Joint Research Committee or Joint Development Committee, as applicable, with meaningful discoveries or advances being communicated as promptly as practicable after such information is obtained or its significance is appreciated. Upon written request by any member of the Joint Research Committee or Joint Development Committee, as applicable, each party will provide the other with copies of the raw data generated in the course of the Collaboration, if reasonably necessary to the other party's work under the Collaboration. Any information disclosed pursuant to this Section 2.6 may be used by the other party solely for the purposes of the Collaboration or as otherwise expressly permitted in this Agreement.

2.7 Material Transfer. In order to facilitate the Collaboration, either party may provide to the other party certain Program Materials and Background Materials Controlled by the supplying party (other than under this Agreement) for use by the other party in furtherance of the Collaboration. All such Program Materials shall be considered the Confidential Information of both parties and shall be subject to the restrictions in Article 13. All Background Materials shall be considered the Confidential Information of the supplying party and shall be subject to the restrictions in Article 13. Except as otherwise provided under this Agreement, all such Program Materials and Background Materials delivered to the other party shall remain the sole property of the supplying party, shall be used only in furtherance of the Collaboration and solely under the control of the other party and its Affiliates, shall not be used or delivered to or for the benefit of any Third Party without the prior written consent of the supplying party and shall not be used in research or testing involving human subjects. The Program Materials and Background Materials supplied under this Section 2.7 must be used with prudence and appropriate caution in any experimental work, since not all of their characteristics may be known. THE PROGRAM MATERIALS AND BACKGROUND MATERIALS ARE PROVIDED "AS IS" AND WITHOUT ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, INCLUDING WITHOUT LIMITATION ANY IMPLIED WARRANTY OF MERCHANTABILITY OR OF FITNESS FOR ANY PARTICULAR PURPOSE OR ANY WARRANTY THAT THE USE OF THE MATERIALS WILL NOT INFRINGE OR VIOLATE ANY PATENT OR OTHER PROPRIETARY RIGHTS OF ANY THIRD PARTY.

2.8 Third Party Licenses. Either party may propose that the Joint Management Committee determine whether a Third Party license is required or beneficial for Biotherapeutics Research of a Program Target or the Development or commercialization of a Collaboration Product in the Collaboration Field. In the event the Joint Management Committee determines that such Third Party license is required or beneficial, the Joint Management Committee shall determine which party shall be responsible for obtaining such license, as applicable. In making any such determination provided for in this Section 2.8 as to the need for or benefit of any such Third Party license, due consideration shall be given to the advisability of seeking an opinion of counsel and the efforts required to design around the patents at issue.

ARTICLE 3. COLLABORATION MANAGEMENT

3.1 Collaboration Committees.

3.1.1 Joint Management Committee. As soon as practicable after the Effective Date, Organon and Lexicon shall establish a Joint Management Committee (the "Joint Management Committee") comprised of [**] representatives designated by Organon and [**] representatives designated by Lexicon, each of whom shall have experience and seniority sufficient to enable him or her to make decisions on behalf of the party he or she represents; provided that Organon and Lexicon may, by mutual agreement, designate an appropriate number of additional representatives from time to time.

3.1.2 Joint Research Committee. As soon as practicable after the Effective Date, Organon and Lexicon shall establish a Joint Research Committee (the "Joint Research Committee") comprised of [**] representatives designated by Organon and [**] representatives designated by Lexicon, each of whom shall have experience and seniority sufficient to enable him or her to make decisions on behalf of the party he or she represents; provided that Organon and Lexicon may, by mutual agreement, designate an appropriate number of additional representatives from time to time. From time to time during the Collaboration Term, the Joint Research Committee may establish one or more Joint Research Project Teams (each, a "Joint Research Project Team") to implement various aspects of the applicable Biotherapeutics Research Plan. Such teams shall be governed in the same manner and subject to the relevant requirements as set forth herein for the Joint Research Committee.

3.1.3 Joint Development Committee. As soon as practicable after the designation of the first Collaboration Product, Organon and Lexicon shall establish a Joint Development Committee (the "Joint Development Committee") comprised of [**] representatives designated by Organon and [**] representatives designated by Lexicon, each of whom shall have experience and seniority sufficient to enable him or her to make decisions on behalf of the party he or she

represents; provided that Organon and Lexicon may, by mutual agreement, designate an appropriate number of additional representatives from time to time. From time to time during the Collaboration Term, the Joint Development Committee may establish one or more Joint Development Project Teams (each, a "Joint Development Project Team") to implement various aspects of the applicable Development Plan(s). Such teams shall be governed in the same manner and subject to the relevant requirements as set forth herein for the Joint Development Committee.

3.2 Program Directors. Each party shall appoint one of its designees on the Joint Management Committee (and who may, but need not also be, a member of the Joint Research Committee or Joint Development Committee) to serve as a program director (each, a "Program Director") with responsibility for overseeing the day-to-day activities of the parties with respect to the Collaboration and for being the primary point of contact between the parties with respect to the Collaboration.

3.3 Replacement of Collaboration Committee Representatives and Program Directors. Each party shall be free to replace its representative members of any Collaboration Committee and its Program Director with new appointees who have authority to act on behalf of such party, on notice to the other party.

3.4 Responsibilities of Joint Management Committee. The Joint Management Committee shall be responsible for overseeing and directing the parties' interaction and performance of their respective obligations under this Agreement. Without limiting the generality of the foregoing, its duties shall include:

> (a) preparing such procedures as may be necessary for the operation of the Joint Management Committee, Joint Research Committee, and Joint Development Committee, and other committees the Joint Management Committee decides to establish to assure the efficient operation of the Collaboration;

(b) approving strategy for the overall Biotherapeutics Research, Development and Manufacturing of Collaboration Products in the Collaboration Field and for all other activities conducted by the parties hereunder;

(c) reviewing and approving the annual Biotherapeutics Research Plans proposed by the Joint Research Committee and approving the budget therefor and any modifications thereto as recommended by the Joint Research Committee;

(d) reviewing and approving the annual [**] Development Plans proposed by the Joint Development Committee and approving the budgets therefor and any modifications thereto as recommended by the Joint Development Committee;

(e) reviewing and approving the Manufacturing Plans proposed by Organon or its Affiliated Subcontractor(s) and approving the budget therefor and any modifications thereto as recommended by Organon or such Affiliated Subcontractor(s);

(f) overseeing the implementation of the Plans and allocation of resources and other activities in support of the Collaboration, including the matters contemplated by Section 2.2 hereof;

(g) establishing criteria for designation of Collaboration $\ensuremath{\mathsf{Products}}\xspace;$

(h) designating Collaboration Products;

(i) facilitating the transfer of technology between the parties through the Joint Research Committee and the Joint Development Committee;

(j) overseeing Patent Prosecution and other matters contemplated by Article 12 and, if appropriate, delegating responsibility for such matters, subject to oversight by the Joint Management Committee, to a committee appointed by the Joint Management Committee for such purpose;

(k) evaluating potential licenses from Third Parties, and determining their utility in the Collaboration (if any);

(1) upon the recommendation of the Joint Development Committee, decisions with respect to the preclinical and clinical Development of Collaboration Products, including the pursuit of additional indications;

(m) developing and overseeing the implementation of a strategy for the commercialization of Collaboration Products;

(n) deciding whether a Collaboration Product should be commercialized by or through one or more of the parties or their respective Affiliates as contemplated by Section 8.2;

(o) deciding whether a Collaboration Product should be commercialized by or through a Joint Marketing/Development Collaborator as contemplated by Section 8.3, monitoring the progress of any negotiations conducted in accordance with Section 8.3 and approving the terms of any final agreement arising therefrom;

(p) evaluating the progress of the Joint Research Committee and Joint Development Committee, and on a quarterly basis at a minimum, evaluating the progress of the Biotherapeutics Research Plan and applicable Development Plan(s) against their respective timelines;

(q) overseeing the maintenance of an inventory of the assets generated pursuant to the Collaboration;

(r) resolving matters within the responsibilities of the Joint Research Committee and Joint Development Committee as to which the members of such Collaboration Committee are unable to reach a consensus, and dissolving each such Collaboration Committee when its duties under the Collaboration are complete;

(s) resolving disagreements between the parties with respect to the matters contemplated by Article 8 and 10 hereof; and

(t) addressing issues and resolving differences that may arise between the parties.

The Joint Management Committee shall not have the power to amend the terms of or waive compliance with this Agreement.

3.5 Responsibilities of Joint Research Committee. The Joint Research Committee shall be responsible for preparing for approval by the Joint Management Committee and implementing the applicable annual Biotherapeutics Research Plan, with the objective of expeditiously identifying Program Antibodies, Program Antisense Compounds and Program Proteins meeting the criteria for designation as Collaboration Products. Without limiting the generality of the foregoing, its duties shall include:

(a) overseeing the implementation of the Target Function Discovery Program;

(b) establishing criteria for the selection of Program Antibodies, Program Antisense Compounds and Program Proteins;

(c) selecting Program Antibodies, Program Antisense Compounds and Program Proteins for characterization and optimization in the conduct of the Collaboration;

(d) monitoring, reviewing and reporting on the progress of the Biotherapeutics Research Program;

(e) recommending Program Antibodies, Program Antisense Compounds and Program Proteins for designation by the Joint Management Committee as Collaboration Products; and

(f) performing such other activities as are contemplated by the terms of this Agreement.

The Joint Research Committee shall report its activities and make proposals to the Joint Management Committee at least [**], but more frequently as appropriate. The Joint Research Committee shall not have the power to amend or waive compliance with this Agreement.

3.6 Responsibilities of Joint Development Committee. The Joint Development Committee shall be responsible for preparing for approval by the Joint Management Committee and implementing the applicable annual [**] Development Plan(s). Without limiting the generality of the foregoing, its duties shall include:

(a) proposing and overseeing the Development strategy of Collaboration Products;

(b) overseeing the filing of INDs with the FDA or other Regulatory Authority by the designated party pursuant to Section 6.4.1;

(c) establishing advisory committees comprised of scientific, medical or other appropriate experts not affiliated with either party to advise the Joint Development Committee on matters related to the preclinical and clinical Development of Collaboration Products;

(d) approving the protocol of any clinical trials of Collaboration
Products;

(e) overseeing the clinical trials of Collaboration Products;

(f) providing all appropriate information regarding the progress of the Development Plan(s) to the Joint Management Committee in advance of each quarterly Joint Management Committee meeting;

(g) overseeing the filing of BLAs with the FDA or other Regulatory Authority by the designated party pursuant to Section 6.4.1; and

(h) $$performing\ such\ other\ activities\ as\ are\ contemplated\ by\ the\ terms\ of\ this\ Agreement.$

The Joint Development Committee shall report its activities and make proposals to the Joint Management Committee at least [**], but more frequently as appropriate. The Joint Development Committee shall not have the power to amend or waive compliance with this Agreement.

3.7 Meetings of Collaboration Committees. As applicable, each Collaboration Committee shall meet at least [**], and more frequently as the parties deem appropriate, on such dates and at such times as the parties shall agree, on [**] written notice to the other party unless such notice is waived by the parties. The first meeting of the Joint Management Committee shall take place within [**] after the Effective Date, at Lexicon's facility in The Woodlands, Texas, United States of America. Each Collaboration Committee may convene or be polled or consulted from time to time by means of telecommunications, videoconferences or correspondence, as deemed necessary or appropriate by the parties. To the extent that meetings are held in person, they shall alternate between the offices of the parties unless the parties otherwise agree.

3.8 Decisions.

3.8.1 Quorum; Voting. A quorum for a meeting of a Collaboration Committee shall require the presence of at least one Lexicon member (or designee) and at least one Organon member (or designee) in person or by telephone. All decisions made or actions taken by a Collaboration Committee shall be made unanimously by its members, with the Lexicon members cumulatively having one vote and the Organon members cumulatively having one vote; provided that, in the event the members of the Joint Research Committee are unable to reach unanimity as to a decision under [**] with respect to [**], then either party may, in its sole discretion, [**].

3.8.2 Dispute Resolution.

3.8.2.1 In the event that unanimity cannot be reached by the Joint Research Committee, or Joint Development Committee, as the case may be, with respect to a matter that is a subject of its decision-making authority, respectively, then the matter shall be referred for further review and resolution to the Joint Management Committee. In the event that unanimity cannot be reached by the Joint Management Committee with respect to a matter that is a subject of its decision-making authority, then the matter shall be referred for further review and resolution to the Chief Executive Officer of Organon, or such other similar position designated by Organon from time to time, and the Chief Executive Officer of Lexicon, or such other similar position designated by Lexicon from time to time. The designated officers of each party shall use reasonable efforts to resolve the matter within [**] after the matter is referred to them.

3.8.2.2 If the designated officers cannot resolve any matter pursuant to Section 3.8.2.1 within such [**] period, the matter shall be referred to a Third Party arbitrator or arbitrators, in accordance with the following procedures, whose decision shall be [**]. The parties shall attempt to mutually agree upon a single independent Third Party arbitrator (who shall be a professional with appropriate experience in the subject matter at issue in such disagreement) within [**] after the initial referral of such matter to the designated officers. If the parties are unable to mutually agree upon one such person, then each party shall appoint one independent Third Party professional with appropriate experience in the subject matter at issue in such disagreement prior to the expiration of such [**] period, and within [**] after the initial referral of such matter to the designated officers, such person(s) shall select a single independent Third Party arbitrator, who shall be a professional with appropriate experience in the subject matter at issue in such disagreement. Each party shall present all information presented pursuant to Section 3.8.2.1 and all other information as such party reasonably desires regarding such disagreement. Within [**] after the initial referral of such matter to the designated officers, the arbitrator shall provide written notice to the parties regarding his or her determination regarding such disagreement.

3.9 Administration. The chairperson of each Collaboration Committee shall be designated annually on an alternating basis between the parties. The initial chairperson shall be selected by Organon. The party not designating the chairperson shall designate one of its representative members as secretary to such Collaboration Committee for such year. The chairperson shall be responsible for calling meetings of such Collaboration Committee, sending notices of meetings to all members and for leading such meetings.

3.10 Minutes. Within [**] after each Collaboration Committee meeting, the secretary of such Collaboration Committee shall prepare and distribute minutes of the meeting, which shall provide a description in reasonable detail of the discussions had at the meeting and a list of any actions, decisions or determinations approved by such Collaboration Committee. The secretary shall be responsible for circulation of all draft and final minutes. Draft minutes shall be first circulated to the chairperson, edited by the chairperson and then circulated in final draft form to all members of such Collaboration Committee sufficiently in advance of the next meeting to allow adequate review and comment prior to the meeting. Minutes shall be approved or disapproved, and revised as necessary, at the next meeting. Final minutes shall be distributed to the members of such Collaboration Committee.

3.11 Term. The Joint Management Committee shall exist until the termination or expiration of the Collaboration Term. Each other Collaboration Committee shall exist until the termination or expiration of the Collaboration Term unless earlier dissolved by the Joint Management Committee following the completion of its duties under the Collaboration.

3.12 Expenses. Each party shall be responsible for all travel and related costs for its representatives to attend meetings of, and otherwise participate on, a Collaboration Committee.

ARTICLE 4. TARGET FUNCTION DISCOVERY PROGRAM

4.1 General.

4.1.1 Generation and Analysis of Mutant Mice. In the Target Function Discovery Program, Lexicon shall use Diligent Efforts to complete (a) the development and Level 1 Phenotypic Analysis of Mutant Mice for each Program Target and (b) Level 2 Phenotypic Analysis of such lines of Mutant Mice that displayed a phenotype suggestive, as determined by the Joint Research Committee, of the potential utility of the corresponding Program Target in the Collaboration Field.

4.1.2 Target Function Discovery Program Term. The Target Function Discovery Program shall continue until the end of the fourth year of the Collaboration Term on May 16, 2009 and all work under the Target Function Discovery Program shall be completed by that date (the "Target Function Discovery Program Term").

4.2 Annual Budget. Lexicon shall prepare and the Joint Management Committee shall approve the annual budget for the Target Function Discovery Program for every Contract Year (other than the First Contract Year) during the Target Function Discovery Program Term at least [**] prior to the commencement of such Contract Year. The annual budget for the Target Function Discovery Program for the First Contract Year shall be prepared by Lexicon and approved by the Joint Management

Committee within [**] after the Effective Date. Each such annual budget shall be in writing and shall set forth with reasonable specificity expected timelines for the development of Mutant Mice and the conduct of Level 1 Phenotypic Analysis and Level 2 Phenotypic Analysis, together with associated Target Function Discovery Program Costs. The Joint Research Committee may agree on modifications, and recommend that the Joint Management Committee approve such modifications, to the provisions of any such annual budget at any time.

4.3 Reporting and Oversight of Target Function Discovery Program Progress. Lexicon shall keep the Joint Research Committee fully informed of the progress of its activities under the Target Function Discovery Program. At a minimum, within [**] following the last day of each [**] during the Target Function Discovery Program Term, Lexicon shall prepare, and provide to the Joint Research Committee, a reasonably detailed written summary report which shall describe (a) the work performed by Lexicon during the preceding [**], including, without limitation, the status of Lexicon's development of Mutant Mice and the conduct of Level 1 Phenotypic Analysis and Level 2 Phenotypic Analysis (or only Level 1 Phenotypic Analysis if Level 2 Phenotypic Analysis has not been performed) of such Mutant Mice, and (b) identify phenotypes identified through such Level 1 Phenotypic Analysis and Level 2 Phenotypic Analysis that are suggestive, in Lexicon's good faith scientific judgment, of the potential utility of the corresponding Program Targets in the Collaboration Field. In addition, Lexicon shall provide the Joint Research Committee with access to all data, information and conclusions from such Level 1 Phenotypic Analysis and/or Level 2 Phenotypic Analysis of Mutant Mice, in each case promptly following the generation thereof, so as to enable the Joint Research Committee to make its own determinations as to which Mutant Mice exhibit a phenotype suggestive of the potential utility of the corresponding Program Target in the Collaboration Field.

4.4 Third Party Licenses. Notwithstanding anything herein to the contrary, but subject to the requirement of unanimously agreed budgets set forth in Section 4.2, Lexicon shall be responsible for the licensing of technologies or patents owned or controlled by Third Parties that are required or beneficial for the conduct of the Target Function Discovery Program. For the avoidance of doubt, no such technologies or patents shall be used by Lexicon in the Target Function Discovery Program unless the Joint Management Committee has agreed and consented to the financial implications of such use.

ARTICLE 5. BIOTHERAPEUTICS RESEARCH PROGRAM

5.1 General. The parties shall jointly pursue Biotherapeutics Research relating to Program Targets that have not become Opt-out Targets under the direction of the Joint Research Committee in accordance with annual Biotherapeutics Research Plans. Unless otherwise agreed by the Joint Management Committee, (a) Lexicon will be principally, but not exclusively, responsible for conducting Biotherapeutics Research activities involving the generation of research Program Antibodies, Program Antisense Compounds and Program Proteins and the conduct of in vivo research and proof of concept studies (although Organon may also be involved in such activities) and (b) Organon will be responsible for conducting Biotherapeutics Research activities involving the generation of human or humanized Program Antibodies and the process development of Program Antibodies, Program Antisense Compounds and Program Proteins in preparation for Development.

5.2 Biotherapeutics Research Plans.

5.2.1 The Joint Research Committee shall prepare and the Joint Management Committee shall approve the Biotherapeutics Research Plan for every Contract Year (other than the First Contract Year) during the Collaboration Term at least [**] prior to the commencement of such Contract Year. The Biotherapeutics Research Plan for the First Contract Year shall be prepared by the Joint Research Committee and approved by the Joint Management Committee

within [**] after the Effective Date. The responsibility of the Joint Research Committee to prepare annual Biotherapeutics Research Plans shall terminate upon the agreement of the parties to cease further Biotherapeutics Research regarding Program Targets.

Each annual Biotherapeutics Research Plan shall be in 5.2.2 writing and shall set forth with reasonable specificity the Biotherapeutics Research objectives, priorities, activities, milestones, budgets, personnel requirements, other resources and allocations of responsibilities between the parties for the period covered by such annual Biotherapeutics Research Plan in a manner consistent with the terms of this Agreement, including, without limitation, the objectives set forth in Section 2.1.1 and the terms and conditions set forth in Section 2.2. The Biotherapeutics Research Plans shall cover all aspects of Biotherapeutics Research relating to the generation of Antibodies, Antisense Compounds and Proteins relating to Program Targets that have not become Opt-out Targets and the identification, characterization, selection, optimization and research of Program Antibodies, Program Antisense Compounds and Program Proteins prior to their designation as Collaboration Products, and shall include, with reasonable specificity, the Biotherapeutics Research activities to be performed by each party and the Biotherapeutics Research activities, if any, to be performed by subcontractors.

5.2.3 The Joint Research Committee may agree on modifications, and recommend that the Joint Management Committee approve such modifications, to the provisions of any Biotherapeutics Research Plan at any time. Without limiting the foregoing, the Joint Research Committee shall conduct a [**] review of each Biotherapeutics Research Plan and shall recommend that the Joint Management Committee approve such modifications to the applicable Biotherapeutics Research Plan as the Joint Research Committee may deem to be appropriate as a result of such review.

ARTICLE 6. DEVELOPMENT PROGRAMS

6.1 General. The parties shall jointly pursue the Development of Collaboration Products in the Collaboration Field under the direction of the Joint Development Committee in accordance with annual [**] Development Plans. Notwithstanding the foregoing, the Joint Development Committee shall consider, and make recommendations to the Joint Management Committee regarding all other commercially reasonable arrangements for the Development of Collaboration Products, including proposals from one or both of the parties and proposals from one or more Third Parties. At its option, each party may designate an Affiliated Subcontractor to carry out its responsibilities under the Development Plan(s) relating to any Collaboration Product, on a Collaboration Product-by-Collaboration Product basis.

6.2 Designation of Collaboration Products. The Joint Management Committee shall adopt criteria for the designation of Program Antibodies, Program Antisense Compounds and Program Proteins as Collaboration Products, which criteria shall include, without limitation, (a) identification and characterization by the parties of a chimeric, humanized or human Program Antibody, a Program Antisense Compound or a human Program Protein, as applicable, that is suitable for the initiation of preclinical studies designed to support the filing of an IND; and (b) achievement of appropriate results in proof of concept studies conducted in the Biotherapeutics Research Program. The Joint Management Committee shall be responsible for designating Collaboration Products in accordance with Section 3.4.

6.3 Development Plans.

6.3.1 Following the first Contract Year in which the parties designate a Collaboration Product, the Joint Development Committee shall prepare and the Joint Management Committee shall approve a Development Plan for each Collaboration Product for every Contract Year at least [**] prior to the commencement of such Contract Year. In the first Contract Year in which the parties designate a Collaboration Product, the Joint Development Committee shall prepare the first Development Plan for such Collaboration Product as soon as commercially reasonable after its designation. The responsibility of the Joint Development Committee to prepare a Development Plan for a Collaboration Product shall terminate upon the earlier of (a) a party Opting Out of further Development of such Collaboration Product or (b) the agreement of the parties to cease further Development of such Collaboration Product.

6.3.2 Each Development Plan shall be in writing and shall set forth with reasonable specificity the Development objectives, priorities, activities, milestones, budgets, personnel requirements, other resources and allocations of responsibilities between the parties for the period covered by such Development Plan in a manner consistent with the terms of this Agreement, including, without limitation, the objectives set forth in Section 2.1.1 and the terms and conditions set forth in Section 2.2. The Development Plans shall cover all aspects of Development relating to Collaboration Products, and shall include, with reasonable specificity, the Development activities to be performed by each party and the Development activities, if any, to be performed by subcontractors.

6.3.3 The Joint Development Committee may agree on modifications, and recommend that the Joint Management Committee approve such modifications, to the provisions of any Development Plan at any time.

6.4 Regulatory Matters.

6.4.1 Regulatory Responsibility. The preparation, filing, prosecution and maintenance of INDs, BLAs and other regulatory filings required to be filed with any Regulatory Authority with regard to each Collaboration Product will be in the name of and the responsibility of the party so designated in the Development Plan covering such Collaboration Product. The party so designated in the Development Plan covering such Collaboration Product shall oversee, monitor and coordinate all regulatory actions, communications and filings with and submissions, including filings and submissions of supplements and amendments thereto, to Regulatory Authorities with respect to each Collaboration Product, shall give the other party a reasonable opportunity for prior review of and comment on all such substantive communications, filings and submissions and shall incorporate those of such comments as can reasonably be incorporated into such communications, filings and submissions.

6.4.2 Regulatory Meetings and Correspondence. The party so designated in the Development Plan covering a particular Collaboration Product shall be responsible for interfacing, corresponding and meeting with Regulatory Authorities with respect to such Collaboration Product, and the other party will promptly refer any contacts or questions from Regulatory Authorities to the party so designated. Both parties will be entitled to attend all meetings and, if reasonably practicable, telephone conferences with Regulatory Authorities.

6.4.3 Reporting Adverse Drug Reactions. The parties will develop and agree upon safety data exchange procedures governing the collection, investigation, reporting, and exchange of information concerning Adverse Drug Reactions, product quality and product complaints involving Adverse Drug Reactions, sufficient to permit each party to comply with its legal obligations, including to the extent applicable, those obligations contained in ICH guidelines E2A, E2B and E2C and the FDC Act. The safety data exchange procedures will be promptly updated if required by changes in the Law or by agreement between the parties. The party so designated in the applicable Development Plan will be responsible for reporting all Adverse Drug Reactions to the appropriate Regulatory Authorities in the applicable Region(s) in accordance with applicable Laws.

ARTICLE 7. MANUFACTURING AND SUPPLY

7.1 Designation of Manufacturing Party.

7.1.1 Manufacture of Collaboration Products by Organon.

7.1.1.1 Manufacture of Collaboration Products for Development. Unless otherwise agreed by the Joint Management Committee, Organon or its Affiliated Subcontractor(s) shall Manufacture and supply all quantities of a Collaboration Product necessary for Development; provided that Organon may decline to Manufacture and supply such quantities of a Collaboration Product for any reason by giving Lexicon written notice to such effect no later than [**] after the Joint Management Committee's designation of such Collaboration Product in accordance with Section 3.4.

7.1.1.2 Manufacture of Collaboration Products for Commercialization. Unless otherwise agreed by the Joint Management Committee within [**] after the commencement of the first Phase 2 Clinical Trial with respect to any Collaboration Product, the parties shall enter into a Manufacturing and Supply Agreement substantially in the form attached to this Agreement as Exhibit 7.1.1.2 under which Organon or its Affiliated Subcontractor(s) shall be responsible for the Manufacture and supply all quantities of such Collaboration Product necessary for commercialization; provided that Organon may decline to enter into such agreement for any reason by giving Lexicon written notice to such effect no later than [**] after the expiration of such [**] period.

Manufacture of Collaboration Products by Third 7.1.2 Parties. In the event that the Joint Management Committee determines that it is in the best interests of the Collaboration for any one or more Manufacturing activities with respect to a Collaboration Product to be undertaken by an identified Third Party (whether as a result of a decision by Organon in accordance with Section 7.1.1 not to undertake such Manufacturing activities or otherwise), the negotiations with such Third Party relating to such Manufacturing activities shall be led by [**]; provided that [**] (a) shall keep [**] informed of the substance and status of such negotiations, and allow [**] to participate in such negotiations and (b) shall take into account in such negotiations the reasonable commercial interests of [**] and the best interests of the Collaboration. The parties each agree to grant such (sub)licenses under their respective Background Materials, Background Technology, Program Materials and Program Technology to any such Third Party as may be reasonably necessary for such Third Party to manufacture the Collaboration Product.

7.1.3 Manufacture of Lexicon Products.

7.1.3.1 Manufacture of Lexicon Products for Development. If requested by Organon within [**] following Organon's Opting Out with respect to a Collaboration Product Manufactured by Organon or its Affiliated Subcontractor(s) for Development pursuant to Section 7.1.1.1, Lexicon shall enter into good faith negotiations with Organon, for a period of [**] following such request, with respect to the continued manufacture of such Lexicon Product by Organon or its Affiliated Subcontractor(s) for Development. In the event Lexicon and Organon or its Affiliated Subcontractor(s) do not enter into a definitive agreement with respect to such manufacturing activities within such [**] period, Lexicon will be free, at any time thereafter, to enter into negotiations and agreements with one or more Third Parties for the manufacture and supply of such Lexicon Product. Nothing in this Section 7.1.3.1 shall be deemed to affect the rights and obligations of the parties under Section 10.3.3.

7.1.3.2 Manufacture of Lexicon Products for Commercialization. Within h [**] after the commencement of the first Phase 2 Clinical Trial with respect to any Lexicon Product which is manufactured for Development by Organon or its Affiliated Subcontractor(s) pursuant to Section 7.1.3.1, Lexicon shall offer to Organon the opportunity to manufacture such Lexicon Product for commercialization. If requested by Organon within [**] of such offer, Lexicon shall enter into good faith negotiations with Organon, for a period of [**] following such request, with respect to the continued manufacture of such Lexicon Product by Organon or its Affiliated Subcontractor(s) for commercialization. In the event Lexicon and Organon or its Affiliated Subcontractor(s) do not enter into a definitive agreement with respect to such manufacturing activities within such [**] period, Lexicon will be free, at any time thereafter, to enter into negotiations and agreements with one or more Third Parties for the manufacture and supply of such Lexicon Product [** *]. Nothing in this Section 7.1.3.2 shall be deemed to affect the rights and obligations of the parties under Section 10.3.3.

7.2 Manufacturing Plans. With respect to all Collaboration Products that Organon or its Affiliated Subcontractor(s) Manufactures pursuant to this Article 7, Organon shall be responsible for implementing all aspects of Manufacturing necessary for Development under the direction and oversight of the Joint Management Committee, as set forth in Section 3.4, and in accordance with a manufacturing plan for the applicable Collaboration Product(s) in the Collaboration Field proposed by Organon and subject to review and approval by the Joint Management Committee, which manufacturing plan shall describe the specific Manufacturing activities to be undertaken by Organon, shall include a general description of the personnel and other resources to be used in the implementation thereof and shall set forth a unanimously agreed budget for such activities (each, as may be modified or amended and approved from time to time in accordance with this Agreement, a "Manufacturing Plan").

Orders; Forecasts for Clinical Requirements. If Organon or its 7.3 Affiliated Subcontractor(s) is the designated manufacturer for a particular Collaboration Product as provided in Section 7.1.1, the party primarily responsible for running a particular clinical trial hereunder will be responsible for generating periodic [**] forecasts of the anticipated requirements for such Collaboration Product that is the subject of such trial and updates of such forecast not less than [**] thereafter, such forecasts and updates to be promptly provided to Organon or its Affiliated Subcontractor. Not less than [**] prior to the required delivery of a specified quantity of Collaboration Product for such purposes, the Joint Development Committee shall meet and agree on a demand order for the Collaboration Product so required; provided, that if the total amount of any demand order for delivery in any [**] period exceeds [**] of the most recent forecast for such period, Organon shall use Diligent Efforts, but shall have no obligation, to deliver the quantities in excess of [**] of the estimated amount for such period. Organon shall ship the Collaboration Product to the facility or facilities designated by the party with responsibility for distribution of clinical requirements of the Collaboration Product at the times set forth in the relevant demand order. For clarity, this paragraph shall not apply in the event that Organon or its Affiliated Subcontractor is not the designated manufacturer for such Collaboration Product.

7.4 Certain Covenants. Organon and its Affiliated Subcontractor(s) agree and covenant that, in the event that it Manufactures a Collaboration Product pursuant to Section 7.1.1, it will (a) use Diligent Efforts to avoid shortfalls of supply based on the forecasts provided to it in accordance with Section 7.3,

shall promptly notify the parties in the event it becomes aware of any probable shortfall and shall use Diligent Efforts to remedy any shortfall of supply as soon as practicable; (b) be responsible for Manufacturing, filling, packaging and warehousing of the Collaboration Product in conformity with applicable cGMP Requirements [**], and in accordance, in all material respects, with all other applicable Law; (c) maintain or cause to be maintained all records necessary and appropriate to demonstrate compliance with applicable cGMP Requirements and the applicable Specifications; and (d) grant Lexicon the right, on reasonable advance notice and during normal business hours during the term of this Agreement, to have its personnel or representatives with quality control or quality assurance responsibilities inspect and audit the facilities and operations directly related to the Manufacture and supply of the Collaboration Product in order to confirm compliance with the covenants contained in this Section 7.4; provided that the foregoing inspection and audit right shall be limited to [**] and [**] per visit; and provided, further, that such personnel or representatives shall be subject to Organon's prior approval, such approval not to be unreasonably withheld.

7.5 Third Party Licenses. Notwithstanding anything herein to the contrary but subject to the requirement of unanimously agreed budgets set forth in Section 7.2, in the event that Organon or its Affiliated Subcontractor(s) Manufactures a Collaboration Product, Organon shall be responsible for the licensing of technologies or patents owned or controlled by Third Parties that are required or beneficial for the Manufacturing of such Collaboration Product. For the avoidance of doubt, no such technologies or patents shall be used by Organon in the Development or Manufacturing of any Product unless the Joint Management Committee has agreed and consented to the financial implications of such use.

ARTICLE 8. COMMERCIALIZATION

8.1 General. Subject to the provisions of this Article 8, the Joint Management Committee shall consider all commercially reasonable arrangements for the commercialization of Collaboration Products, including proposals from one or both of the parties and proposals from one or more Third Parties.

8.2 Commercialization by the Parties or Their Affiliates. In the event the Joint Management Committee determines that a particular Collaboration Product should be commercialized entirely by or through one or more of the parties or their respective Affiliates throughout the Territory or in one or more particular countries or regions, then subject to the parties' entry, in their respective sole discretion, into a definitive agreement with respect thereto with the applicable party, parties or Affiliate(s):

> (a) the parties shall grant to each such applicable party, parties or Affiliate(s) appropriate rights and licenses under their respective rights in (i) the Lexicon Background Materials, the Lexicon Background Technology, the Organon Background Materials and the Organon Background Technology, including, without limitation, any Patent Rights Controlled by either party Covering the foregoing, and (ii) the Program Intellectual Property, in each case to Develop, make, have made, import, use, have used, offer for sale, sell and have sold Collaboration Products in the Collaboration Field throughout the Territory or in such country, countries or region(s), as applicable; and

(b) each party shall take, or cause to be taken, all actions reasonably necessary to consummate and make effective any such agreement with such party, parties or Affiliate(s).

8.3 Commercialization by Joint Marketing/Development Collaborators. In the event the Joint Management Committee determines that a particular Collaboration Product should be commercialized entirely by or through one or more Third Parties (each, a "Joint Marketing/Development Collaborator") throughout the Territory or in one or more particular countries or regions, then subject to the parties' entry, in their respective sole discretion, into a definitive agreement with respect thereto with the applicable Joint Marketing/Development Collaborator(s):

(a) the parties shall grant to each such Joint Marketing/Development Collaborator(s) appropriate rights and licenses under their respective rights in (i) the Lexicon Background Materials, the Lexicon Background Technology, the Organon Background Materials and the Organon Background Technology, including, without limitation, any Patent Rights Controlled by either party Covering the foregoing, and (ii) the Program Intellectual Property, in each case to Develop, make, have made, import, use, have used, offer for sale, sell and have sold Collaboration Products in the Collaboration Field throughout the Territory or in such country, countries or region(s), as applicable; and

(b) each party shall take, or cause to be taken, all actions reasonably necessary to consummate and make effective any such agreement with such Joint Marketing/Development Collaborator.

Option to Co-Commercialize Collaboration Products. In the event that, within [**], there remain one or more countries with respect to which the parties have not entered into a definitive agreement pursuant to Section 8.2 or 8.3 for the commercialization of a Collaboration Product in the Collaboration Field (or, if applicable, if the parties have not entered into a definitive agreement pursuant to Section 8.2 or 8.3 with respect to the commercialization of a Collaboration Product in the Collaboration Field in any country within the Territory), either party may provide the other party with written notice of its offer to enter into an agreement in the form attached hereto as Exhibit 8.4 for the joint commercialization of such Collaboration Product in the Collaboration Field in all countries within the Territory for which the parties have not entered a definitive agreement pursuant to Section 8.2 or 8.3. The other party shall have the right, within the period of [**] following such notice, to accept such offer and enter into such joint commercialization agreement. In the event the other party fails to accept such offer and enter into such joint commercialization agreement within such [**] period, such other party shall be deemed to have Opted Out with respect to such Collaboration Product pursuant to Section 10.1.1 or 10.1.2, as applicable.

8.5 Notice of Adverse Reactions. Each party shall advise the other as promptly as reasonably practical by facsimile or overnight delivery service addressed to the attention of its Vice President, Regulatory Affairs (or equivalent), of any Adverse Drug Reaction that has been brought to that party's attention.

8.6 Product Recall. In the event that either party determines that an event, incident or circumstance has occurred that may result in the need for a recall or other removal of any Collaboration Product, or any lot or lots thereof, from a market in any Region, it shall advise and consult with the other party with respect thereto. The owner of the relevant Regulatory Approval (or proposed Regulatory Approval), determined in accordance with Section 6.4, shall make the final determination to recall or otherwise remove the Collaboration Product or any lot or lots thereof from the market.

ARTICLE 9. GRANTS OF RIGHTS

9.1 Grants of Research Licenses.

9.1.1 By Lexicon. Subject to the terms of this Agreement and any applicable [**], during the Collaboration Term, Lexicon hereby grants to Organon and its Affiliates, within the Territory, (a) a non-exclusive right and license (without any right to sublicense, except as set forth below) under Lexicon's rights in the Lexicon Background Materials and the Lexicon

Background Technology, including, without limitation, any Patent Rights Controlled by Lexicon Covering the foregoing, and (b) a co-exclusive right and license (without any right to sublicense, except as set forth below) under Lexicon's rights in the Program Intellectual Property, in each case to (i) generate Antibodies, Antisense Compounds and Proteins relating to Program Targets that have not become Lexicon Opt-out Targets and (ii) identify, characterize, select, optimize and research Program Antibodies, Program Targets prior to their designation as Collaboration Products, in each case in the conduct of the Collaboration. Such right and license shall include the right to grant sublicenses to Affiliates of Organon and to Third Parties that are approved by the Joint Management Committee.

By Organon. Subject to the terms of this Agreement 9.1.2 and any applicable [**], during the Collaboration Term, Organon hereby grants to Lexicon and its Affiliates, within the Territory, (a) a non-exclusive right and license (without any right to sublicense, except as set forth below) under Organon's rights in the Organon Background Materials and the Organon Background Technology, including, without limitation, any Patent Rights Controlled by Organon Covering the foregoing, and (b) a co-exclusive right and license (without any right to sublicense, except as set forth below) under Organon's rights in the Program Intellectual Property, in each case to (i) generate Antibodies, Antisense Compounds and Proteins relating to Program Targets that have not become Organon Opt-out Targets and (ii) identify, characterize, select, optimize and research Program Antibodies, Program Antisense Compounds and Program Proteins relating to such Program Targets prior to their designation as Collaboration Products, in each case in the conduct of the Collaboration. Such right and license shall include the right to grant sublicenses to Affiliates of Lexicon and to Third Parties that are approved by the Joint Management Committee.

9.1.3 Restrictions on Clinical Development of Collaboration Products. Neither party nor their respective Affiliates shall administer to humans any Program Antibody, Program Antisense Compound, Program Protein or Collaboration Product or Opt-out Product that incorporates or is derived from any Program Antibody, Program Antisense Compound or Program Protein, unless and until (and then only to the extent that) such party has received a license under Section 9.2 for the clinical Development of such Collaboration Product, Opt-out Product, Program Antibody, Program Antisense Compound or Program Protein.

9.2 Grants of Development and Commercialization Licenses.

9.2.1 By Lexicon.

9.2.1.1 Development of Collaboration Products in Collaboration Field. Subject to the terms of this Agreement and any applicable [**], Lexicon hereby grants to Organon and its Affiliates, within the Territory, a co-exclusive right and license, with the limited right to sublicense (as set forth below), under Lexicon's rights in (a) the Lexicon Background Materials and the Lexicon Background Technology, including, without limitation, any Patent Rights Controlled by Lexicon Covering the foregoing, and (b) the Program Intellectual Property, in each case to Develop Collaboration Products in the Collaboration Field, including rights to make, have made, import, use, or have used any Collaboration Products in the Collaboration Field solely for purposes of such Development. Such right and license shall include the right to grant sublicenses to Affiliates of Organon and to Third Parties that are approved by the Joint Management Committee.

9.2.1.2 Development and Commercialization of Organon Products in Collaboration Field. Subject to the terms of this Agreement and any applicable [**], Lexicon hereby grants to Organon and its Affiliates, within the Territory, an exclusive right and license, with the right to sublicense, under Lexicon's rights in (a) the Lexicon Background Materials and the Lexicon Background Technology, including, without limitation, any Patent Rights Controlled by Lexicon Covering the foregoing, and (b) the Program Intellectual Property to Develop, make, have made, import, use, have used, offer for sale, sell and have sold Organon Products (and, except for Collaboration Products relating to the same Program Target as to which Lexicon has not Opted Out, any Program Antibodies, Program Antisense Compounds and Program Proteins relating to the foregoing) in the Collaboration Field. Any such sublicense shall be set forth in a written agreement containing confidentiality, non-use, ownership of intellectual property and audit provisions consistent with and no less restrictive than those contained herein, shall be subject and subordinate to the terms and conditions of this Agreement, and shall obligate the (sub)licensee to make the royalty payments required hereunder; provided that Organon shall remain responsible for all payments due to Lexicon hereunder. Organon shall provide Lexicon with a copy of each sublicense agreement promptly after executing the same; provided, however, that subject to the exceptions set forth in Section 1.23, each such sublicense agreement shall be Confidential Information of Organon.

9.2.2 By Organon.

9.2.2.1 Development of Collaboration Products in Collaboration Field. Subject to the terms of this Agreement and any applicable [**], Organon hereby grants to Lexicon and its Affiliates, within the Territory, a co-exclusive right and license, with the limited right to sublicense (as set forth below), under Organon's rights in (a) the Organon Background Materials and the Organon Background Technology, including, without limitation, any Patent Rights Controlled by Organon Covering the foregoing, and (b) the Program Intellectual Property, in each case to Develop Collaboration Products in the Collaboration Field, including rights to make, have made, import, use, or have used any Collaboration Products in the Collaboration Field solely for purposes of such Development. Such right and license shall include the right to grant sublicenses to Affiliates of Lexicon and to Third Parties that are approved by the Joint Management Committee.

9.2.2.2 Development and Commercialization of Lexicon Products in Collaboration Field. Subject to the terms of this Agreement and any applicable [**], Organon hereby grants to Lexicon and its Affiliates, within the Territory, an exclusive right and license, with the right to sublicense, under Organon's rights in (a) the Organon Background Materials and the Organon Background Technology, including, without limitation, any Patent Rights Controlled by Organon Covering the foregoing, and (b) the Program Intellectual Property to Develop, make, have made, import, use, have used, offer for sale, sell and have sold Lexicon Products (and, except for Collaboration Products relating to the same Program Target as to which Organon has not Opted Out, any Program Antibodies, Program Antisense Compounds and Program Proteins relating to the foregoing) in the Collaboration Field. Any such sublicense shall be set forth in a written agreement containing confidentiality, non-use, ownership of intellectual property and audit provisions consistent with and no less restrictive than those contained herein, shall be subject and subordinate to the terms and conditions of this Agreement, and shall obligate the (sub)licensee to make the royalty payments required hereunder; provided that Lexicon shall remain responsible for all payments due to Organon hereunder. Lexicon

shall provide Organon with a copy of each sublicense agreement promptly after executing the same; provided, however, that subject to the exceptions set forth in Section 1.23, each such sublicense agreement shall be Confidential Information of Lexicon.

9.2.3 Restrictions on Commercialization of Collaboration Products. Neither party nor their respective Affiliates (a) shall make, have made, import, use, or have used any Collaboration Products in the Collaboration Field for any purpose other than Development or (b) shall offer for sale, sell or have sold Collaboration Products in the Collaboration Field, unless and until (and then only to the extent that) such party has received a license in accordance with Section 8.2 or 8.4 for the commercialization of such Collaboration Product.

9.3 Small Molecule Research and Development.

Option to Collaborate in Oncology/Immunology Field. 9.3.1 Subject to the terms of this Agreement and any applicable [**], in the event that either party believes that (a) [**] and (b) [**], such party may, at any time during the Target Function Discovery Program Term before Opting Out with respect to such Program Target, notify the other party of its desire to enter into a collaboration for the research, development and commercialization of Small Molecule Compounds modulating such Program Target in the Oncology/Immunology Field. Promptly following such notice, Organon and Lexicon shall enter into a definitive agreement for the research, development and commercialization of Small Molecule Compounds modulating such Program Target in the Oncology/Immunology Field on substantially the same terms applicable to the research, development and commercialization of Collaboration Products under this Agreement. In such case, any Small Molecule Compound selected for Development in accordance with the terms of such definitive agreement that is not a Back-up Product under the terms of such definitive agreement shall be deemed to be a Collaboration Product for purposes of Sections 11.3 and 11.4 if (y) [**] or (z) [**].

9.3.2 Option to Collaborate Outside the Oncology/Immunology Field. Subject to the terms of this Agreement, any applicable [**], in the event that either party believes that (a) [**] and (b) [** '], such party may, at any time during the Target Function Discovery Program Term before Opting Out with respect to such Program Target, notify the other party of its desire to enter into a collaboration for the research, development and commercialization of Small Molecule Compounds modulating such Program Target in the Collaboration Field outside the Oncology/Immunology Field. Promptly following such notice, Organon and Lexicon shall enter into a definitive agreement for the research, development and commercialization of Small Molecule Compounds modulating such Program Target in the Collaboration Field outside the Oncology/Immunology Field on substantially the same terms applicable to the research, development and commercialization of Collaboration Products under this Agreement. In such case, any Small Molecule Compound selected for Development in accordance with the terms of such definitive agreement that is not a Back-up Product under the terms of such definitive agreement shall be deemed to be a Collaboration Product for purposes of Sections 11.3 and 11.4 if (y) [**] or (z) [**]. Nothing herein shall obligate Lexicon to provide any notice to or enter into any agreement with Organon under this Section 9.3.2 with respect to a collaboration for the research, development and commercialization of Small Molecule Compounds modulating a Program Target in the Collaboration Field outside the Oncology/Immunology Field in the event Lexicon has any obligations to a Third Party with respect thereto at the time it proposes to commence such activities.

9.3.3 Organon License to Lexicon Outside the Oncology/Immunology Field. In the event that any applicable [**] that either party believes that (a) [**] and (b) [**] would preclude

Lexicon from entering into a collaboration for the research, development and commercialization of Small Molecule Compounds modulating such Program Target in the Collaboration Field outside the Oncology/Immunology Field pursuant to Section 9.3.2, or if Organon elects not to enter into a definitive agreement with respect thereto, then Organon shall grant to Lexicon and its Affiliates, within the Territory, an exclusive right and license, with the right to sublicense, under Organon's rights in Program Intellectual Property, if any, Covering the research, development or commercialization of Small Molecule Compounds modulating such Program Target, to research, develop, make, have made, import, use, have used, offer for sale, sell and have sold Small Molecule Compounds that modulate such Program Target in all fields of use other than the Oncology/Immunology Field. Nothing in this Section 9.3.3 shall be deemed to grant Lexicon or its Affiliates any right or license under any Patent Rights or other intellectual property rights Controlled by Organon in or to (a) any inventions, information, methods, know-how, trade secrets or data relating to Small Molecule Compounds developed in collaboration with Lexicon in accordance with Section 9.3.1 or 9.3.2 or (b) any other inventions, information, methods, know-how, trade secrets or data other than Program Intellectual Property.

Right of First Negotiation for Development and 9.4 Commercialization of Collaboration Products and Opt-out Products in the Veterinary Field. Prior to commencing development or commercialization of any Collaboration Product or Opt-out Product in the Veterinary Field or granting any license or otherwise transferring any rights to any Third Party with respect thereto, the Joint Management Committee (or the Continuing Party, in the case of an Opt-out Product) shall first offer to Intervet (or its designated Affiliate) the opportunity to develop and commercialize such Collaboration Product or Opt-out Product in the Veterinary Field. If requested by Intervet within [**] of such offer, the Joint Management Committee (or the Continuing Party, in the case of an Opt-out Product) shall enter into good faith negotiations with Intervet, for a period of [**] following such request, with respect to such an agreement for the development and commercialization of such Collaboration Product or Opt-out Product in the Veterinary Field. In the event Lexicon and Organon (or the Continuing Party, in the case of an Opt-out Product) and Intervet do not enter into a definitive agreement with respect to such an agreement within such [**] period, the Joint Management Committee (or the Continuing Party, in the case of an Opt-out Product) will be free, at any time thereafter, to enter into negotiations and agreements with one or more Third Parties for the development or commercialization of such Collaboration Product or Opt-out Product in the Veterinary Field [**]. All license fees, royalties, milestone payments and other income or items of value received from Intervet, its Affiliates or a Third Party with respect to Collaboration Products shall be considered Joint Marketing/Developer Collaboration Revenues under this Agreement.

9.5 No Grant of Other Technology or Patent Rights. Except as otherwise expressly provided in this Agreement, under no circumstances shall a party hereto, as a result of this Agreement, obtain any ownership interest in or other right to any technology, know-how, patents, patent applications, gene or genomic sequence data or information, products, or biological materials of the other party, including items owned, controlled or developed by, or licensed to, the other party, or transferred by the other party to said party, at any time pursuant to this Agreement.

ARTICLE 10. OPT-OUT PROVISIONS

10.1 Opt Out of Program Targets and Collaboration Products.

10.1.1 Right to Opt Out. Either party (the "Opt-out Party") may decline to participate in ("Opt Out" of) any further (a) Biotherapeutics Research with respect to a Program Target, on a Program Target-by-Program Target basis, or (b) Development of a Collaboration Product, on a Collaboration Product-by-Collaboration Product basis, by providing [**] prior written notice to the other party (the "Continuing Party") specifying the relevant provision below under which such Opt-out Party is making such Opt Out election and, as applicable, the Program Target(s) and Collaboration Product(s) as to which such Opt Out election relates.

10.1.1.1 Opt Out of All Biotherapeutics Research, Development and Commercialization Relating to a Program Target. The Opt-out Party may elect to Opt Out of all further Biotherapeutics Research and Development associated with a Program Target and related Collaboration Products, Program Antibodies, Program Antisense Compounds and Program Proteins. In such event, following the end of the [**] notice period specified above, such Program Target and any related Collaboration Products shall become an Opt-out Target and Opt-out Product(s), respectively, and the Continuing Party shall have exclusive (even as to the Opt-out Party) licenses under Section 9.2 with respect to such Program Target and all Opt-out Products, Program Antibodies, Program Antisense Compounds and Program Proteins relating thereto.

10.1.1.2 Opt Out of Biotherapeutics Research and Development of Follow-on Collaboration Products. The Opt-out Party may elect to continue to participate in the Development of Collaboration Product(s) that are at a more advanced stage of Development or that have been commercialized, while Opting Out of all further Biotherapeutics Research, Development associated with a Program Target and related Collaboration Products, Program Antibodies, Program Antisense Compounds and Program Proteins at an earlier stage of Development. In such event, following the end of the [**] notice period specified above, the Continuing Party shall have exclusive (even as to the Opt-out Party) licenses under Section 9.2 with respect to such Program Target and all Collaboration Products, Program Antibodies, Program Antisense Compounds and Program Proteins relating thereto other than the Collaboration Product(s) for which the Opt-out Party continues to participate.

10.1.2 Deemed Opt Out. In the event that a party desires to proceed with (a)(i) Biotherapeutics Research with respect to a Program Target or (ii) designation as a Collaboration Product and the commencement of Development of a Program Antibody, Program Antisense Compound or Program Protein with respect to a Program Target, in each case for which no material Biotherapeutics Research activity has been authorized by the Joint Management Committee (in a Biotherapeutics Research Plan or otherwise) for a period of [**] and for which no Collaboration Product is then in active Development, or (b) Development of a Collaboration Product (i) for which no material Development activity has been authorized by the Joint Management Committee (in a Development Plan or otherwise) for a period of [**] and (ii) relating to a Program Target for which no material Biotherapeutics Research activity has been authorized by the Joint Management Committee (in a Biotherapeutics Research Plan or otherwise) for a period of [**] and for which no other Collaboration Product is then in active Development, such party may request that the Joint Management Committee authorize such Biotherapeutics Research, designation as a Collaboration Product and the commencement of Development of such Program Antibody, Program Antisense Compound or Program Protein, or Development of such Collaboration Product, as applicable. In the event the Joint Management Committee fails to approve such request within [**] (otherwise than as a result of the requesting party's failure to vote in favor of authorizing such activities), then, effective upon such failure, the other party shall be deemed to have Opted Out of such Program Target and related Opt-out Products, Program Antibodies, Program Antisense Compounds and Program Proteins with the same effect as if such other party had Opted Out under Section 10.1.1.1.

10.2 Opt-out Party Noncompetition Covenant. For a period of [**] from the date the Opt-out Party gives the Continuing Party a written notice under Section 10.1.1 (or, if applicable, is deemed to

have Opted Out under Section 10.1.2), such Opt-out Party and its Affiliates shall not research, develop or commercialize (a) any Program Antibody, Program Antisense Compound or other Antibody or Antisense Compound acting through such Opt-out Target or (b) any Program Protein or other Protein encoded by such Opt-out Targets, except for the specific Collaboration Product(s), if any, for which the Opt-out Party retains Development or commercialization rights and obligations hereunder. For clarity, the expiration of such Opt-out Party's obligations under this Section 10.2 shall not be deemed to grant any right or license to such Opt-out Party.

10.3 Certain Effects of Opting Out.

10.3.1 Opt-out Targets Generally. Following the expiration of the [**] notice period specified in Section 10.1.1 with respect to any Opt-out Target (or, if applicable, of the [**] request period under Section 10.1.2), (a) the Biotherapeutics Research of such Opt-out Target shall no longer be carried out jointly by the parties or as part of the Collaboration; (b) the Party that has Opted Out shall have no rights to develop or commercialize, or to participate in Operating Profits from the commercialization of, any Antibodies or Antisense Compounds that act through or Proteins that are encoded by such Opt-out Target; and (c) the Continuing Party shall be free to conduct research activities relating to such Opt-out Target and develop and commercialize any Antibodies or Antisense Compounds that act through or Proteins that are encoded by such Opt-out Target, either alone or with one or more Third Parties. For clarity, the research activities with respect to any Opt-out Target shall not be subject to the oversight or direction of the Committees and shall not be the subject of any Biotherapeutics Research Plans.

10.3.2 Opt-out Products Generally. Following the expiration of the [**] notice period specified in Section 10.1.1 with respect to any Opt-out Product (or, if applicable, of the [**] request period under Section 10.1.2), (a) the Development and commercialization of such Opt-out Product shall no longer be carried out jointly by the parties or as part of the Collaboration; (b) the Party that has Opted Out shall have no rights to develop or commercialize, or to participate in Net Sales (except as provided in Section 11.5) from the commercialization of such Opt-out Product and shall not bear any of the costs related to commercialization of such Opt-out Product; and (c) the Continuing Party shall be free to develop and commercialize such Opt-out Product, either alone or with one or more Third Parties. For clarity, the development and commercialization of any Opt-out Product shall not be subject to the oversight or direction of the Committees and shall not be the subject of any Development Plans or plans relating to commercialization.

10.3.3 Manufacturing Obligation. Subject to the other provisions of this Agreement regarding expiration and termination, in the event Organon Opts Out as to a particular Collaboration Product and, at the time of such Opt-Out Organon or its Affiliated Subcontractor(s) is manufacturing such Collaboration Product pursuant to Article 7, at Lexicon's election, the manufacture and supply provisions of Article 7 shall nonetheless apply to such Collaboration Product for a period of (a) [**] following such Opt Out in the event Organon Opts Out prior to [**] and (b) [**] following such Opt Out in the event Organon Opts Out after [**].

10.3.4 Regulatory Matters. Following an Opt Out, the Continuing Party shall own all INDs, BLAs and other regulatory filings and submissions with respect to the relevant Opt-out Product(s) and all Regulatory Approvals with respect thereto. To the extent any such regulatory filings or submissions or Regulatory Approvals are held by the Opt-out Party, the Opt-out Party shall use Diligent Efforts to effect the assignment and transfer thereof to the Continuing Party. Notwithstanding any Opt Out, The Continuing Party and the Opt-out Party shall have the same obligations under Sections 8.5 and 8.6 with respect to an Opt-out Product as with respect to a Collaboration Product. 10.3.5.1 Diligent Efforts. Each Continuing Party shall use Diligent Efforts to actively Develop and obtain Regulatory Approval for at least one Opt-out Product acting through an Opt-out Target, and following such Regulatory Approval to maximize Net Sales of such Opt-out Product.

10.3.5.2 Effect of Failure to Satisfy Diligence Obligations.

With respect to each Opt-out Target (a) for which the Continuing Party fails to timely satisfy its diligence obligations under Section 10.3.5.1 above, at the option of the Opt-out Party as its sole and exclusive remedy therefor, subject to Sections 3.8.2 and 17.7 hereof, (i) the commercial licenses granted under Section 9.2 with respect to Opt-out Products acting through such Opt-out Target shall terminate, (ii) the Continuing Party shall deliver to the other party copies of all data, information, registrations and applications therefor relating to Opt-out Products acting through such Opt-out Target, and (iii) the Opt-out Party shall have the right, within the period of [**] following the Continuing Party's delivery of such copies, to obtain a commercial license under Section 9.2 for such Opt-out Products by delivering written notice thereof to the Continuing Party (in which case the Opt-out Party exercising such right shall become the Continuing Party with respect to such Opt-out Product).

(b) With respect to each Opt-out Product for which a party exercises its right, under Section 10.3.5.2(a), to obtain a commercial license regarding such Opt-out Product, the other party promptly shall deliver to such party all materials and copies of all data and information, and shall assign and transfer to such party all regulatory filings and submissions and Regulatory Approvals relating to such Opt-out Product.

ARTICLE 11. FINANCIAL TERMS

11.1 Up-Front Payment. Organon shall pay Lexicon an upfront access fee of twenty-two million five hundred thousand dollars (U.S. \$22,500,000), which access fee shall be invoiced within [**] of the Effective Date and due and payable in accordance with Section 11.11. For clarity, the upfront access fee paid by Organon under this Section 11.1 shall not be considered part of, or included in the calculation of Organon's Allowable Costs.

11.2 Research Funding During the Target Function Discovery Program Term. During the Target Function Discovery Program Term, Organon and/or Lexicon shall fund its proportion of the net Allowable Costs incurred by the other party in excess of the Cost Sharing Ratio as follows:

> (a) Organon shall make annual research payments to Lexicon of [**] for each of the first two (2) years of the Target Function Discovery Program Term. To the extent those payments exceed Organon's proportionate share of overall Allowable Costs based on the Cost Sharing Ratio, such payments will be credited against payments to be made by Organon after the second year in accordance with Section 11.6.1.4; and

> (b) Subject to Section 11.6.1, Organon or Lexicon shall make annual research payments to the other party in the final two (2) years of the Target Function Discovery Program Term in an amount equal to such party's proportionate share of the net Allowable Costs to be

incurred by the other party in excess of the Cost Sharing Ratio during such year in accordance with the annual budget agreed upon by the Joint Management Committee prior to the start of such year.

Notwithstanding the foregoing, the amounts payable by Organon to Lexicon under this Section 11.2 shall not exceed an aggregate of fifty million dollars (U.S. \$50,000,000) for Allowable Costs incurred during the Target Function Discovery Program Term, unless expressly authorized by the Joint Management Committee. The annual research payments for each such year shall be payable in [**] installments, which installments shall be invoiced within [**] of the start of each [**] and due and payable in accordance with Section 11.11. Such amounts shall be reconciled in accordance with Section 11.6.1 on a quarterly basis with actual Allowable Costs incurred by the parties.

11.3 Research Program Milestone Payments. Organon shall pay Lexicon a milestone payment of [**] for each Collaboration Product in excess of five (5) designated by the Joint Management Committee during the Collaboration Term; provided that Back-up Products for previously-designated Collaboration Products shall be excluded from such number. Each such milestone payment shall be invoiced within [**] following the designation by the Joint Management Committee of the applicable Collaboration Product and shall be due and payable in accordance with Section 11.11. For clarity, milestone payment(s) made by Organon under this Section 11.3 shall not be considered part of, or included in the calculation of, Organon's Allowable Costs.

Collaboration Revenue and Cost Sharing. All revenue from 11.4 Collaboration Products, Joint Marketing/Development Collaborator Revenue and Allowable Costs shall be shared in accordance with the applicable Cost Sharing Ratio. In the event that (a) fewer than five (5) Collaboration Products in total have been designated over the course of the Collaboration Term (including, in such number, Collaboration Products that have become Opt-out Products but excluding Back-up Products) and (b) Organon has Opted Out with respect to all Program Targets other than those for which Collaboration Products have been designated, then, effective upon Organon's Opt Out with respect to the last such Program Target, the Cost Sharing Ratio shall be adjusted [**] as follows: (i) in the event that four (4) such Collaboration Products in total have been designated over the course of the Collaboration Term (including, in such number, Collaboration Products that have become Opt-out Products but excluding Back-up Products), the Cost Sharing Ratio shall be adjusted to be, with respect to Lexicon, [**], and with respect to Organon, [**]; (ii) in the event that three (3) such Collaboration Products in total have been designated over the course of the Collaboration Term (including, in such number, Collaboration Products that have become Opt-out Products but excluding Back-up Products), the Cost Sharing Ratio shall be adjusted to be, with respect to Lexicon, [**], and with respect to Organon, [**]; (iii) in the event that two (2) Collaboration Products in total have been designated over the course of the Collaboration Term (including, in such number, Collaboration Products that have become Opt-out Products but excluding Back-up Products), the Cost Sharing Ratio shall be adjusted to be, with respect to Lexicon, [**], and with respect to Organon, [**]; and (iv) in the event that one (1) Collaboration Product in total has been designated over the course of the Collaboration Term (including, in such number, Collaboration Products that have become Opt-out Products but excluding Back-up Products), the Cost Sharing Ratio shall be adjusted to be, with respect to Lexicon, [**], and with respect to Organon, [**]. Notwithstanding the foregoing, no adjustment shall be made with respect to any Collaboration Product for which a Phase 2b Clinical Trial shall have commenced prior to the time such adjustment would otherwise have been made. In such event, the Cost Sharing Ratio applicable to Collaboration Product(s) for which no Phase 2b Clinical Trial shall have commenced shall be adjusted as set forth above, but the Cost Sharing Ratio applicable to Collaboration Product(s) for which a Phase 2b Clinical Trial shall have commenced shall remain unchanged from that in effect immediately prior to such adjustment. Notwithstanding the foregoing, in the event that Lexicon has not [**], then the previous two sentences shall not apply until such time as [**].

11.5 Royalties Payable with Respect to Opt-out Products in the Collaboration Field.

11.5.1 Royalty Rates for Opt-out Products. The Continuing Party shall pay royalties on Net Sales of Opt-out Products in the Collaboration Field at the rates specified below:

DEVELOPMENT STATUS OF OPT-OUT PRODUCT AT THE TIME OF OPT-OUT NOTICE	ROYALTY RATE
No IND Filed	[**]%
IND filed, but no Phase 2 Clinical Trial commenced	[**]%
Phase 2 Clinical Trial commenced but no Phase 3 Clinical Trial commenced	[**]%
Phase 3 Clinical Trial commenced but no Regulatory Approval	
in a Major Market	[* *]%
Regulatory Approval in a Major Market	[* *]%

[**]. For the avoidance of doubt, no royalties shall be payable under this Section 11.5.1 with respect to Net Sales of Opt-out Products in countries within the Territory for which the parties share Joint Marketing/Development Collaborator Revenue pursuant to Section 11.4.

11.5.2 Royalty Term. Royalties shall be payable, on a product-by-product and country-by-country basis, on Net Sales of Opt-out Products in the Collaboration Field for the longer of (a) the term of any Patent Rights Controlled by a party with a Valid Claim Covering the composition of matter or therapeutic use of such Opt-out Product in such country or (b) [**] after the First Commercial Sale of such Opt-out Product in such country.

11.5.3 Royalty Reduction.

11.5.3.1 Third Party Patents. If a Continuing Party, in its reasonable judgment, is required to obtain a license from any Third Party under any patent in order to import, manufacture, use or sell any Opt-Out Product, and if such party is required to pay to such Third Party a royalty under such license calculated on sales of such Opt-out Product and the infringement of such patent cannot reasonably be avoided by such party, or if such party is required by a court of competent jurisdiction to pay such a royalty to such a Third Party (and the infringement of such patent cannot reasonably be avoided by such party), then such party's obligation to pay royalties under Section 11.5.1 hereof shall be reduced by [**] of the amount of the royalty paid to such Third Party; provided however, that the royalties payable under Section 11.5.1 hereof shall not be reduced in any such event below [**] of the amounts set forth in Section 11.5.1. Such party shall use commercially reasonable efforts to minimize the amount of any of the foregoing payments owed by such party to a Third Party.

11.5.3.2 Generic Competition. In the event that (a) an Opt-Out Product is not Covered by any Valid Claim in a country providing marketing exclusivity with respect to such Opt-out Product in such country and (b) generic competition with respect to such Opt-out Product accounts for more than [**] of the market share of units sold in that country, the royalty rate with respect to such Opt-out Product (as set forth in Section 11.5.1 and adjusted, if applicable, in accordance with Section 11.5.3.1) shall be reduced by [**] in such country for so long as such conditions exist; provided however, that the royalties payable under Section 11.5.1 hereof shall not be reduced in any such event below [**] of the amounts set forth in Section 11.5.1. For purposes of the foregoing, "generic competition" with respect to such Opt-out Product refers to the marketing and sale of one or more Third Party products that (i) comprise the same Antibody, Antisense

Compound or Protein as such Opt-out Product or Collaboration Product and (ii) have been approved by the relevant Regulatory Authority for such country through an expedited process that relies in whole or in part on safety and efficacy data generated for such Opt-out Product.

11.6 Allowable Cost, Joint Marketing/Development Collaborator Revenue and Royalty Reports.

11.6.1 Allowable Cost and Joint Marketing/Development Collaborator Revenue Reports.

11.6.1.1 Reports of Allowable Costs. During the term of this Agreement, each party shall, within [**] after each [**], furnish to the other party a written [**] report showing in reasonable detail the Allowable Costs incurred by such party during such [**] in each of the categories specified in Section 1.5, on a Program Target-by-Program Target and Collaboration Product-by-Collaboration Product basis. [**]. Reports provided by either party with respect to all other activities will include, but not be limited to, a breakdown of FTE Costs by category of work, including the number of persons under each such category performing the work and the number of hours of work performed by each such person.

11.6.1.2 Reports of Joint Marketing/Development Collaborator Revenue. During the term of this Agreement, each party shall, within [**] after each [**], furnish to the other party a written [**] report showing in reasonable detail the Joint Marketing/Development Collaborator Revenue received by such party during such [**], on a Program Target-by-Program Target and Collaboration Product-by-Collaboration Product basis.

11.6.1.3 Lexicon Report of Net Payments Due. During the term of this Agreement, Lexicon shall, within [**] after each [**], furnish to Organon a written [**] report showing in reasonable detail:

> (a) the Allowable Costs incurred by Lexicon and Organon, respectively, and in total, during such Calendar Quarter in each of the categories specified in Section 1.5, on a Program Target-by-Program Target and Collaboration Product-by-Collaboration Product basis;

(b) the Joint Marketing/Development Collaborator Revenue received by Lexicon and Organon, respectively, and in total, during such Calendar Quarter, if any; and

(c) the amount, if any, owed by one party to the other party in reconciliation of the amounts set forth in subsections (a) and (b) above with the Cost Sharing Ratio.

11.6.1.4 Payment of Amounts Owed. Following receipt of the report delivered in accordance with Section 11.6.1.3(c), amounts shown as owed by one party shall be invoiced within [**] following the delivery of such report and shall be due and payable in accordance with Section 11.11. Notwithstanding the foregoing, during the Target Function Discovery Program Term, amounts, if any, owed by Lexicon to Organon shall not be invoiced or paid under this Section 11.6.1.4 but instead shall be carried over and credited against the amount of the next succeeding research payment(s) payable by Organon under Section 11.2(b).

11.6.1.5 Exchange Rates. Allowable Costs and Joint Marketing/Development Collaborator Revenue attributable to countries other than the United States shall be calculated in accordance with the standard exchange rate published in the Wall Street Journal (New York edition) and otherwise in accordance with conversion practices used by the reporting party for financial accounting purposes. The currency conversion system used by a party hereunder shall be subject to audit by the other party as described in Section 11.6.3 and, if not determined to be a system reflecting the fair market value of the currencies in question, shall be modified as necessary to effect currency conversion at fair market value.

11.6.1.6 Records. The parties shall each keep accurate books and accounts of record in connection with the Target Function Discovery Program, the Biotherapeutics Research Program and the Development, Manufacture and commercialization of Collaboration Products in a manner consistent with GAAP and in sufficient detail to permit accurate determination of all figures necessary for verification of Allowable Costs and Net Sales hereunder.

11.6.2 Royalty Reporting and Payment.

11.6.2.1 Royalty Reports. During the term of this Agreement following the First Commercial Sale of any Opt-out Product (or, as applicable, of any Collaboration Product in the Veterinary Field), the Continuing Party (or, as applicable, Organon) shall, within [**] after each [**], furnish to the other party a written [**] report showing, on a product-by-product and country-by-country basis:

(a) the gross sales and Net Sales of
 Opt-out Products (or, as applicable, of any
 Collaboration Product in the Veterinary Field) sold
 by such Continuing Party (or, as applicable,
 Organon), its (sub)licensees and their respective
 Affiliates during the reporting period and the
 calculation of Net Sales from such gross sales;

(b) the royalties payable in United States dollars which shall have accrued hereunder in respect of such Net Sales;

(c) withholding taxes, if any, required by law to be deducted in respect of such royalties;

(d) the dates of the First Commercial Sales of Opt-out Products (or, as applicable, of Collaboration Product in the Veterinary Field) in any country during the reporting period; and

(e) the exchange rates used in determining the amount of United States dollars payable hereunder.

Royalties payable on sales in countries other than the United States shall be calculated in accordance with the standard exchange rate conversion practices used by the Continuing Party (or, as applicable, Organon) for financial accounting purposes. If no royalty or payment is due for any royalty period hereunder, such party shall so report. Each party shall keep, and shall require its (sub)licensees to keep (all in accordance with generally accepted accounting principles, consistently applied), complete and accurate records in sufficient detail to properly reflect all gross sales and Net Sales and to enable the royalties payable hereunder to be determined. Each party shall include in each agreement with each applicable (sub)licensee a provision requiring such (sub)licensee to make reports to such party, to keep and maintain records of sales made pursuant to such agreement and to grant access to such records by the other party's independent certified public accountant to the same extent required of a party under this Agreement.

11.6.2.2 Royalty Payment Terms. Royalty payments for each [**] shall be invoiced at the time the applicable report under Section 11.6.2.1 for such [**] shall be due and shall be payable in accordance with Section 11.11.

Audits. Upon the written request of a party, the 11.6.3 other party shall permit an independent certified public accountant selected by the requesting party and acceptable to the other party, which acceptance shall not be unreasonably withheld, to have access, at reasonable times and during normal business hours, to such records of such other party as may be reasonably necessary to verify the accuracy of the reports described herein, in respect of any fiscal year ending not more than [**] prior to the date of such request. Each party shall use commercially reasonable efforts to schedule all such verifications within [**] after the requesting party makes its written request. All such verifications shall be conducted not more than [**]. The report of the requesting party's independent certified public accountant shall be made available to both parties. Subject to the other party's rights under Section 17.7, in the event requesting party's independent certified public accountant concludes that additional amounts were owed to the requesting party for such period, the additional amounts shall be paid by the other party within [**] of the date the requesting party delivers to the other party such independent certified public accountant's written report so concluding, unless such report contains manifest error. In the event requesting party's independent certified public accountant concludes that there was an overpayment to such party during such period, the overpayment shall be repaid by the requesting party within [**] of the date the requesting party received such independent certified public accountant's written report so concluding, unless such report contains manifest error. The fees charged by such independent certified public accountant shall be paid by the requesting party unless such audit discloses an underpayment or overpayment of more than [**] of the amount due under this Agreement for the period in question, in which case the party responsible for such underpayment or overpayment will bear the full cost of such audit. Each party agrees that all information subject to review under this Section 11.6.3 or under any agreement with a (sub)licensee of a party is confidential and that the party receiving such information shall cause its independent certified public accountant to retain all such information in confidence. The requesting party's independent certified public accountant shall only report to the requesting party as to the computation of Allowable Costs, Joint Marketing/Development Collaborator Revenue or royalties payable under this Agreement, and shall not disclose to the requesting party any other information of the other party or any (sub)licensee of a Continuing Party.

11.7 Withholding Taxes. In the event that any royalties or other payments due to a party are subject to withholding tax required by law to be paid to the taxing authority of any foreign country, the amount of such tax may be withheld from the applicable royalties or other payment due such party. The party owing such payment shall promptly pay such tax on behalf of the party to which such payment is owed and shall furnish the party to which such payment is owed with a certificate of withholding tax so deducted for such party's avoidance of duplicate taxation in multiple countries. The party owing such payment may not deduct any other withholding or any other governmental charges from the payments agreed upon under this Agreement, except to the extent same are paid on behalf of, or for the benefit of, the party to which such payment is owed. The party owing such payment shall maintain official receipts of payment of any such withholding taxes and shall forward such receipts to the party to which such payment is owed.

Blocked Currency. If by law, regulation, or fiscal policy of a 11.8 particular country, conversion into United States dollars or transfer of funds of a convertible currency to the United States is restricted or forbidden, the party owing such payment shall give the party to which such payment is owed prompt written notice and shall make such payment due under this Article 11 through such means or methods as are lawful in such country as the party to which such payment is owed may reasonably designate. Failing the designation by the party to which such payment is owed of such lawful means or methods within [**]after such written notice is given to such party, the party owing such payment shall deposit such royalty payment in local currency to the credit of the party to which such payment is owed in a recognized banking institution designated by such party, or if none is designated by such party within the [**] period described above, in a recognized banking institution selected by the party owing such payment and identified in a written notice to other party, and such deposit shall fulfill all obligations of the party owing such payment to the other party with respect to such payment.

11.9 Interest on Late Payments. A party to which payment is owed under this Agreement shall have the right to seek to collect interest on any payments that are not paid on or before [**] after the date such payments are due under this Agreement at a rate equal to [**], calculated on the total number of days payment is delinquent. The party to which interest is owed shall send a written notice to the delinquent party notifying it of any such delinquency; provided, that such party shall not be required to send more than two (2) such notices in any Contract Year and; provided, further, that the date of delivery of such notice shall not affect the calculation of the amount of interest owed hereunder.

11.10 Manner of Payment. Except as provided in Section 11.8, payments to be made by one party to the other under this Agreement shall be payable in United States dollars and shall be paid by bank wire transfer in immediately available funds to such bank account in the state and country in which such principal office is located as is designated in writing by such party from time to time.

11.11 Invoices. All amounts to be paid under this Agreement shall be due and payable within [**] after receipt by the party owing such payment of an invoice setting forth the amounts owed and the calculations for determining such amounts owed. Invoices shall be sent to the following addresses:

For Lexicon:	Lexicon Genetics Incorporated
	8800 Technology Forest Place
	The Woodlands, Texas 77381
	Attention: Accounts Payable

For Organon: N.V. Organon Purchase Accounting (KA1041) P.O. Box 20 5340 BH Oss The Netherlands Attention: Manager, Research Alliances

12.1 Ownership of Intellectual Property.

12.1.1 Ownership by Lexicon of the Lexicon Background Materials and Lexicon Background Technology. Subject to the rights and licenses granted under this Agreement, Lexicon (and its licensors, as applicable) shall own and retain all rights to the Lexicon Background Materials and Lexicon Background Technology.

12.1.2 Ownership by Organon of the Organon Background Materials and Organon Background Technology. Subject to the rights and licenses granted under this Agreement, Organon (and its licensors, as applicable) shall own and retain all rights to the Organon Background Materials and Organon Background Technology.

12.1.3 Ownership of Program Intellectual Property.

12.1.3.1 Inventorship. Inventorship for patentable inventions and discoveries conceived or reduced to practice during the course of the performance of activities pursuant to this Agreement shall be determined in accordance with U.S. patent laws for determining inventorship. In the event of a dispute regarding inventorship, if the parties are unable to resolve such inventorship dispute, the Joint Management Committee shall establish a procedure to resolve such dispute, which may include engaging a Third Party patent attorney jointly selected by the parties to resolve such dispute, which resolution by such patent attorney shall be binding upon the parties.

12.1.3.2 Ownership of Program Technology and Program Intellectual Property for Program Targets and Related Antibodies, Antisense Compounds and Proteins. Subject to the rights and licenses granted under this Agreement, the parties shall jointly own all Program Technology and Program Intellectual Property that directly relates to (a) a Program Target, (b) the utility of such Program Target, (c) the use of a Program Target to identify Antibodies or Antisense Compounds acting through such Program Target, and the use of such Antibodies or Antisense Compounds in the Collaboration Field or Veterinary Field, (d) the use of Protein(s) encoded by such Program Target in the Collaboration Field or Veterinary Field, (e) Program Antibodies, (f) Program Antisense Compounds, (g) Program Proteins and (h) Collaboration Products, whether invented or discovered by employees, Affiliates, agents, independent contractors or consultants of Lexicon, Organon or both parties. In the event either party would otherwise be deemed to be the sole owner of any such invention, then such party shall assign to the other party an undivided joint interest in such invention.

12.1.3.3 Ownership of Other Program Technology and Program Intellectual Property. Except as set forth in Section 12.1.3.2, title to all Program Technology and Program Intellectual Property shall be based upon the inventorship for such Program Technology and Program Intellectual Property. Except as set forth in Section 12.1.3.2, (a) Lexicon shall own Program Technology and Program Intellectual Property invented solely by employees, agents, consultants or contractors of Lexicon or a Lexicon Affiliate; (b) Organon shall own Program Technology and Program Intellectual Property invented solely by employees, agents, consultants or contractors of Organon or a Organon Affiliate; and (c) Lexicon and Organon shall jointly own Program Technology and Program Intellectual Property invented jointly by employees, agents, consultants or contractors of both Lexicon and Organon or Affiliates of Lexicon and Organon. Each

party shall disclose to the other party promptly any inventions within the Program Technology and Program Intellectual Property made by such party's Affiliates, employees, agents or consultants.

12.2 Prosecution and Maintenance of Program Patent Rights.

12.2.1 Primary Prosecution Rights. The responsibility for (a) preparing, filing and prosecuting patent applications (including, but not limited to, provisional, reissue, continuing, continuation, continuation-in-part, divisional, and substitute applications and any foreign counterparts thereof) Covering inventions within the Program Technology and Program Intellectual Property; (b) maintaining any Program Patent Rights; and (c) managing any interference or opposition or similar proceedings relating to the foregoing ((a) through (c), collectively, "Patent Prosecution") shall be the responsibility of Lexicon, unless otherwise determined by the Joint Management Committee. In making such determination, the Joint Management Committee shall consider, among other factors, the nature of the claimed subject matter, the relative contribution of each party to the claimed subject matter and the relatedness of the claimed subject matter to that in other patent applications being prosecuted by the parties. Notwithstanding the foregoing, all decisions related to (y) [**] or (z) [**], shall be the responsibility of the Joint Management Committee.

12.2.2 Secondary Prosecution Rights. If the prosecuting party elects not to continue pursuing Patent Prosecution for an invention within the Program Technology and Program Intellectual Property (and the other party has joint ownership of or a license under such Program Patent Rights pursuant to this Agreement), then the prosecuting party shall notify the other party in writing of such election at least [**] prior to the last available date for action to preserve such Program Patent Rights. If such other party elects to continue Patent Prosecution, it will not be liable to the other party in any way with respect to its handling of, or the results obtained from, such Patent Prosecution. The other party will provide the party taking over Patent Prosecution with such assistance and execute such documents as are necessary to continue or permit such Patent Prosecution.

12.2.3 Right of Review. The prosecuting party under Sections 12.2.1 or 12.2.2 shall provide the other party with a reasonable opportunity to review and provide substantive input to material decisions relating to Patent Prosecution. The prosecuting party shall furnish to the other party copies of any substantive actions prepared for the U.S. Patent and Trademark Office or its foreign counterparts that may materially affect the Program Patent Rights being prosecuted or maintained reasonably in advance of the filing of such action in order to provide such party with a meaningful opportunity to comment thereon. Such action filed by the prosecuting party shall reflect any comments received from the other party that are received in a reasonably timely manner and are reasonably directed to maximizing the coverage of the claims of such Program Patent Rights being prosecuted or maintained.

12.2.4 Patent Prosecution Costs. All Patent Prosecution expenses, including attorneys' fees, incurred in the performance of Patent Prosecution under Section 12.2.1 or 12.2.2 shall be shared by the parties in accordance with the Cost Sharing Ratio.

12.2.5 Cooperation. Each party hereby agrees:

(a) to take all reasonable additional actions and execute such agreements, instruments and documents as may be reasonably required to perfect the other's

ownership interest in accordance with the intent of this Agreement, including, without limitation, requiring inventors to make appropriate patent assignments;

(b) to make its employees, Affiliates, agents, independent contractors and consultants reasonably available to the other party (or to the other party's authorized attorneys, agents or representatives), to the extent reasonably necessary to enable the prosecuting party to undertake Patent Prosecution;

(c) to provide the other party with copies of all material correspondence with the U.S. Patent and Trademark Office or its foreign counterparts;

(d) to cooperate, if necessary and appropriate, with the other party in gaining patent term extensions wherever applicable to Program Patent Rights for Program Inventions; and

(e) to endeavor in good faith to coordinate its efforts with the other party to minimize or avoid interference with the Patent Prosecution of the other party's patent applications related to inventions within the Program Technology and Program Intellectual Property.

12.3 Patent Term Extension. Each party shall cooperate with the other in obtaining patent term extension or supplemental protection certificates or their equivalents in any country with respect to the Program Patent Rights.

12.4 Enforcement of the Program Patent Rights.

12.4.1 Notices of Third Party Infringement. Each Party shall promptly provide the other Party with written notice reasonably detailing any known or alleged infringement of Program Patent Rights by a Third Party.

12.4.2 Hatch-Waxman Notifications. Each party shall provide to the other party copies of any allegations of alleged patent invalidity, unenforceability or non-infringement of a patent or patents with respect to Program Technology, Program Materials or Collaboration Products pursuant to a Paragraph IV Patent Certification by a Third Party filing an Abbreviated New Drug Application (i.e., an action under the Hatch-Waxman Act). Such copies shall be provided promptly after receipt of such certification.

12.4.3 Other Notifications. Each party shall provide to the other party copies of any notices it receives from Third Parties regarding any patent nullity actions, any declaratory judgment actions, any alleged infringement of Program Patent Rights or any alleged misappropriation of intellectual property with respect to Program Technology, Program Materials or Collaboration Products. Such copies shall be provided promptly following receipt thereof.

12.4.4 Product-Related Infringement.

12.4.4.1 Lexicon shall have the first right, but not the obligation, to institute and direct legal proceedings against any Third Party believed to be infringing the Program Patent Rights of either party by the manufacture, use, importation, offer for sale or sale of a product competitive with a Collaboration Product (whether a clinical or commercial product). Each party will bear its own costs, including attorneys' fees, relating to such legal proceedings; provided that Lexicon shall bear Organon's out-ofpocket expenses, including attorneys' fees, incurred in complying with requests for cooperation made by Lexicon. Any recovery in connection with such suit or proceeding will first be applied to reimburse the parties for their out-of-pocket expenses, including attorneys' fees. All recoveries resulting from such legal proceedings that are in excess of the parties' costs of bringing or participating in such action, including attorneys' fees, shall be allocated in accordance with the Cost Sharing Ratio.

12.4.4.2 If Lexicon elects not to institute and direct legal proceedings against any Third Party believed to be infringing the Program Patent Rights of either party as described in Section 12.4.4.1, Organon shall have the right, but not the obligation, to institute and direct such legal proceedings. Each party will bear its own costs, including attorneys' fees, relating to such legal proceedings; provided that Organon shall bear Lexicon's out-of-pocket expenses, including attorneys' fees, incurred in complying with requests for cooperation made by Organon. Any recovery in connection with such suit or proceeding will first be applied to reimburse the parties for their out-of-pocket expenses, including attorneys' fees. All recoveries resulting from such legal proceedings that are in excess of the parties' costs of bringing or participating in such action, including attorneys' fees, shall be allocated in accordance with the Cost Sharing Ratio.

12.4.4.3 In the event that a party takes action under this Section 12.4.4, the other party shall cooperate to the extent reasonably necessary at the sole expense of the party taking such action. Upon the reasonable request of the party taking such action, the other party shall join the suit and shall be represented in any such legal proceedings using counsel of its own choice. Neither party shall settle or otherwise agree to the final disposition of any claim or proceeding relating to Program Patent Rights Controlled in whole or in part by the other party or licensed under this Agreement to the other party without the prior written consent of such other party, which consent shall not be unreasonably withheld.

12.4.5 Non-Product-Related Infringement. Each party shall have the sole right, but not the obligation, to institute and direct legal proceedings against any Third Party believed to be infringing the Program Patent Rights solely owned by such party other than infringement relating to a Collaboration Product. All costs, including attorneys' fees, relating to such legal proceedings shall be borne by the party instituting such legal proceedings, and all recoveries resulting from such legal proceedings shall be retained by such party. The parties shall consult with each other regarding the institution, prosecution and control of any action or proceeding with respect to infringement of any of the Program Patent Rights jointly owned by the parties other than infringement relating to a Collaboration Product.

12.5 Notices of Other Proceedings.

12.5.1 Each party shall notify the other in writing of any allegations it receives from a Third Party that the manufacture, use, sale, offer for sale or import of Program Technology, Program Materials or any Collaboration Product infringes the intellectual property rights of such Third Party. Such notice shall be provided promptly following receipt of such allegations.

12.5.2 In the event that a party receives notice that it or any of its Affiliates have been individually named as a defendant in a legal proceeding by a Third Party alleging infringement of a Third Party patent or other intellectual property right as a result of the manufacture, use, sale, offer for sale or import of Program Technology, Program Materials or a Collaboration Product, such party shall immediately notify the other party in writing after the receipt of such notice. Such written notice shall include a copy of any summons or complaint (or the equivalent thereof) received regarding the foregoing.

ARTICLE 13. CONFIDENTIALITY

13.1 Nondisclosure Obligations.

13.1.1 General. Except as otherwise provided in this Article 13, during the term of this Agreement and for a period of five (5) years thereafter, each Receiving Party shall maintain the Confidential Information of each Disclosing Party in confidence and use it only for purposes specifically authorized under this Agreement.

13.1.2 Limitations. To the extent it is reasonably necessary or appropriate to fulfill its obligations or exercise its rights under this Agreement and subject to advance written notification to the Disclosing Party: (a) a party may disclose to Third Parties Confidential Information it is otherwise obligated not to disclose under this Section 13.1, to its Affiliates, (sub)licensees, consultants, outside contractors and clinical investigators, on a strict need-to-know basis for the purposes contemplated by this Agreement and on condition that such entities or persons agree to keep the Confidential Information confidential for the same time periods and to the same extent as such party is required to keep the Confidential Information confidential hereunder; and (b) a party or its (sub)licensees may disclose, using appropriate measures to preserve confidentiality, such Confidential Information to government or other regulatory authorities to the extent that such disclosure is reasonably necessary to obtain authorizations to conduct clinical trials of, and to commercially market, Collaboration Products pursuant to this Agreement. Furthermore, a Receiving Party may request permission from the Disclosing Party to disclose such Confidential Information to the extent that such disclosure is reasonably necessary to [**].

13.1.3 Required Disclosure. A Receiving Party may disclose Confidential Information pursuant to interrogatories, requests for information or documents, subpoena, civil investigative demand issued by a court or governmental agency or as otherwise required by law; provided, however, that the Receiving Party shall notify the Disclosing Party promptly upon receipt thereof, giving [**] the Disclosing Party sufficient advance notice to permit it to oppose, limit or seek confidential treatment for such disclosure; and provided, further, that the Receiving Party shall furnish only that portion of the Confidential Information which it is advised by counsel is legally required whether or not a protective order or other similar order is obtained by the Disclosing Party.

13.2 Injunctive Relief. The parties hereto understand and agree that remedies at law may be inadequate to protect against any breach of any of the provisions of this Article 13 by either party or their employees, agents, officers or directors or any other person acting in concert with it or on its behalf. Accordingly, each party shall be entitled to the granting of injunctive relief by a court of competent jurisdiction against any action that constitutes any such breach of this Article 13.

13.3 Publication. Organon and Lexicon (each, a "Submitting Party") may each publish or present data and results relating to a Collaboration Product or Opt-out Product for which the Submitting Party holds a commercial license under Section 9.2 hereof, subject to the prior review of the proposed disclosure by the other party (each, a "Reviewing Party"), solely to determine (a) whether the proposed disclosure contains the Confidential Information of the Reviewing Party or (b) whether the information contained in the proposed disclosure should be the subject of a patent application to be filed by the Reviewing Party prior to such disclosure. Each Submitting Party shall provide the Reviewing Party with

the opportunity to review any proposed abstract, manuscript or presentation by delivering a copy thereof to the Reviewing Party no less than [**] before its intended submission for publication or presentation. The Reviewing Party shall have [**] from its receipt of any such abstract, manuscript or presentation in which to notify the Submitting Party in writing of any specific objections to the disclosure, based on either the need to seek patent protection or concern regarding the specific disclosure of the Confidential Information of the Reviewing Party. In the event the Reviewing Party objects to the disclosure, the Submitting Party agrees not to submit the publication or abstract or make the presentation containing the objected-to information until the Reviewing Party is given a reasonable additional period of time (not to exceed an additional [**]) to seek patent protection for any material in the disclosure which the Reviewing Party believes is patentable (subject, in all events, to Section 13.2) or, in the case of Confidential Information, to allow the Submitting Party to delete any Confidential Information of the Reviewing Party from the proposed disclosure. The Submitting Party agrees to delete from the proposed disclosure any Confidential Information of the Reviewing Party upon request.

ARTICLE 14. REPRESENTATIONS AND WARRANTIES

14.1 Representations, Warranties and Covenants of Lexicon. Lexicon represents and warrants to and covenants with Organon that:

14.1.1 Lexicon is a corporation duly organized, validly existing and in corporate good standing under the laws of the State of Delaware, United States of America;

14.1.2 Lexicon has the corporate and legal right, authority and power to enter into this Agreement, and to extend the rights and licenses granted to Organon in this Agreement;

14.1.3 Lexicon has taken all necessary action to authorize the execution, delivery and performance of this Agreement;

14.1.4 upon the execution and delivery of this Agreement, this Agreement shall constitute a valid and binding obligation of Lexicon, enforceable in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' and contracting parties' rights generally and except as enforceability may be subject to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law);

14.1.5 the performance of Lexicon's obligations under this Agreement will not conflict with its charter documents or result in a breach of any agreements, contracts or other arrangements to which it is a party;

14.1.6 Lexicon will not during the term of this Agreement enter into any agreements, contracts or other arrangements that would be inconsistent with its obligations under this Agreement;

14.1.7 Lexicon has no Pre-existing Obligations with respect to the genes in the list provided by Lexicon to Organon prior to the Effective Date from which Program Targets are to be selected in accordance with Section 2.3 to which the licenses granted by Lexicon in Article 9 would be subject; and

14.1.8 To the best of Lexicon's knowledge after reasonable inquiry, Lexicon is not aware of any patent or other intellectual property rights of any Third Party that would be

infringed by its conduct of the Target Function Discovery Program, and has received no notice from any Third Party claiming any such infringement.

14.2 Representations, Warranties and Covenants of Organon. Organon represents and warrants to and covenants with Lexicon that:

14.2.1 Organon is a registered company duly organized, validly existing and in good standing under the laws of the Netherlands;

14.2.2 Organon has the corporate and legal right, authority and power to enter into this Agreement, and to extend the rights and licenses granted to Lexicon in this Agreement;

14.2.3 Organon has taken all necessary action to authorize the execution, delivery and performance of this Agreement;

14.2.4 upon the execution and delivery of this Agreement, this Agreement shall constitute a valid and binding obligation of Organon enforceable in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' and contracting parties' rights generally and except as enforceability may be subject to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law);

14.2.5 the performance of its obligations under this Agreement will not conflict with Organon's charter documents or result in a breach of any agreements, contracts or other arrangements to which it is a party; and

14.2.6 Organon will not after the Effective Date enter into any agreements, contracts or other arrangements that would be inconsistent with its obligations under this Agreement.

14.3 Warranty Disclaimer. EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN THIS AGREEMENT, NEITHER PARTY MAKES ANY WARRANTY WITH RESPECT TO ANY PRODUCT, PATENT RIGHTS, GOODS, SERVICES, BACKGROUND MATERIALS OR ANY OTHER SUBJECT MATTER OF THIS AGREEMENT, AND EACH PARTY HEREBY DISCLAIMS WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NON-INFRINGEMENT WITH RESPECT TO ANY AND ALL OF THE FOREGOING. IN ADDITION, THE PARTIES ACKNOWLEDGE THAT THE GENERATION OR USE OF BACKGROUND MATERIALS MAY BE COVERED BY ONE OR MORE VALID PATENTS OF THIRD PARTIES. EACH PARTY ACKNOWLEDGES THAT EXERCISE BY IT OF THE RIGHTS AND LICENSES GRANTED TO IT PURSUANT TO ARTICLE 9 HEREOF MAY BE COVERED BY ONE OR MORE VALID PATENTS OF THIRD PARTIES.

14.4 Limited Liability. EXCEPT AS SPECIFICALLY SET FORTH IN THIS AGREEMENT, NEITHER LEXICON NOR ORGANON WILL BE LIABLE WITH RESPECT TO ANY MATTER ARISING UNDER THIS AGREEMENT UNDER ANY CONTRACT, NEGLIGENCE, STRICT LIABILITY OR OTHER LEGAL OR EQUITABLE THEORY FOR ANY PUNITIVE, EXEMPLARY, INCIDENTAL OR CONSEQUENTIAL DAMAGES OR LOST PROFITS.

ARTICLE 15. INDEMNITY

15.1 Lexicon Indemnity Obligations. Lexicon agrees to defend, indemnify and hold Organon, its Affiliates and their respective employees and agents harmless from all claims, losses, damages or

expenses (including reasonable attorneys' fees and costs of litigation) arising as a result of: (a) actual or asserted violations of any applicable law or regulation by Lexicon, its (sub)licensees and their respective Affiliates by virtue of which any Lexicon Products manufactured, distributed or sold by Lexicon hereunder as Continuing Party shall be alleged or determined to be adulterated, misbranded, mislabeled or otherwise not in compliance with any applicable law or regulation; (b) claims for bodily injury, death or property damage attributable to the manufacture, distribution, sale or use of any Lexicon Products by Lexicon as Continuing Party, its (sub)licensees and their respective Affiliates; (c) a recall of a Lexicon Product manufactured, distributed or sold by Lexicon hereunder as the Continuing Party ordered by a governmental agency or required by a confirmed Lexicon Product failure as reasonably determined by the parties hereto; or (d) Lexicon's breach of any of its representations, warranties or covenants hereunder.

15.2 Organon Indemnity Obligations. Organon agrees to defend, indemnify and hold Lexicon, its Affiliates and their respective employees and agents harmless from all claims, losses, damages or expenses (including reasonable attorneys' fees and costs of litigation) arising as a result of: (a) actual or asserted violations of any applicable law or regulation by Organon, its (sub)licensees and their respective Affiliates by virtue of which any Organon Products manufactured, distributed or sold by Organon hereunder as the Continuing Party shall be alleged or determined to be adulterated, misbranded, mislabeled or otherwise not in compliance with any applicable law or regulation; (b) claims for bodily injury, death or property damage attributable to the manufacture, distribution, sale or use of any Organon Products by Organon as Continuing Party hereunder, its (sub)licensees and their respective Affiliates; (c) a recall of an Organon Product manufactured, distributed or sold by Organon hereunder as the Continuing Party ordered by a governmental agency or required by a confirmed Organon Product failure as reasonably determined by the parties hereto; or (d) Organon's breach of any of its representations, warranties or covenants hereunder.

15.3 Limitation on Indemnity Obligations. Neither party, its Affiliates or their respective employees and agents shall be entitled to the indemnities set forth in Sections 15.1 or 15.2, respectively, to the comparative extent the claim, loss, damage or expense for which indemnification is sought was caused by a grossly negligent, reckless or intentional act or omission by such party, its directors, officers, employees or authorized agents.

15.4 Procedure. If a party or any of its Affiliates or their respective employees or agents (collectively, the "Indemnitee") intends to claim indemnification under this Article 15, the Indemnitee shall promptly notify the other party (the "Indemnitor") of any loss, claim, damage, liability or action in respect of which the Indemnitee intends to claim such indemnification, and the Indemnitor shall assume the defense thereof with counsel selected by the Indemnitor and reasonably acceptable to the Indemnitee, provided, however, that an Indemnitee shall have the right to retain its own counsel, with the fees and expenses to be paid by the Indemnitee, if representation of such Indemnitee by the counsel retained by the Indemnitor would be inappropriate due to actual or potential differing interests between such Indemnitee and any other party represented by such counsel in such proceedings. The Indemnitor shall have the right to settle or compromise any claims for which it is providing indemnification under this Article 15, provided that the consent of the Indemnitee (which shall not be unreasonably withheld or delayed) shall be required in the event any such settlement or compromise would adversely affect the interests of the Indemnitee. The indemnity agreement in this Article 15 shall not apply to amounts paid in settlement of any loss, claim, damage, liability or action if such settlement is effected without the consent of the Indemnitor. The failure to deliver notice to the Indemnitor within a reasonable time after the commencement of any such action, if prejudicial to the Indemnitor's ability to defend such action, shall relieve such Indemnitor of any liability to the Indemnitee under this Article 15, but the omission so to deliver notice to the Indemnitor will not relieve it of any liability that it may have to any Indemnitee otherwise than under this Article 15. The Indemnitee under this Article 15, its employees and agents, shall

cooperate fully with the Indemnitor and its legal representatives in the investigation of any action, claim or liability covered by this indemnification.

15.5 Insurance. Each party shall maintain appropriate product liability insurance with respect to Development, Manufacture and commercialization of Collaboration Products by such party in such amount as such party customarily maintains with respect to sales of its other products. Each party shall maintain such insurance for so long as it continues to manufacture or sell Collaboration Products, and thereafter for so long as such party customarily maintains insurance with respect to sales of its other products.

ARTICLE 16. EXPIRATION AND TERMINATION

16.1 Term of Agreement. The term of this Agreement shall commence on the Effective Date and shall continue until the later of (a) the expiration or termination of the last to expire of any Valid Claim included in the Program Patent Rights and (b) the cessation of Development or commercialization activities with respect to all Collaboration Products generated pursuant to the terms hereof.

16.2 Events of Default. An Event of Default by either party shall have occurred upon (a) the occurrence of a material breach of this Agreement if such party fails to remedy such breach within [**] after written notice thereof by the non-breaching party ([**] in the event of a party's failure to make a payment required hereunder) or, if remediation of such breach in [**] is not practicable, if such party fails to commence and diligently pursue such remediation during such [**] period, or (b) the commencement of any proceeding in or for bankruptcy, insolvency, dissolution or winding up by or against such party that is not dismissed or otherwise disposed of within [**] thereafter.

16.3 Effect of an Event of Default. In the event of an Event of Default, the non-defaulting party shall have the right, at its option exercisable in its sole discretion, in addition to any other rights or remedies available to it at law or in equity and subject to the limitations set forth in Sections 3.8.2, 14.4 and 17.7 hereof, to terminate this Agreement upon [**] notice thereof to the other party, in which case (a) the licenses granted to the defaulting party pursuant to Article 9 shall terminate and (b) the defaulting party shall return to the non-defaulting party or, upon the non-defaulting party's written instruction, destroy all information, materials or documentation provided by the non-defaulting party pursuant to this Agreement; provided that such termination shall apply to the rights and licenses granted to the defaulting party under Sections 9.1 and 9.2 with respect to a Collaboration Product or Opt-out Product only in the event, and to the extent, that such Event of Default relates to such specific Collaboration Product or Opt-out Product, in which case the defaulting party shall be deemed to have Opted Out of any such Collaboration Product with the same effect (subject to any other rights or remedies available to the non-defaulting party at law or in equity) as if such breaching party had Opted Out under Section 10.1.1.1. For the avoidance of doubt, the termination rights of the non-defaulting party under this Section 16.3 shall be in addition to any other rights or remedies available to it at law or in equity. The rights and licenses granted to the defaulting party under Sections 9.1 and 9.2 with respect to any Collaboration Product or Opt-out Product with respect to which no Event of Default has occurred shall, subject to such party's applicable obligations under this Agreement with respect thereto, continue. In the event that this Agreement is terminated pursuant to this Section 16.3, the defaulting party's obligations under the exclusivity provisions set forth in Section 2.4 shall survive for five (5) years following such termination.

16.4 Effect of Expiration or Termination of Agreement. The expiration or termination of this Agreement shall not relieve the parties of any obligation accruing prior to such expiration or termination. The provisions of Articles 12, 13 and 15, Sections 8.5, 8.6 and 17.2 and Sections 17.4 through 17.7 hereof shall survive the expiration or termination of this Agreement. The rights and licenses granted to the non-defaulting party under Sections 9.1 and 9.2 hereof shall survive any termination of this Agreement by such non-defaulting party pursuant to Section 16.3. The provisions of Sections 11.2 through 11.11 hereof shall survive any termination of this Agreement under which a party, its (sub)licensees or their respective Affiliates retains the right to sell Collaboration Products until such time as this Agreement would have expired with respect to any Collaboration Product, as the case may be, in any country pursuant to Section 16.1 hereof had this Agreement not been earlier terminated.

ARTICLE 17. MISCELLANEOUS

17.1 Force Majeure. Neither party shall be held liable or responsible to the other party nor be deemed to have defaulted under or breached this Agreement for failure or delay in fulfilling or performing any obligation under this Agreement when such failure or delay is caused by or results from causes beyond the reasonable control of the affected party, including but not limited to fire, floods, embargoes, war, acts of war (whether war is declared or not), insurrections, riots, civil commotions, strikes, lockouts or other labor disturbances, acts of God or acts, omissions or delays in acting by any governmental authority; provided, however, that the party so affected shall use reasonable commercial efforts to avoid or remove such causes of nonperformance, and shall continue performance hereunder with reasonable dispatch whenever such causes are removed. Either party shall provide the other party with prompt written notice of any delay or failure to perform that occurs by reason of force majeure. The parties shall mutually seek a resolution of the delay or the failure to perform as noted above.

17.2 Assignment. This Agreement may not be assigned or otherwise transferred, in whole or in part, by either party without the consent of the other party; provided, however, that either Lexicon or Organon may, without such consent, assign its rights and obligations under this Agreement (i) to any Affiliate, or (ii) in connection with a merger, consolidation or sale of such portion of a party's assets that includes rights under this Agreement to an unrelated Third Party; provided, further, that such party's rights and obligations under this Agreement to an unrelated Third Party; provided, further, that such party's rights and obligations under this Agreement shall be assumed by its successor in interest in any such transaction and shall not be transferred separate from all or substantially all of its other business assets, including those business assets that are the subject of this Agreement. Any purported assignment in violation of the preceding sentence shall be void. Any permitted assignee shall assume all obligations of its assignor under this Agreement, unless the parties otherwise agree. Any permitted assignee shall also, within [**] following such assignment, provide the continuing party with a written summary of how it intends to fulfill its obligations under this Agreement with respect to each-Collaboration Product.

17.3 Change in Control.

17.3.1 Notice of and Reasonable Assurances upon a Change in Control. Within [**] after any Change in Control of a party, the surviving party to such Change in Control (the "Acquiring Party") (a) will notify the other party of the occurrence of such Change in Control and (b) will provide the other party with reasonable assurances of its commitment to fulfill its obligations under this Agreement and any applicable Biotherapeutics Research Plan(s), Development Plan(s) and Manufacturing Plan(s) then in effect.

17.3.2 Effect of Failure to Provide Assurances upon a Change in Control Occurring during the Target Discovery Program Term. To the extent that a Change in Control occurs during the Target Function Discovery Program Term and the Acquiring Party fails to provide the assurances contemplated by Section 17.3.1 within the [**] period set forth therein, and does not cure such failure within [**] after notice from the other party, the other party shall have the right, with respect to any matter under Sections [**] as to which the Joint Management Committee is unable to reach unanimity and that remains unresolved following the procedures set forth in Section 3.8.2, to make the final decision with respect to such matter, consistent with the terms and conditions of this Agreement. 17.3.3 Effect of Failure to Provide Assurances upon a Change in Control Occurring after the Target Function Discovery Program Term. To the extent that a Change in Control occurs following the conclusion of the Target Function Discovery Program Term and the Acquiring Party fails to provide the assurances contemplated by Section 17.3.1 within the [**] period set forth therein, and does not cure such failure within [**] after notice from the other party, the Acquiring Party shall be deemed to have Opted Out of any Collaboration Product to which such failure reasonably relates, with the same effect as if the Acquiring Party had Opted Out under Section 10.1.1.1; provided that the Acquiring Party shall not [**] with respect to (a) [**] or (b) [**].

17.3.4 Option to Negotiate for Rights to Collaboration Product upon a Change in Control Occurring after the Target Function Discovery Program Term. Without limiting the foregoing, if requested by the other party at any time within the [**] period following the Acquiring Party's notice under Section 17.3.1 of a Change in Control occurring following the conclusion of the Target Function Discovery Program Term, the Acquiring Party and the other party shall enter into good faith negotiations, for a period of [**] following such request, for the other party to obtain from the Acquiring Party the exclusive rights to Develop and commercialize any Collaboration Product. In the event the Acquiring Party and the other party do not enter into a definitive agreement with respect to such exclusive rights within such [**] period, except to the extent set forth in Section 17.3.5 below, the Acquiring Party and the other party shall each maintain their respective rights with respect to any such Collaboration Product as otherwise described in this Agreement.

17.3.5 [**] upon a Change in Control Occurring after the Target Function Discovery Program Term. In the event that (a) the Acquiring Party and the other party enter into negotiations in accordance with Section 17.3.4 with respect to a Collaboration Product but do not enter into a definitive agreement within the [**] period contemplated thereby and (b) the Acquiring Party (i) [**] and (ii) has not [**], the other party shall have the right, by delivering written notice thereof within [**] after the expiration of such [**] period, to [**] consisting of (i) [**] and (ii) [**]. [**] at the time such notice is delivered or as soon thereafter as [**] in accordance with the terms of this Section 17.3.5. Unless otherwise agreed by the Acquiring Party and the other party, [**], shall be determined by binding arbitration in accordance with the following provisions. The parties shall attempt to mutually agree upon a single independent Third Party arbitrator (who shall be a professional with appropriate experience in [**]) within [**] after the other party's delivery of the written notice in accordance with this Section 17.3.5 of [**]. If the parties are unable to mutually agree upon one such person, then each party shall appoint one independent Third Party professional with appropriate experience in [**] prior to the expiration of such [**] period, and within [**] after the initial notice, such person(s) shall select a single independent Third Party arbitrator, who shall be a professional with appropriate experience in [**]. Each party shall present all information as such party reasonably desires regarding [**]. Within [**] after the initial notice, the arbitrator shall provide written notice to the parties regarding his or her determination regarding [**].

17.4 Severability. Each party hereby agrees that it does not intend to violate any public policy, statutory or common laws, rules, regulations, treaty or decision of any government agency or executive body thereof of any country or community or association of countries. Should one or more provisions of this Agreement be or become invalid, the parties hereto shall substitute, by mutual consent, valid provisions for such invalid provisions which valid provisions in their economic effect are sufficiently similar to the invalid provisions that it can be reasonably assumed that the parties would have entered into this Agreement with such valid provisions in lieu of such invalid provisions. In case such valid provisions cannot be agreed upon, the invalidity of one or several provisions of this Agreement shall

not affect the validity of this Agreement as a whole, unless the invalid provisions are of such essential importance to this Agreement that it is to be reasonably assumed that the parties would not have entered into this Agreement without the invalid provisions.

17.5 Notices. Any consent, notice or report required or permitted to be given or made under this Agreement by one of the notification parties hereto to the other shall be in writing, delivered personally or by facsimile (and promptly confirmed by telephone, personal delivery or courier) or courier, postage prepaid (where applicable), addressed to such other party at its address indicated below, or to such other address as the addressee shall have last furnished in writing to the addressor and shall be effective upon receipt by the addressee.

If to Lexicon:	Lexicon Genetics Incorporated 8800 Technology Forest Place The Woodlands, Texas 77381 Attention: President and Chief Executive Officer Telephone: (281) 863-3000 Facsimile: (281) 863-8095
With copies to:	Lexicon Genetics Incorporated 8800 Technology Forest Place The Woodlands, Texas 77381 Attention: Chief Financial Officer Telephone: (281) 863-3000 Facsimile: (281) 863-8095
	Lexicon Genetics Incorporated 8800 Technology Forest Place The Woodlands, Texas 77381 Attention: General Counsel Telephone: (281) 863-3000 Facsimile: (281) 863-8010
If to Organon:	N.V. Organon Kloosterstraat 6 5349 AB Oss The Netherlands Attention: General Manager Telephone: 31 412 661 397 Facsimile: 31 412 663 529
With a copy to:	Director Legal Affairs Akzo Nobel Pharma B.V. Wethouder van Eschstraat 1 P.O. Box 20 5340 BH Oss The Netherlands
If to Intervet:	Intervet, Inc. 29160 Intervet Lane Millsboro, Delaware 19966

Attention:	Director of Business Development
Telephone:	(302) 934-4422
Facsimile:	(302) 933-4015

All such communications shall be effective upon receipt.

17.6 Applicable Law. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, United States of America, without reference to the conflicts of law principles thereof.

17.7 Dispute Resolution. Subject to Section 3.8.2, the parties hereby agree that they will first attempt in good faith to resolve any controversy or claim arising out of or relating to this Agreement promptly by negotiations. If a controversy or claim should arise hereunder, the matter shall be referred to an individual designated by the Chief Executive Officer (or the equivalent position) of Lexicon and by the Chief Executive Officer (or the equivalent position) of Organon (the "Representatives"). If the matter has not been resolved within [**] of the first meeting of the Representatives of the parties (which period may be extended by mutual agreement) concerning such matter, the parties shall be free to pursue all available recourse both at law and in equity.

17.8 Entire Agreement. This Agreement, together with the exhibits and appendices hereto and any confidentiality agreement(s) executed in contemplation of this Agreement, contains the entire understanding of the parties with respect to the subject matter hereof. All express or implied agreements and understandings, either oral or written, heretofore made are expressly merged in and made a part of this Agreement. This Agreement may be amended, or any term hereof modified, only by a written instrument duly executed by both parties hereto.

17.9 Publicity. Lexicon and Organon each agree not to disclose any terms or conditions of this Agreement to any Third Party without consulting the other party prior to such disclosure. Notwithstanding the foregoing, prior to execution of this Agreement, Lexicon and Organon shall agree upon the substance of information that can be used as a routine reference in the usual course of business to describe the existence and general nature of this transaction, and Lexicon and Organon may disclose such information without consulting the other party. The parties may thereafter from time to time mutually agree on revisions to material to be used as a routine reference, which revisions shall be submitted by one party for the review and approval of the other party at least [**] prior to the anticipated use or disclosure of the revised material, such approval not to be unreasonably withheld. The terms of this Agreement shall be treated as the Confidential Information of Lexicon and Organon, and, except to the extent required by applicable law, shall not be disclosed to anyone (except for the parties' respective employees, consultants, agents and attorneys assisting in the review and negotiation of this Agreement who have a need to know the terms of this Agreement) without the written permission of Organon or Lexicon. If either party desires to release a separate announcement relating to this Agreement, it shall first allow the other party to approve in writing such proposed announcement; provided that such approval shall not be unreasonably withheld or delayed.

17.10 Headings. The captions to the several Articles and Sections hereof are not a part of this Agreement, but are merely guides or labels to assist in locating and reading the several Articles and Sections hereof.

17.11 No Partnership. It is expressly agreed that the relationship between Lexicon and Organon shall not constitute a partnership, joint venture or agency. Neither Lexicon nor Organon shall have the authority to make any statements, representations or commitments of any kind, or to take any action, which shall be binding on the other, without the prior consent of the other party to do so.

17.12 Exports. The parties acknowledge that the export of technical data, materials or products is subject to the exporting party receiving any necessary export licenses and that the parties cannot be responsible for any delays attributable to export controls which are beyond the reasonable control of either party. Lexicon and Organon agree not to export or re-export, directly or indirectly, any information, technical data, the direct product of such data, samples or equipment received or generated under this Agreement in violation of any applicable export control laws or governmental regulations. Lexicon and Organon agree to obtain similar covenants from their licensees, (sub)licensees, or corporate partners, as the case may be, and contractors with respect to the subject matter of this Section 17.12.

17.13 Waiver. The waiver by either party hereto of any right hereunder or the failure to perform or of a breach by the other party shall not be deemed a waiver of any other right hereunder or of any other breach or failure by said other party whether of a similar nature or otherwise.

17.14 Counterparts. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

IN WITNESS WHEREOF, the parties have caused their duly authorized officers to execute and deliver this Agreement as of the Effective Date.

LEXICON GENETICS INCORPORATED

Date:	
Date:	
Date:	
Date:	
Date:	
	Date: Date:

EXHIBIT 1.55

LEVEL 1 PHENOTYPIC ANALYSIS

Level 1 Phenotypic Analysis is an initial screen designed to identify primary characteristics resulting from selected mutations in Mutant Mice. Level 1 Phenotypic Analysis currently includes the following assays, which may be changed from time to time at the Joint Research Committee's reasonable scientific discretion.

[**]

EXHIBIT 1.56

LEVEL 2 PHENOTYPIC ANALYSIS

Level 2 Phenotypic Analysis is an advanced screen designed to provide more detailed and focused data relating to primary characteristics identified as a result of Level 1 Phenotypic Analysis. The Joint Research Committee may determine that Level 2 Phenotypic Analysis be performed for any one or more Therapeutic Area(s). Level 2 Phenotypic Analysis currently includes (a) assays previously performed for Level 1 Phenotypic Analysis with respect to such Therapeutic Area(s) (utilizing greater numbers of Mutant Mice) and (b) additional assays represented by those described below, which may be changed from time to time at the Joint Research Committee's reasonable scientific discretion.

METABOLISM AND ENDOCRINOLOGY

[**]

CARDIOLOGY

[**]

NEUROLOGY

[**]

ONCOLOGY

[**]

IMMUNOLOGY

[**]

OPHTHALMOLOGY

[**]

EXHIBIT 7.1.1.2

MANUFACTURING AND SUPPLY AGREEMENT

THIS MANUFACTURING AND SUPPLY AGREEMENT (this "Agreement") is dated as of _____, 2005 (the "Effective Date") and is made by and among LEXICON GENETICS INCORPORATED, a corporation organized under the laws of the State of Delaware, United States of America ("Lexicon"), and N.V. ORGANON, a registered company organized under the laws of the Netherlands ("Organon"). Lexicon and Organon are sometimes referred to herein individually as a "party" and collectively as the "parties." All terms not otherwise defined herein shall have the meanings given to such terms in the Collaboration Agreement (as defined below) and all exhibits thereto.

RECITALS

WHEREAS, Lexicon and Organon are parties to that certain Collaboration and License Agreement dated as of May 16, 2005 (the "Collaboration Agreement") under which Lexicon and Organon are collaborating in the discovery and development of antibody and protein therapeutics for selected gene and protein targets;

WHEREAS, pursuant to the terms and conditions set forth in the Collaboration Agreement, unless otherwise specified by the Joint Management Committee or declined by Organon in accordance with the terms thereof, Organon or its Affiliated Subcontractor(s) shall be responsible for the Manufacture and supply of all quantities of any Collaboration Product necessary for commercialization;

WHEREAS, pursuant to the terms and conditions set forth in the Collaboration Agreement, Organon or its Affiliated Subcontractor(s) shall Manufacture and supply all quantities of the [insert Collaboration Product description] (the "Collaboration Product") necessary for commercialization;

NOW, THEREFORE, in consideration of the premises and of the covenants herein contained, the parties hereto mutually agree as follows:

ARTICLE 1. MANUFACTURING AND SUPPLY

1.1 Manufacturing and Supply Generally. Organon or its Affiliated Subcontractor(s) shall Manufacture and supply all quantities of the Collaboration Product necessary for commercialization and shall be responsible for implementing all aspects of Manufacturing under the direction and oversight of the Joint Management Committee, as set forth in the Collaboration Agreement and in accordance with the Manufacturing Plan established for the Collaboration Product. Organon and its Affiliated Subcontractor(s) shall use their Diligent Efforts to provide an adequate and timely supply of all properly forecasted requirements of the Collaboration Product in accordance with this Article 1.

1.2 Orders; Forecasts for Commercial Requirements. The party with principal responsibility for sales and distribution of the Collaboration Product (the "Responsible Party") will be responsible for generating periodic [**] forecasts of the anticipated requirements for such Collaboration Product for commercialization purposes and updates of such forecasts not less than [**] thereafter, such forecasts and updates to be promptly provided to Organon and its Affiliated Subcontractor(s). The Responsible Party shall provide to Organon and its Affiliated Subcontractor(s) a firm order for the amount of Collaboration Product to be delivered during each [**] following the filing of a BLA relating to the Collaboration Product, such firm order to be delivered no later than [**] prior to the requested delivery date for such order. Notwithstanding the foregoing, (i) if the total of such party's firm orders for delivery in any Contract Quarter is less than [**] of its most recent estimate for such Contract

required to purchase at least [**] of the estimate for such Contract Quarter and (ii) if the total of such party's firm orders for delivery in any Contract Quarter exceeds [**] of its most recent estimate for such Contract Quarter, Organon and its Affiliated Subcontractor(s) shall use Diligent Efforts, but shall have no obligation, to deliver quantities in excess of [**] of the estimate for such Contract Quarter. Organon and its Affiliated Subcontractor(s) shall package for shipment and ship all quantities of the Collaboration Product, samples or other materials in accordance with full written and reasonable instructions provided by the Responsible Party. Freight terms for such shipment shall be Ex Works according to Incoterms 2000. In the event that all of the Responsible Party's commercialization activities with respect to the Collaboration Product become the responsibility of a Third Party in accordance with the terms of the Collaboration Agreement, then the foregoing obligations of such party in this Section 1.2 shall become obligations of such Third Party with respect to the Collaboration Product; provided that in no event shall such party be relieved of any responsibility therefor unless Organon and its Affiliated Subcontractor(s) is a party to an agreement with such Third Party with respect to the commercialization of the Collaboration Product (in which case such party shall be relieved of all responsibility therefor).

1.3 Certain Covenants. Organon and its Affiliated Subcontractor(s) agree and covenant that it will (a) use Diligent Efforts to avoid shortfalls of supply based on the forecasts provided to it in accordance with Section 1.2, shall promptly notify the parties in the event it becomes aware of any probable shortfall and shall use Diligent Efforts to remedy any shortfall of supply as soon as practicable; (b) be responsible for Manufacturing, filling, packaging and warehousing of the Collaboration Product in conformity with applicable cGMP Requirements [**], and in accordance, in all material respects, with all other applicable Law; (c) maintain or cause to be maintained all records necessary and appropriate to demonstrate compliance with applicable cGMP Requirements and the Specifications; and (d) grant Lexicon the right, on reasonable advance notice and during normal business hours during the term of this Agreement, to have its personnel or representatives with quality control or quality assurance responsibilities inspect and audit the facilities and operations directly related to the Manufacture and supply of the Collaboration Product in order to confirm compliance with the covenants contained in this Section 1.3; provided that the foregoing inspection and audit right shall be limited to [**] and [**] per visit; and provided, further, that such personnel or representatives shall be subject to Organon's prior approval, such approval not to be unreasonably withheld.

ARTICLE 2. FINANCIAL TERMS

2.1 Payment. Either Lexicon or Organon, as determined by the Joint Management Committee, shall pay Organon's Affiliated Subcontractor(s) an amount equal to the Product Supply Costs for quantities of the Collaboration Product Manufactured and supplied by such Affiliated Subcontractor(s) hereunder, [**]. Payment for quantities of the Collaboration Product Manufactured and supplied during each [**] shall be due within [**] of receipt of the [**] report specified in Section 2.2, or earlier if invoiced pursuant to the Affiliated Subcontractor's standard practices based on delivery of the applicable product or services. Such Product Supply Costs shall be subject to review by the Joint Management Committee and subject to audit as provided in Section 2.3.

2.2 Reports and Records. Within [**] after each [**], Organon's Affiliated Subcontractor(s) shall furnish to Lexicon and Organon a written [**] report showing the Product Supply Costs for quantities of the Collaboration Product Manufactured and supplied by the Affiliated Subcontractor during such [**], including a reasonably detailed calculation of such Product Supply Costs. The Affiliated Subcontractor shall keep accurate books and records in connection with the Manufacture and supply of the Collaboration Product in a manner consistent with GAAP and in sufficient detail to permit accurate determination of all figures necessary for verification of Product Supply Costs for the Collaboration Product.

2.3 Audits. Upon the written request of Lexicon or Organon, the Affiliated Subcontractor shall permit an independent certified public accountant selected by the requesting party and acceptable to the Affiliated Subcontractor, which acceptance shall not be unreasonably withheld, to have access, at reasonable times and during normal business hours, to such records of the Affiliated Subcontractor as may be reasonably necessary to verify the accuracy of the Product Supply Costs for the Collaboration Product, in respect of any fiscal year ending not more than [**] prior to the date of such request. Each party shall use commercially reasonable efforts to schedule all such verifications within [**] after the requesting party makes its written request. All such verifications shall be conducted not more than [**]. The report of the requesting party's independent certified public accountant shall be made available to all parties. Each party agrees that all information subject to review under this Section 2.3 is confidential and that the party receiving such information shall cause its independent certified public accountant to retain all such information in confidence. The requesting party's independent certified public accountant shall only report to the requesting party as to the computation of Product Supply Costs and shall not disclose to the requesting party any other information of the other party.

2.4 Interest on Late Payments. The Affiliated Subcontractor(s) shall have the right to seek to collect interest on any payments that are not paid on or before [**] after the date such payments are due under this Agreement at a rate equal to [**], calculated on the total number of days payment is delinguent.

2.5 Manner of Payment. Payments to be made by a party to the Affiliated Subcontractor(s) under this Agreement shall be payable in the currency of the Affiliated Subcontractor(s) and shall be paid by bank wire transfer in immediately available funds to such bank account in the state and country in which such principal office is located as is designated in writing by the Affiliated Subcontractor from time to time.

ARTICLE 3. CONFIDENTIALITY

3.1 Nondisclosure Obligations.

3.1.1 General. Except as otherwise provided in this Article 3, during the term of this Agreement and for a period of [**] thereafter, each Receiving Party shall maintain the Confidential Information of each Disclosing Party in confidence and use it only for purposes specifically authorized under this Agreement.

3.1.2 Limitations. To the extent it is reasonably necessary or appropriate to fulfill its obligations or exercise its rights under this Agreement and subject to advance written notification to the Disclosing Party: (a) a party may disclose to Third Parties Confidential Information it is otherwise obligated not to disclose under this Section 3.1, to its Affiliates, (sub)licensees, consultants, outside contractors and clinical investigators, on a strict need-to-know basis for the purposes contemplated by this Agreement and on condition that such entities or persons agree to keep the Confidential Information confidential for the same time periods and to the same extent as such party is required to keep the Confidential Information confidential hereunder; and (b) a party or its (sub)licensees may disclose, using appropriate measures to preserve confidentiality, such Confidential Information to government or other regulatory authorities to the extent that such disclosure is reasonably necessary to Manufacture the Collaboration Product pursuant to this Agreement.

3.1.3 Required Disclosure. A Receiving Party may disclose Confidential Information pursuant to interrogatories, requests for information or documents, subpoena, civil investigative demand issued by a court or governmental agency or as otherwise required by law; provided, however, that the Receiving Party shall notify the Disclosing Party promptly upon receipt thereof, giving [**] the Disclosing Party sufficient advance notice to permit it to oppose, limit or seek confidential treatment for such disclosure; and provided, further, that the Receiving Party shall furnish only that portion of the Confidential Information which it is advised by counsel is legally required whether or not a protective order or other similar order is obtained by the Disclosing Party.

3.2 Injunctive Relief. The parties hereto understand and agree that remedies at law may be inadequate to protect against any breach of any of the provisions of this Article 3 by either party or their employees, agents, officers or directors or any other person acting in concert with it or on its behalf. Accordingly, each party shall be entitled to the granting of injunctive relief by a court of competent jurisdiction against any action that constitutes any such breach of this Article 3.

ARTICLE 4. REPRESENTATIONS AND WARRANTIES

4.1 Representations, Warranties and Covenants of Lexicon. Lexicon represents and warrants to and covenants with the other parties that:

4.1.1 Lexicon is a corporation duly organized, validly existing and in corporate good standing under the laws of the State of Delaware, United States of America;

4.1.2 Lexicon has the corporate and legal right, authority and power to enter into this Agreement;

4.1.3 Lexicon has taken all necessary action to authorize the execution, delivery and performance of this Agreement;

4.1.4 upon the execution and delivery of this Agreement, this Agreement shall constitute a valid and binding obligation of Lexicon, enforceable in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' and contracting parties' rights generally and except as enforceability may be subject to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law);

4.1.5 the performance of Lexicon's obligations under this Agreement will not conflict with its charter documents or result in a breach of any agreements, contracts or other arrangements to which it is a party; and

4.1.6 Lexicon will not during the term of this Agreement enter into any agreements, contracts or other arrangements that would be inconsistent with its obligations under this Agreement.

4.2 Representations, Warranties and Covenants of Organon. Organon represents and warrants to and covenants with the other parties that:

4.2.1 Organon is a corporation duly organized, validly existing and in corporate good standing under the laws of the Netherlands;

4.2.2 Organon has the corporate and legal right, authority and power to enter into this Agreement, and to extend the rights and licenses granted to Lexicon in this Agreement;

4.2.3 Organon has taken all necessary action to authorize the execution, delivery and performance of this Agreement;

4.2.4 upon the execution and delivery of this Agreement, this Agreement shall constitute a valid and binding obligation of Organon enforceable in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' and contracting parties' rights generally and except as enforceability may be subject to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law);

4.2.5 the performance of its obligations under this Agreement will not conflict with Organon's charter documents or result in a breach of any agreements, contracts or other arrangements to which it is a party; and

4.2.6 Organon will not after the Effective Date enter into any agreements, contracts or other arrangements that would be inconsistent with its obligations under this Agreement.

4.3 Warranty Disclaimer. EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN THIS AGREEMENT, NO PARTY MAKES ANY WARRANTY WITH RESPECT TO ANY COLLABORATION PRODUCT OR ANY OTHER SUBJECT MATTER OF THIS AGREEMENT, AND EACH PARTY HEREBY DISCLAIMS WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NON-INFRINGEMENT WITH RESPECT TO ANY AND ALL OF THE FOREGOING.

4.4 Limited Liability. EXCEPT AS SPECIFICALLY SET FORTH IN THIS AGREEMENT, NO PARTY WILL BE LIABLE WITH RESPECT TO ANY MATTER ARISING UNDER THIS AGREEMENT UNDER ANY CONTRACT, NEGLIGENCE, STRICT LIABILITY OR OTHER LEGAL OR EQUITABLE THEORY FOR ANY PUNITIVE, EXEMPLARY, INCIDENTAL OR CONSEQUENTIAL DAMAGES OR LOST PROFITS.

ARTICLE 5. EXPIRATION AND TERMINATION

5.1 Term of Agreement. The term of this Agreement shall commence on the Effective Date and shall continue until the expiration or termination of the Collaboration Agreement.

5.2 Events of Default. An Event of Default by a party shall have occurred upon (a) the occurrence of a material breach of this Agreement if such party fails to remedy such breach within [**] after written notice thereof by a non-breaching party ([**] in the event of a party's failure to make a payment required hereunder) or, if remediation of such breach in [**] is not practicable, if such party fails to commence and diligently pursue such remediation during such [**] period, or (b) the commencement of any proceeding in or for bankruptcy, insolvency, dissolution or winding up by or against such party that is not dismissed or otherwise disposed of within [**] thereafter.

5.3 Effect of an Event of Default. In the event of an Event of Default, any non-defaulting party that is not an Affiliate of the defaulting party shall have the right, at its option exercisable in its sole discretion, in addition to any other rights or remedies available to it at law or in equity and subject to the limitations set forth in Sections 4.4 and 6.6 hereof, to terminate this Agreement upon [**] notice thereof to the other party.

5.4 Effect of Expiration or Termination of Agreement. The expiration or termination of this Agreement shall not relieve the parties of any obligation accruing prior to such expiration or termination.

The provisions of Article 3 and Sections 6.2 through 6.6 hereof shall survive the expiration or termination of this Agreement.

ARTICLE 6. MISCELLANEOUS

6.1 Force Majeure. No party shall be held liable or responsible to any other party nor be deemed to have defaulted under or breached this Agreement for failure or delay in fulfilling or performing any obligation under this Agreement when such failure or delay is caused by or results from causes beyond the reasonable control of the affected party, including but not limited to fire, floods, embargoes, war, acts of war (whether war is declared or not), insurrections, riots, civil commotions, strikes, lockouts or other labor disturbances, acts of God or acts, omissions or delays in acting by any governmental authority; provided, however, that the party so affected shall use reasonable commercial efforts to avoid or remove such causes of nonperformance, and shall continue performance hereunder with reasonable dispatch whenever such causes are removed. Each party shall provide the other party with prompt written notice of any delay or failure to perform that occurs by reason of force majeure. The parties shall mutually seek a resolution of the delay or the failure to perform as noted above.

6.2 Assignment. This Agreement may not be assigned or otherwise transferred, in whole or in part, by any party without the consent of the other party; provided, however, that any party may, without such consent, assign its rights and obligations under this Agreement (i) to any Affiliate, or (ii) in connection with a merger, consolidation or sale of such portion of a party's assets that includes rights under this Agreement to an unrelated Third Party; provided, further, that such party's rights and obligations under this Agreement shall be assumed by its successor in interest in any such transaction and shall not be transferred separate from all or substantially all of its other business assets, including those business assets that are the subject of this Agreement. Any purported assignment in violation of the preceding sentence shall be void. Any permitted assigne shall assume all obligations of its assignor under this Agreement, unless the parties otherwise agree.

6.3 Severability. Each party hereby agrees that it does not intend to violate any public policy, statutory or common laws, rules, regulations, treaty or decision of any government agency or executive body thereof of any country or community or association of countries. Should one or more provisions of this Agreement be or become invalid, the parties hereto shall substitute, by mutual consent, valid provisions for such invalid provisions which valid provisions in their economic effect are sufficiently similar to the invalid provisions that it can be reasonably assumed that the parties would have entered into this Agreement with such valid provisions in lieu of such invalid provisions. In case such valid provisions cannot be agreed upon, the invalidity of one or several provisions of this Agreement shall not affect the validity of this Agreement as a whole, unless the invalid provisions are of such essential importance to this Agreement that it is to be reasonably assumed that the parties would not have entered into this Agreement without the invalid provisions.

6.4 Notices. Any consent, notice or report required or permitted to be given or made under this Agreement by one of the notification parties hereto to any other party shall be in writing, delivered personally or by facsimile (and promptly confirmed by telephone, personal delivery or courier) or courier, postage prepaid (where applicable), addressed to such other party at its address indicated below, or to such other address as the addressee shall have last furnished in writing to the addressor and shall be effective upon receipt by the addressee.

- If to Lexicon: Lexicon Genetics Incorporated 8800 Technology Forest Place The Woodlands, Texas 77381 Attention: President and Chief Executive Officer Telephone: (281) 863-3000 Facsimile: (281) 863-8095
- With a copy to: Lexicon Genetics Incorporated 8800 Technology Forest Place The Woodlands, Texas 77381 Attention: General Counsel Telephone: (281) 863-3000 Facsimile: (281) 863-8010
- If to Organon: N.V. Organon Kloosterstraat 6 5349 AB Oss The Netherlands Attention: General Manager Telephone: 31 412 661 397 Facsimile: 31 412 663 529
- With copies to: Akzo Nobel Pharma Wethouder van Eschstraat 1 P.O. Box 20 5340 BH Oss The Netherlands Attention: Director Legal Affairs

Organon International Inc. 56 Livingston Ave. Roseland, New Jersey 07068 Attention: General Counsel

All such communications shall be effective upon receipt.

6.5 Applicable Law. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, United States of America, without reference to the conflicts of law principles thereof.

6.6 Dispute Resolution. Subject to Section 3.8.2 of the Collaboration Agreement, the parties hereby agree that they will first attempt in good faith to resolve any controversy or claim arising out of or relating to this Agreement promptly by negotiations. If a controversy or claim should arise hereunder, the matter shall be referred to an individual designated by the Chief Executive Officer (or the equivalent position) of Lexicon and by the Chief Executive Officer (or the equivalent position) of Organon (the "Representatives"). If the matter has not been resolved within [**] of the first meeting of the Representatives of the parties (which period may be extended by mutual agreement) concerning such matter, the parties shall be free to pursue all available recourse both at law and in equity.

6.7 Entire Agreement. This Agreement, together with the exhibits and appendices hereto and any confidentiality agreement(s) executed in contemplation of this Agreement, contains the entire

understanding of the parties with respect to the subject matter hereof. All express or implied agreements and understandings, either oral or written, heretofore made are expressly merged in and made a part of this Agreement. In the event of any discrepancy between the terms and conditions contained herein and those contained in the Collaboration Agreement, the terms and conditions of the Collaboration Agreement shall control. This Agreement may be amended, or any term hereof modified, only by a written instrument duly executed by both parties hereto.

6.8 Publicity. Each party agrees not to disclose any terms or conditions of this Agreement to any Third Party without consulting the other parties prior to such disclosure. Notwithstanding the foregoing, prior to execution of this Agreement, the parties shall agree upon the substance of information that can be used as a routine reference in the usual course of business to describe the existence and general nature of this transaction, and the parties may disclose such information without consulting the other party. The parties may thereafter from time to time mutually agree on revisions to material to be used as a routine reference, which revisions shall be submitted by one party for the review and approval of the other parties at least [**] prior to the anticipated use or disclosure of the revised material, such approval not to be unreasonably withheld. The terms of this Agreement shall be treated as the Confidential Information of the parties, and, except to the extent required by applicable law, shall not be disclosed to anyone (except for the parties' respective employees, consultants, agents and attorneys assisting in the review and negotiation of this Agreement who have a need to know the terms of this Agreement) without the written permission of the other parties. If any party desires to release a separate announcement relating to this Agreement, it shall first allow the other parties to approve in writing such proposed announcement; provided that such approval shall not be unreasonably withheld or delayed.

6.9 Headings. The captions to the several Articles and Sections hereof are not a part of this Agreement, but are merely guides or labels to assist in locating and reading the several Articles and Sections hereof.

6.10 No Partnership. It is expressly agreed that the relationship between the parties shall not constitute a partnership, joint venture or agency. No party shall have the authority to make any statements, representations or commitments of any kind, or to take any action, which shall be binding on another party, without the prior consent of the other party to do so.

6.11 Exports. The parties acknowledge that the export of technical data, materials or products is subject to the exporting party receiving any necessary export licenses and that the parties cannot be responsible for any delays attributable to export controls which are beyond the reasonable control of either party. The parties agree not to export or re-export, directly or indirectly, any information, technical data, the direct product of such data, samples or equipment received or generated under this Agreement in violation of any applicable export control laws or governmental regulations. The parties agree to obtain similar covenants from their licensees, (sub)licensees, or corporate partners, as the case may be, and contractors with respect to the subject matter of this Section 6.11.

6.12 Waiver. The waiver by any party hereto of any right hereunder or the failure to perform or of a breach by the other party shall not be deemed a waiver of any other right hereunder or of any other breach or failure by said other party whether of a similar nature or otherwise.

6.13 Counterparts. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the parties have caused their duly authorized officers to execute and deliver this Agreement as of the Effective Date.					
LEXICON GENETICS INCORPORATED					
Ву:	Date:				
Name:					
Title:					
N.V. ORGANON					
Ву:	Date:				
Name:					
Title:					

JOINT COMMERCIALIZATION AGREEMENT

THIS JOINT COMMERCIALIZATION AGREEMENT (this "Agreement") is dated as of _____, ____ (the "Effective Date") and is made by and between LEXICON GENETICS INCORPORATED, a corporation organized under the laws of the State of Delaware, United States of America ("Lexicon"), and N.V. ORGANON, a registered company under the laws of the Netherlands ("Organon"). [Affiliates of Lexicon and Organon may be designated as the signatories to this Agreement] Lexicon and Organon are sometimes referred to herein individually as a "party" and collectively as the "parties."

RECITALS

WHEREAS, Lexicon and Organon are parties to a Collaboration and License Agreement dated May 16, 2005 (the "Collaboration Agreement");

WHEREAS, Lexicon and Organon are Developing the Collaboration Product described in Schedule A hereto (the "Licensed Product"); and

WHEREAS, Lexicon and Organon desire to collaborate in the commercialization of the Licensed Product in the Collaboration Field in the Co-Commercialization Territory (as defined herein);

NOW, THEREFORE, in consideration of the premises and of the covenants herein contained, the parties hereto mutually agree as follows:

ARTICLE 1. DEFINITIONS

For purposes of this Agreement, the terms defined in this Article 1 shall have the respective meanings specified below:

1.1 "Acquiring Party" has the meaning specified in Section 11.3 hereof.

1.2 "Affiliate" means any corporation, company, partnership, joint venture or firm that controls, is controlled by or is under common control with a party to this Agreement. For purposes hereof, "control" means (a) in the case of a corporate entity, direct or indirect ownership of more than fifty percent (50%) of the stock or shares entitled to vote for the election of directors; and (b) in the case of a non-corporate entity, direct or indirect ownership of more than fifty percent (50%) of the equity interests with the power to direct the management and policies of such non-corporate entity.

1.3 "Allocable Overhead" means costs incurred by a party with respect to the Licensed Product that (a) are attributable to such party's [**] and (b) are allocated to company departments based on [**]. Allocable Overhead shall not include any costs attributable to [**].

1.4 "Allowable Commercialization Costs" means the sum of the following (without any item being accounted for more than once): (a) Cost of Goods Sold for Commercialization (without duplication of Product Supply Costs for Development under the Collaboration Agreement); (b) Regulatory and Licensing Costs; (c) Distribution Costs; and (d) Sales and Marketing Costs.

1.5 "Background Materials" has the meaning specified in the Collaboration Agreement.

1.6 "Background Technology" has the meaning specified in the Collaboration Agreement.

1.7 "BLA" means a Biologics Licensing Application (as defined in the FDC Act) filed with the FDA and any other equivalent marketing authorization application or other license, registration or other application seeking approval from a Regulatory Authority to market the Licensed Product in the Collaboration Field in any country or region within the Co-Commercialization Territory.

1.8 "Bundled Transaction" has the meaning specified in Section 1.40 hereof.

1.9 "Change in Control" has the meaning specified in the Collaboration Agreement.

1.10 "Co-Commercialization Territory" means [all of the countries of the world other than those for which rights have been granted to a Joint Marketing/Development Collaborator]. For clarity, the Co-Commercialization Territory shall not include any country for which rights have been granted to a Joint Marketing/Development Collaborator from and after the effectiveness of the parties' grant of such rights to such Joint Marketing/Development Collaborator.

1.11 "Collaboration Field" has the meaning specified in the Collaboration Agreement.

1.12 "Combination Product" has the meaning specified in Section 1.40 hereof.

1.13 "Commercialization" or "Commercialize" means any and all activities associated with marketing, promoting, communicating (including medical communications and publishing), distributing, importing, exporting or selling the Licensed Product in the Collaboration Field as set forth in the applicable Commercialization Plan, including the conduct of any activities (including any Post-approval Studies) directed to obtaining pricing and reimbursement approvals and any other Post-approval Studies not included in Development, in each case by a party, its Affiliates or (sub)licensees.

1.14 "Commercialization Plan" means the plan to be developed by the Joint Commercialization Committee and approved by the Joint Management Committee for each Contract Year with respect to the Licensed Product in accordance with Section 4.2 hereof.

1.15 "Confidential Information" means any information and data received by a party or its Affiliates (the "Receiving Party") from the other party or its Affiliates (the "Disclosing Party") in connection with this Agreement (including, without limitation, any research, testing, clinical, regulatory, marketing or other scientific or business information, plans, or data pertaining to the Licensed Product). Notwithstanding the foregoing, Confidential Information shall not include any part of such information or data that:

(a) is or becomes part of the public domain other than by unauthorized acts of the Receiving Party or its Affiliates;

(b) can be shown by written documents to have been already in the possession of the Receiving Party or its Affiliates prior to disclosure under this Agreement, provided such Confidential Information was not obtained directly or indirectly from the Disclosing Party under an obligation of confidentiality;

(c) can be shown by written documents to have been disclosed to the Receiving Party or its Affiliates by a Third Party, provided such Confidential Information was not obtained directly or indirectly from the Disclosing Party under an obligation of confidentiality; or (d) can be shown by written documents to have been independently developed by the Receiving Party or its Affiliates without use of, or access to, Confidential Information of the Disclosing Party.

Specific Confidential Information of a Disclosing Party shall not be deemed to come under the foregoing exceptions merely because it is embraced by more general information that is or becomes part of the public domain, or is known by, disclosed to or independently developed by the Receiving Party.

1.16 "Contract Year" means (a) with respect to the first Contract Year, the period beginning on the Effective Date and ending on December 31, _____ (the "First Contract Year"), and (b) with respect to each subsequent Contract Year, the twelve (12) month period beginning on the day following the end of the First Contract Year and each succeeding twelve (12) month period thereafter. Each Contract Year (other than the First and last Contract Year, as applicable) shall be divided into four (4) "Contract Quarters" comprised of successive three (3) month periods. In the First Contract Year, the first Contract Quarter shall begin on the Effective Date and end on _____.

1.17 "Control" or "Controlled" has the meaning specified in the Collaboration Agreement.

1.18 "Cost and Revenue Sharing Ratio" means (a) with respect to Lexicon, [fifty percent (50%)], and (b) with respect to Organon, [fifty percent (50%)]. [Ratio in effect under Collaboration Agreement when Joint Commercialization Agreement is signed]

"Cost of Goods Sold" means (a) to the extent that the Licensed 1.19 Product is sourced from a party, the unit cost of manufacture of the Licensed Product (i.e., direct material and direct labor costs, plus manufacturing overhead fairly allocated to the Licensed Product, all calculated in accordance with GAAP), and (b) to the extent that the Licensed Product is sourced from [the applicable Organon Affiliated Subcontractor under that certain Manufacturing and Supply Agreement dated ____] [and/or] [a Third Party, subject to Section 2.2.3], the actual price paid to such [Organon Affiliated Subcontractor] [and/or] [Third Party] for the manufacture, supply and packaging of the Licensed Product. For purposes of the foregoing, (i) "direct material costs" means actual costs incurred in manufacturing or purchasing materials, including freight-in costs, sales and excise taxes imposed thereon and customs duty and charges levied by government authorities, and all costs of packaging components; (ii) "direct labor costs" means actual cost of employees engaged in direct manufacturing activities and quality control and quality assurance activities who are directly employed in manufacturing and packaging the Licensed Product; and (iii) "manufacturing overhead" attributable to the Licensed Product will include a reasonable allocation of indirect labor (not previously included in direct labor costs), a reasonable allocation of administrative costs, and a reasonable allocation of facilities costs, all in accordance with GAAP, but will not include corporate administrative overhead or plant start-up costs or costs associated with excess capacity. All allocations will be based on the assumption that such party's plant and equipment are utilized to their reasonable full capacity, and all costs and allocations shall be consistent with the methods used for such costs and allocations for such party's internal purposes. More specifically, the components of Cost of Goods Sold shall comprise: (A) direct labor (fermentation, purification personnel); (B) direct materials; (C) facility costs (rent, property taxes, depreciation of leaseholds, utilities, spare parts, maintenance contracts); (D) manufacturing equipment depreciation; (E) allocations for information technology, document control, quality engineering, purchasing, warehouse management, microbiology (with such allocations to be based on estimated service levels, headcount or square footage occupancy, depending on the category); (F) indirect labor (manufacturing supervision); (G) manufacturing department overhead (uniforms, materials used in plant maintenance); (H) quality assurance/quality control; and (I) such other similar costs as may be reasonably included in such definition.

1.20 "Detail" means a sales presentation by a professional sales representative to a target physician involved in prescribing the Licensed Product in which the primary purpose is to discuss the benefits and features of the Licensed Product.

1.21 "Detail Costs" means the direct compensation and benefits paid to professional sales representatives making Details.

1.22 "Development" or "Develop" has the meaning specified in the Collaboration Agreement.

1.23 "Diligent Efforts" has the meaning specified in the Collaboration Agreement.

1.24 "Disclosing Party" has the meaning specified in Section 1.15 hereof.

1.25 "Distribution Costs" means a party's reasonable costs and expenses (including direct labor) related to storage and distribution of the Licensed Product incurred in accordance with a budget approved by the Joint Management Committee, including: (a) handling and transportation to fulfill orders; (b) customer services, including order entry, billing and adjustments, inquiry and credit and collection; (c) cost of facilities and labor utilized for the storage or distribution of the Licensed Product; and (d) amounts paid to Third Parties in respect of storage or distribution of the Licensed Product.

1.26 "Effective Date" means the date specified in the initial paragraph of this Agreement.

1.27 "Event of Default" means an event described in Section 10.2 hereof.

1.28 "FDA" has the meaning specified in the Collaboration Agreement.

1.29 "FDC Act" has the meaning specified in the Collaboration Agreement.

1.30 "GAAP" means United States or international generally accepted accounting principles, as they exist from time to time, consistently applied.

1.31 "Joint Commercialization Committee" has the meaning specified in Section 3.1.2 hereof.

1.32 "Joint Management Committee" has the meaning specified in Section 3.1.1 hereof.

1.33 "Joint Marketing/Development Collaborator" has the meaning specified in the Collaboration Agreement.

1.34 "Joint Marketing/Development Collaborator Revenue" has the meaning specified in the Collaboration Agreement.

1.35 "[**]" means [**].

1.36 "Laws" means all laws, statutes, rules, regulations, ordinances and other pronouncements having the effect of law of any federal, national, multinational, state, provincial, county, city or other political subdivision, domestic or foreign.

1.37 "Lexicon Background Materials" has the meaning specified in the Collaboration Agreement.

1.38 $\,$ "Lexicon Background Technology" has the meaning specified in the Collaboration Agreement.

1.39 "Manufacturing" or "Manufacture" has the meaning specified in the Collaboration Agreement.

1.40 "Net Sales" means the gross amount invoiced for sales of the Licensed Product by Lexicon, Organon and their respective Affiliates to customers which are not Affiliates (or which are Affiliates but are end users of the Licensed Product), less:

(a) trade, quantity and cash discounts actually allowed;

(b) discounts, refunds, rebates, chargebacks, retroactive price adjustments, billing errors and any other allowances (including, without limitation, government-mandated and managed health care-negotiated rebates) actually granted which effectively reduce the net sales price;

(c) product returns credits and allowances actually
granted;

(d) any tax imposed on the production, sale, delivery or use of the product (excluding federal, state or local taxes based on income);

(e) freight, postage, shipping, customs duties, excises, tariffs, surcharges, other governmental charges (excluding federal, state or local taxes based on income) and insurance charges actually allowed or paid for delivery of the product;

(f) payments or rebates paid with respect to the Licensed Product in connection with state or federal Medicare, Medicaid or similar programs in the United States or in connection with similar programs in other countries in which there are sales; and

(g) adjustments for bad debts.

Such amounts shall be determined from the books and records of Lexicon, Organon and their respective Affiliates maintained in accordance with GAAP.

In the event the Licensed Product is sold as part of a Combination Product (as defined below), the Net Sales of the Licensed Product attributable to the Combination Product will be determined by multiplying the Net Sales of the Combination Product by the fraction, A/A+B where A is the average sales price of the Licensed Product when sold separately in finished form and B is the average sales price of the other active compounds or ingredients in the Combination Product sold separately in finished form.

In the event that the average sale price of the Licensed Product can be determined but the average sales price of the other active compounds or ingredients cannot be determined, the Net Sales of the Licensed Product attributable to the Combination Product will be determined by multiplying the Net Sales of the Combination Product by the fraction C/C+D where C is the selling party's average sales price of the Licensed Product and D is the difference between the average sales product. If the average sales price of the other active compounds or ingredients can be determined but the average sales price of the Licensed Product. Sales of the Licensed Product attributable to the Combination Product will be determined by multiplying the Net Sales of the Combination Product by the following formula: one minus C/C+D where C is the average sales price of the combination Product and D is the difference between the average sales price of the combination Product and the average sales price of the Combination Product will be determined by multiplying the Net Sales of the Combination Product by the following formula: one minus C/C+D where C is the average sales price of the combination Product and the average sales price of the other active compounds or ingredients.

In the event that the average sales price of both the Licensed Product, as applicable, and the other active compounds or ingredients in the Combination Product cannot be determined, the Net Sales of the Licensed Product attributable to the Combination Product shall be negotiated in good faith by the parties.

The Net Sales price for a Combination Product in a given country will be calculated once each Contract Year and such price will be used during all applicable revenue reporting periods for the entire Contract Year for such country, absent extraordinary conditions or events. When determining the average sales price of the Licensed Product or the other active compounds or ingredients in the Combination Product, the average sales price will be calculated using data arising from the twelve (12) months preceding the calculation of the Net Sales price for the Combination Product. As used above, the term "Combination Product" means the Licensed Product sold in conjunction with any other active component(s) (whether packaged together or in the same therapeutic formulation).

If Lexicon, Organon or their Affiliates sells the Licensed Product to a customer which also purchases other products or services from such seller in a bundled, combination or capitated transaction (a "Bundled Transaction"), and such seller discounts the sales price of the Licensed Product to a greater degree than such seller generally discounts the price of its other products to such customer, then the aggregate amount received with respect to such Bundled Transaction shall be allocated to Net Sales pursuant to the formula set forth in Exhibit 1.40 hereto. For purposes of the foregoing, "discounting" includes establishing the list price at lower than the seller's normal pricing level.

Free samples of Licensed Product and the disposition of Licensed Product for, or the use of Licensed Product in Post-approval Studies in which such Licensed Product is provided to patients without any payment shall not result in any Net Sales.

1.41 "Operating Profit/Loss" means (a) Net Sales of the Licensed Product in the Collaboration Field in the Co-Commercialization Territory during the relevant time period less (b) the Allowable Commercialization Costs incurred by Lexicon and Organon in respect of the Licensed Product during such time period. For clarity, Operating Profit/Loss shall not include (i) Net Sales or Regulatory and Licensing Costs from and after the effectiveness of a party's Opting Out of the Licensed Product or (ii) Joint Marketing/Development Collaborator Revenue.

1.42 "Opt Out" has the meaning specified in the Collaboration Agreement.

1.43 "Organon Background Materials" has the meaning specified in the Collaboration Agreement.

1.44 "Organon Background Technology" has the meaning specified in the Collaboration Agreement.

1.45 "Patent Rights" has the meaning specified in the Collaboration Agreement.

1.46 "Post-approval Study" means a clinical trial conducted after Regulatory Approval of the Licensed Product for the applicable indication has been obtained in the relevant country.

1.47 "Pre-existing Obligations" has the meaning specified in the Collaboration Agreement.

1.48 "Product Collaboration" has the meaning specified in Section 2.1.1 hereof.

1.49 "Product Collaboration Committee" means the Joint Management Committee or Joint Commercialization Committee.

1.50 "Product Trademark" means one or more trademarks or logos that are used for the Commercialization of the Licensed Product in the Collaboration Field in the Co-Commercialization Territory.

1.51 "Program Director" has the meaning specified in Section 3.2 hereof.

1.52 "Program Intellectual Property" has the meaning specified in the Collaboration Agreement.

1.53 "Receiving Party" has the meaning specified in Section 1.15 hereof.

1.54 "Regulatory Approval" means any and all approvals (including any applicable governmental price and reimbursement approvals), licenses, registrations, or authorizations of any federal, national, multinational, state, provincial or local regulatory agency, department bureau or other governmental entity that are necessary for the manufacture, use, storage, import, transport, promotion, marketing and sale of the Licensed Product in the Collaboration Field in a country or group of countries within the Co-Commercialization Territory.

1.55 "Regulatory Authority" has the meaning specified in the Collaboration Agreement.

"Regulatory and Licensing Costs" means the following costs and 1.56 expenses attributable to Commercialization of the Licensed Product that are incurred in accordance with an applicable Commercialization Plan and associated budget approved by the Joint Management Committee, including direct labor costs and Allocable Overhead: (a) costs and expenses related to Post-approval Studies related to the Licensed Product; (b) infrastructure required to support and maintain patient/safety surveillance as required by applicable Regulatory Authorities attributable to the Licensed Product, including medical staff support and pharmacovigilance systems and procedures; (c) out-of-pocket costs and expenses of maintaining Regulatory Approvals; (d) royalty and other amounts paid to Third Parties related to the Commercialization of the Licensed Product under Third Party licenses entered into (i) prior to the Effective Date and disclosed to the other party prior to the Effective Date or (ii) in accordance with the Collaboration Agreement; (e) the costs of product recalls, product liability claims, awards and damages which are not subject to the indemnities set forth in Article 9, and Licensed Product liability insurance premiums; and (f) costs and expenses incurred in connection with pricing and reimbursement matters, managed care and formulary management, and governmental affairs activities directly relating to the Licensed Product.

1.57 "Sales and Marketing Costs" means all costs and expenses attributable to the distribution, sale, promotion and marketing of the Licensed Product other than Cost of Goods Sold, Regulatory and Licensing Costs, Distribution Costs and Detail Costs. Sales and Marketing Costs include the following:

> (a) all costs and expenses associated with advertising of the Licensed Product;

(b) all costs and expenses associated with programs to promote the Licensed Product directly to the prescriber or end user, including expenses associated with promoting the Licensed Product directly to the professional community such as professional samples, professional literature, promotional material costs, patient aids and detailing aids;

(c) all costs and expenses associated with professional education with respect to the Licensed Product through any means, including articles appearing in journals, newspapers, magazines or other media; seminars, and conventions; symposia, advisory boards and opinion leader development activities; and the costs and expense of medical liaisons; (d) all compensation and departmental expenses for market and consumer research personnel and payments to Third Parties related to conducting and monitoring professional and consumer appraisals of existing, new or proposed competitors to the Licensed Product, such as market share services (e.g., IMS data), special research testing and focus groups;

(e) all product management and sales promotion management compensation and departmental costs and expenses, including costs associated with developing overall sales and marketing strategies (e.g., product line or customer segment), as well as planning and programs for the Licensed Product; and

(f) all field sales force management and departmental costs and expenses, including costs associated with field sales offices; district, regional and home offices; staffs directly involved in the management of the selling functions; sales training and meetings; and call reporting and other monitoring/tracking activities.

1.58 "Third Party" means any person or entity other than Lexicon, Organon and their respective Affiliates.

1.59 "Valid Claim" has the meaning specified in the Collaboration Agreement.

ARTICLE 2. COLLABORATION OVERVIEW

2.1 General. The parties intend to carry out a program in which Lexicon and Organon will collaborate to Commercialize the Licensed Product in the Collaboration Field in the Co-Commercialization Territory (the "Collaboration"), consistent with the objectives set forth in and the resources allocated to such activities in the applicable Commercialization Plan(s). It is intended that the Collaboration will be conducted as a unified collaborative effort, and this intent shall be reflected in the applicable Commercialization Plan(s). It is intended that each party shall bear its own Detail Costs; provided, however, that one party may retain the other party at a commercially reasonable rate to perform all or a part of its obligations with respect to Details. It is further intended that each party shall contribute to Allowable Commercialization Costs in proportion to the Cost and Revenue Sharing Ratio, and the Commercialization Plan(s) will be consistent with and provide for such proportional contribution.

2.2 Conduct of Collaboration.

2.2.1 Efforts. The Joint Management Committee shall adopt project progression guidelines for Commercialization activities. The parties shall conduct the Collaboration in good scientific manner in accordance with such project progression guidelines and in compliance with applicable Laws. Each party shall use Diligent Efforts to conduct the activities of the Collaboration that are assigned to it in the then-applicable Commercialization Plan(s), and each shall devote sufficient resources to carry out such respective activities. While the parties acknowledge and agree that neither party guarantees the success of the Collaboration or any individual task undertaken thereunder, each party agrees that it will perform the activities assigned to it under the Collaboration in a professional manner in accordance with the highest industry standards.

2.2.2 Personnel. The parties agree to commit to the Collaboration the personnel necessary to meet their respective responsibilities set forth in each Commercialization Plan.

2.2.3 Subcontractors. As necessary and in furtherance of the Collaboration, either party may enter into agreements or subcontracts for Commercialization activities in accordance with

this Section 2.2.3; provided that (a) none of the rights of the other party hereunder are diminished or otherwise adversely affected as a result of such subcontracting, (b) such party obtains the written approval of the other party prior to engaging any subcontractor, which approval shall not be unreasonably withheld or delayed (for purposes of which it shall not be deemed unreasonable for a party to withhold consent when the withholding party is capable of conducting the activities proposed to be conducted by such subcontractor and is willing to do so on terms, including cost, time and quality, equivalent to those offered by such contractor), and (c) the subcontractor undertakes in writing obligations of confidentiality and non-use regarding the other party's Confidential Information that are substantially the same as those undertaken by Organon and Lexicon pursuant to Article 7 hereof. In the event a party performs one or more of its obligations under the Collaboration through a subcontractor, then such party shall at all times be responsible for the performance of such subcontractor. The Joint Management Committee shall decide the allocation of the cost of any such agreement between the parties or if the cost is to be borne by one party and whether it can be allocated to offset other obligations set forth in the applicable Commercialization Plan(s).

2.2.4 Reports. Each party shall submit [**] reports to the Joint Management Committee, as may be required by the then-current Commercialization Plan(s), detailing its activities under the Collaboration. The Joint Management Committee shall use such [**] reports to monitor the parties' respective contributions to the Collaboration. The Joint Management Committee may amend the Commercialization Plan(s) as necessary to maintain substantial compliance over the course of the Collaboration in resources devoted and participation by the parties in accordance with the principles and objectives set forth in Section 2.2.2.

2.3 Conduct of Post-approval Studies. All Post-approval Studies shall be conducted and the results thereof recorded and reported to the parties in accordance with the terms and conditions set forth in Sections 2.5 through 2.7 of the Collaboration Agreement. Any inventions, information, methods, know-how, trade secrets or data that (a) are Controlled by a party or jointly by the parties and (b)(i) relate to the use of the Licensed Product or (ii) are first identified or discovered in the conduct of Post-approval Studies shall be deemed to be Program Technology and shall be subject to the provisions of Article 12 of the Collaboration Agreement.

2.4 Opt Out Rights. The rights of either party to Opt Out with respect to the Licensed Product shall be governed by Article 10 of the Collaboration Agreement.

ARTICLE 3. COLLABORATION MANAGEMENT

3.1 Product Collaboration Committees.

3.1.1 Joint Management Committee. As soon as practicable after the Effective Date, Organon and Lexicon shall establish a Joint Management Committee (the "Joint Management Committee") comprised of [**] representatives designated by Organon and [**] representatives designated by Lexicon, each of whom shall have experience and seniority sufficient to enable him or her to make decisions on behalf of the party he or she represents; provided that Organon and Lexicon may, by mutual agreement, designate an appropriate number of additional representatives from time to time.

3.1.2 Joint Commercialization Committee. As soon as practicable after the Effective Date, Organon and Lexicon shall establish a Joint Commercialization Committee (the "Joint Commercialization Committee") comprised of [**] representatives designated by Organon and [**] representatives designated by Lexicon, each of whom shall have experience and seniority sufficient to enable him or her to make decisions on behalf of the party he or she represents; provided that Organon and Lexicon may, by mutual agreement, designate an appropriate number of additional representatives from time to time

3.2 Program Directors. Each party shall appoint one of its designees on the Joint Management Committee (and who may, but need not also be, a member of the Joint Commercialization Committee) to serve as a program director (each, a "Program Director") with responsibility for overseeing the day-to-day activities of the parties with respect to the Collaboration and for being the primary point of contact between the parties with respect to the Collaboration.

3.3 Replacement of Product Collaboration Committee Representatives and Program Directors. Each party shall be free to replace its representative members of any Product Collaboration Committee and its Program Director with new appointees who have authority to act on behalf of such party, on notice to the other party.

3.4 Responsibilities of Joint Management Committee. The Joint Management Committee shall be responsible for overseeing and directing the parties' interaction and performance of their respective obligations under this Agreement. Without limiting the generality of the foregoing, its duties shall include:

> (a) preparing such procedures as may be necessary for the operation of the Joint Management Committee and Joint Commercialization Committee, and other committees the Joint Management Committee decides to establish to assure the efficient operation of the Collaboration;

(b) approving strategy for the overall Commercialization of the Licensed Product in the Collaboration Field in the Co-Commercialization Territory and for all other activities conducted by the parties hereunder;

(c) reviewing and approving the annual Commercialization Plans proposed by the Joint Commercialization Committee and approving any modifications thereto as recommended by the Joint Commercialization Committee;

(d) in connection with such annual Commercialization Plans, approving the budget(s) for Regulatory and Licensing Costs and Third Party Sales and Marketing Costs associated therewith and approving any modifications thereto as recommended by the Joint Commercialization Committee;

(e) overseeing the implementation of the Commercialization Plans and allocation of resources and other activities in support of the Collaboration, including the matters contemplated by Section 2.2 hereof;

(f) coordinating with the Joint Management Committee under the Collaboration Agreement with respect to (i) deciding whether the Licensed Product should be Commercialized by or through a Joint Marketing/Development Collaborator in any country or countries within the Co-Commercialization Territory; (ii) overseeing the progress of the applicable Development Plan(s) for the Licensed Product under the Collaboration Agreement and evaluating such progress against their respective timelines; and (iii) overseeing Manufacturing and supply of the Licensed Product;

(g) resolving matters within the responsibilities of the Joint Commercialization Committee as to which the members thereof are unable to reach a consensus; and

(h) addressing issues and resolving differences that may arise between the parties.

The Joint Management Committee shall not have the power to amend the terms of or waive compliance with this Agreement.

3.5 Responsibilities of Joint Commercialization Committee. The Joint Commercialization Committee shall be responsible for preparing for approval by the Joint Management Committee and implementing the Commercialization Plans. Without limiting the generality of the foregoing, its duties shall include:

> (a) proposing to the Joint Management Committee the Commercialization strategy for the Licensed Product throughout the Co-Commercialization Territory;

(b) reviewing and overseeing the activities of the parties with respect to the Commercialization of the Licensed Product in such countries and regions;

(c) providing all appropriate information regarding the progress of the Commercialization Plan(s) to the Joint Management Committee in advance of each [**] Joint Management Committee meeting; and

(d) performing such other activities as are contemplated by the terms of this Agreement.

The Joint Commercialization Committee shall report its activities and make proposals to the Joint Management Committee at least once each [**], but more frequently as appropriate. The Joint Commercialization Committee shall not have the power to amend or waive compliance with this Agreement.

3.6 Meetings of Product Collaboration Committees. As applicable, each Product Collaboration Committee shall meet at least [**], and more frequently as the parties deem appropriate, on such dates and at such times as the parties shall agree, on [**] written notice to the other party unless such notice is waived by the parties. The first meeting of the Joint Management Committee shall take place within [**] after the Effective Date, at Lexicon's facility in The Woodlands, Texas, United States of America. Each Product Collaboration Committee may convene or be polled or consulted from time to time by means of telecommunications, videoconferences or correspondence, as deemed necessary or appropriate by the parties. To the extent that meetings are held in person, they shall alternate between the offices of the parties unless the parties otherwise agree.

3.7 Decisions.

3.7.1 Quorum; Voting. A quorum for a meeting of a Product Collaboration Committee shall require the presence of at least one Lexicon member (or designee) and at least one Organon member (or designee) in person or by telephone. All decisions made or actions taken by a Product Collaboration Committee shall be made unanimously by its members, with the Lexicon members cumulatively having one vote and the Organon members cumulatively having one vote.

3.7.2 Dispute Resolution.

3.7.2.1 In the event that unanimity cannot be reached by Joint Commercialization Committee with respect to a matter that is a subject of its decision-making authority, then the matter shall be referred for further review and resolution to the Joint Management Committee. In the event that unanimity cannot be reached by the Joint Management Committee with respect to a matter that is a subject of its decision-making authority, then the matter shall be referred for further review and resolution to the Chief Executive Officer of Organon, or such other similar position designated by Organon from time to time, and the Chief Executive Officer of Lexicon, or such other similar position designated by Lexicon from time to time. The designated officers of each party shall use reasonable efforts to resolve the matter within [**] after the matter is referred to them.

3.7.2.2 If the designated officers cannot resolve any matter pursuant to Section 3.7.2.1 within such [**] period, the matter shall be referred to a Third Party arbitrator or arbitrators, in accordance with the following procedures, whose decision shall be [**]. The parties shall attempt to mutually agree upon a single independent Third Party arbitrator (who shall be a professional with appropriate experience in the subject matter at issue in such disagreement) within [**] after the initial referral of such matter to the designated officers. If the parties are unable to mutually agree upon one such person, then each party shall appoint one independent Third Party professional with appropriate experience in the subject matter at issue in such disagreement prior to the expiration of such [**] period, and within [**] after the initial referral of such matter to the designated officers, such person(s) shall select a single independent Third Party arbitrator, who shall be a professional with appropriate experience in the subject matter at issue in such disagreement. Each party shall present all information presented pursuant to Section 3.7.2.1 and all other information as such party reasonably desires regarding such disagreement. Within [**] after the initial referral of such matter to the designated officers, the arbitrator shall provide written notice to the parties regarding his or her determination regarding such disagreement.

3.8 Administration. The chairperson of each Product Collaboration Committee shall be designated annually on an alternating basis between the parties. The initial chairperson shall be selected by Organon. The party not designating the chairperson shall designate one of its representative members as secretary to such Product Collaboration Committee for such year. The chairperson shall be responsible for calling meetings of such Product Collaboration Committee, sending notices of meetings to all members and for leading such meetings.

3.9 Minutes. Within [**] after each Product Collaboration Committee meeting, the secretary of such Product Collaboration Committee shall prepare and distribute minutes of the meeting, which shall provide a description in reasonable detail of the discussions had at the meeting and a list of any actions, decisions or determinations approved by such Product Collaboration Committee. The secretary shall be responsible for circulation of all draft and final minutes. Draft minutes shall be first circulated to the chairperson, edited by the chairperson and then circulated in final draft form to all members of such Product Collaboration Committee sufficiently in advance of the next meeting to allow adequate review and comment prior to the meeting. Minutes shall be approved or disapproved, and revised as necessary, at the next meeting. Final minutes shall be distributed to the members of such Product Collaboration Committee. 3.10 Expenses. Each party shall be responsible for all travel and related costs for its representatives to attend meetings of, and otherwise participate on, a Product Collaboration Committee.

ARTICLE 4. COMMERCIALIZATION

4.1 General. The parties shall jointly pursue the Commercialization of the Licensed Products in the Collaboration Field under the direction of the Joint Commercialization Committee in accordance with annual Commercialization Plans.

4.2 Commercialization Plans.

4.2.1 The Joint Commercialization Committee shall prepare the first Commercialization Plan for the Licensed Product as soon as commercially reasonable after the Effective Date and annually thereafter.

Each annual Commercialization Plan shall be in 4.2.2 writing and shall set forth with reasonable specificity the Commercialization objectives, priorities, activities, milestones, personnel requirements, other resources and allocations of responsibilities between the parties for the period covered by such annual Commercialization Plan, together with budget(s) for Regulatory and Licensing Costs and Sales and Marketing Costs associated therewith and for the Details to be performed by each party in connection therewith, in a manner consistent with the terms of this Agreement, including, without limitation, the objectives set forth in Section 2.1 and the terms and conditions set forth in Section 2.2. The Commercialization Plans shall cover all aspects of Commercialization relating to the Licensed Product, and shall include, with reasonable specificity, the Commercialization activities to be performed by each party and the Commercialization activities, if any, to be performed by subcontractors.

4.2.3 The Joint Commercialization Committee may agree on modifications, and recommend that the Joint Management Committee approve such modifications, to the provisions of any Commercialization Plan at any time.

4.3 Marketing and Marketing Plans. The Joint Commercialization Committee shall coordinate and implement the marketing and detailing strategies and tactics, sales force training programs, sales forecasts and Post-approval Studies for the Licensed Product for each Contract Year in the Co-Commercialization Territory.

4.4 Sales Force. Except to the extent otherwise agreed by the Joint Management Committee, the parties shall be responsible for providing Details in proportion to their respective Cost and Revenue Sharing Ratio. Notwithstanding the foregoing, if either party elects to provide less than its proportionate share of Details, it may pay the other party to fulfill those obligations at a commercially reasonable rate. For clarity, the measurement of the provision of Details by each party will be made based on the number of Details provided and not based on the Detail Costs.

4.5 Labeling and Promotion. Unless otherwise agreed with a Joint Marketing/Development Collaborator: (a) the Licensed Product will be marketed in each country with one label and will bear one or more Product Trademarks; and (b) all advertising and promotional material in respect of the Licensed Product in each country (including any Licensed Product labeling or packaging inserts to the extent permitted by law or required by any Regulatory Authority and approved by the Joint Commercialization Committee) will include both parties' respective name and address, with the size and placement of each such name and address to be determined by the Joint Commercialization Committee.

Regulatory and Other Inquiries. Upon being contacted by any 4.6 Regulatory Authority for any regulatory purpose pertaining to this Agreement or to the Licensed Product, each party shall promptly notify and consult with the other, and the party that prepared the relevant portion(s) of the regulatory filing to which such contact relates (or, in the event the contact does not relate to a particular portion of a regulatory filing, the owner (or proposed owner) of the relevant Regulatory Approval (or proposed Regulatory Approval), determined in accordance with Section 6.4 of the Collaboration Agreement) shall prepare a response as it deems appropriate, and such party shall (a) have principal responsibility for responding to all inquiries to either party, as the case may be, regarding such portion of such regulatory filing (or, in the event the contact does not relate to a particular portion of a regulatory filing, such Regulatory Approval or proposed Regulatory Approval), (b) give the other party a reasonable opportunity for prior review of and comment on all such responses, (c) consider in good faith the incorporation of such comments and (d) incorporate into such responses those of such comments as are reasonable and not inconsistent with the overall response initially prepared by such party.

4.7 Product Recall. In the event that either party determines that an event, incident or circumstance has occurred that may result in the need for a recall or other removal of the Licensed Product, or any lot or lots thereof, from a market in any country, it shall advise and consult with the other party with respect thereto. The owner of the relevant Regulatory Approval (or proposed Regulatory Approval), determined in accordance with Section 6.4 of the Collaboration Agreement, shall make the final determination to recall or otherwise remove the Licensed Product or any lot or lots thereof from the market. The costs and expenses of such recall or removal in each country, including expenses and other costs or obligations to Third Parties, the cost and expense of notifying customers and costs and expenses associated with shipment of the recalled Licensed Product from a customer to either party shall be included in Regulatory and Licensing Costs.

ARTICLE 5. GRANTS OF RIGHTS

5.1 Grants of Commercialization Licenses.

5.1.1 By Lexicon. Subject to the terms of this Agreement and any applicable [**], Lexicon hereby grants to Organon and its Affiliates, within the Co-Commercialization Territory, a co-exclusive right and license, with the limited right to sublicense (as set forth below), under Lexicon's rights in (a) the Lexicon Background Materials and the Lexicon Background Technology, including, without limitation, any Patent Rights Controlled by Lexicon Covering the foregoing, and (b) the Program Intellectual Property, in each case to make, have made, import, use, have used, offer for sale, sell and have sold the Licensed Product in the Collaboration Field. Such right and license shall include the right to grant sublicenses to Affiliates of Organon and to Third Parties that are approved by the Joint Management Committee.

5.1.2 By Organon. Subject to the terms of this Agreement and any applicable [**], Organon hereby grants to Lexicon and its Affiliates, within the Co-Commercialization Territory, a co-exclusive right and license, with the limited right to sublicense (as set forth below), under Organon's rights in (a) the Organon Background Materials and the Organon Background Technology, including, without limitation, any Patent Rights Controlled by Organon Covering the foregoing, and (b) the Program Intellectual Property, in each case to make, have made, import, use, have used, offer for sale, sell and have sold the Licensed Product in the Collaboration Field. Such right and license shall include the right to grant sublicenses to Affiliates of Lexicon and to Third Parties that are approved by the Joint Management Committee.

5.2 No Grant of Other Technology or Patent Rights. Except as otherwise expressly provided in this Agreement, under no circumstances shall a party hereto, as a result of this Agreement, obtain any ownership interest in or other right to any technology, know-how, patents, patent applications, gene or genomic sequence data or information, products, or biological materials of the other party, including items owned, controlled or developed by, or licensed to, the other party, or transferred by the other party to said party, at any time pursuant to this Agreement.

ARTICLE 6. FINANCIAL TERMS

6.1 Collaboration Cost and Revenue Sharing. Beginning with the Effective Date and throughout the term of this Agreement, all Operating Profit/Loss shall be shared in accordance with the Cost and Revenue Sharing Ratio.

6.2 Reporting of Operating Profit/Loss and Detail Efforts

6.2.1 Reports of Net Sales, Allowable Commercialization Costs and Details. During the term of this Agreement, each party shall, within [**] after each [**], furnish to the other party a written [**] report showing in reasonable detail (a) Net Sales and, [**], gross sales of the Licensed Product sold by such party, its (sub)licensees and their respective Affiliates during such [**] and, [**], the calculation of Net Sales from such gross sales, in each case on a country-by-country basis, (b) the Allowable Commercialization Costs incurred by such party during such [**] in each of the categories specified in Section 1.4, and (c) the Details performed by such party during such [**] in accordance with the applicable Commercialization Plan.

6.2.2 Consolidated Report of Operating Profit/Loss and Detail Efforts. During the term of this Agreement, Lexicon shall, within [**] after each [**], furnish to Organon a written [**] report showing in reasonable detail and in United States Dollars:

> (a) the Net Sales and, [**], gross sales of the Licensed Product sold by the parties and their respective (sub)licensees and Affiliates during such [**] and, [**], the calculation of Net Sales from such gross sales, in each case on a country-by-country basis;

(b) the Allowable Commercialization Costs incurred by Lexicon and Organon, respectively, and in total, during such [**] in each of the categories specified in Section 1.4;

(c) Operating Profit/Loss for such [**], and the amount, if any, owed by one party to the other party in respect thereof; and

(d) the Details performed by Lexicon and Organon, respectively, and in total, during such [**] in accordance with the applicable Commercialization Plan, and the amount, if any, owed by one party to the other party in respect thereof.

6.3 Payment of Operating Profit/Loss Amounts Owed. Amounts shown as owed by one party to the other party under the report delivered in accordance with Section 6.2.2 shall be payable within [**] following the delivery of such report.

6.4 Exchange Rates. Gross sales, Net Sales, Allowable Commercialization Costs and Operating Profit/Loss attributable to countries other than the United States shall be calculated in accordance with the standard exchange rates published in the Wall Street Journal (New York edition) and otherwise in accordance with the standard exchange rate conversion practices used by the reporting party for financial accounting purposes. 6.5 Records. The parties shall each keep accurate books and accounts of record in connection with the Commercialization of the Licensed Product in a manner consistent with GAAP and in sufficient detail to permit accurate determination of all figures necessary for verification of gross sales, Net Sales, Allowable Commercialization Costs, Operating Profit/Loss and Details performed hereunder.

Audits. Upon the written request of a party, the other party 6.6 shall permit an independent certified public accountant selected by the requesting party and acceptable to the other party, which acceptance shall not be unreasonably withheld, to have access, at reasonable times and during normal business hours, to such records of such other party as may be reasonably necessary to verify the accuracy of the reports described herein (and, if not included in such reports, the gross sales of the Licensed Product sold by such party, its (sub)licensees and their respective Affiliates along with the calculation of Net Sales from such gross sales, in each case, on a country-by-country basis), in respect of any fiscal year ending not more than [**] prior to the date of such request. Each party shall use commercially reasonable efforts to schedule all such verifications within [**] after the requesting party makes its written request. All such verifications shall be conducted not more than [**]. The report of the requesting party's independent certified public accountant shall be made available to both parties. Subject to the other party's rights under Section 11.7, in the event requesting party's independent certified public accountant concludes that additional amounts were owed to the requesting party for such period, the additional amounts shall be paid by the other party within [**] of the date the requesting party delivers to the other party such independent certified public accountant's written report so concluding, unless such report contains manifest error. In the event requesting party's independent certified public accountant concludes that there was an overpayment to such party during such period, the overpayment shall be repaid by the requesting party within [**] of the date the requesting party received such independent certified public accountant's written report so concluding, unless such report contains manifest error. The fees charged by such independent certified public accountant shall be paid by the requesting party unless such audit discloses an underpayment or overpayment of more than [**] of the amount due under this Agreement for the period in question, in which case the party responsible for such underpayment or overpayment will bear the full cost of such audit. Each party agrees that all information subject to review under this Section 6.6 or under any agreement with a (sub)licensee of a party is confidential and that the party receiving such information shall cause its independent certified public accountant to retain all such information in confidence. The requesting party's independent certified public accountant shall only report to the requesting party as to the computation of Operating Profit/Loss or royalties payable under this Agreement, as applicable, and shall not disclose to the requesting party any other information of the other party or any (sub)licensee of a Continuing Party.

6.7 Withholding Taxes. In the event that any royalties or other payments due to a party are subject to withholding tax required by law to be paid to the taxing authority of any foreign country, the amount of such tax may be withheld from the applicable royalties or other payment due such party. The party owing such payment shall promptly pay such tax on behalf of the party to which such payment is owed and shall furnish the party to which such payment is owed with a certificate of withholding tax so deducted for such party's avoidance of duplicate taxation in multiple countries. The party owing such payment may not deduct any other withholding or any other governmental charges from the payments agreed upon under this Agreement, except to the extent same are paid on behalf of, or for the benefit of, the party to which such payment is owed. The party owing such payment shall maintain official receipts of payment of any such withholding taxes and shall forward such receipts to the party to which such payment is owed.

6.8 Blocked Currency. If by law, regulation, or fiscal policy of a particular country, conversion into United States dollars or transfer of funds of a convertible currency to the United States is restricted or forbidden, the party owing such payment shall give the party to which such payment is owed prompt written notice and shall make such payment due under this Article 6 through such means or methods as are lawful in such country as the party to which such payment is owed may reasonably designate. Failing the designation by the party to which such payment is owed of such lawful means or methods within [**] after such written notice is given to such party, the party owing such payment shall deposit such royalty payment in local currency to the credit of the party to which such payment is owed in a recognized banking institution designated by such party, or if none is designated by such party within the [**] period described above, in a recognized banking institution selected by the party owing such payment and identified in a written notice to other party, and such deposit shall fulfill all obligations of the party owing such payment to the other party with respect to such payment.

6.9 Interest on Late Payments. A party to which payment is owed under this Agreement shall have the right to seek to collect interest on any payments that are not paid on or before [**] after the date such payments are due under this Agreement at a rate equal to [**], calculated on the total number of days payment is delinquent.

6.10 Manner of Payment. Except as provided in Section 6.8, payments to be made by one party to the other under this Agreement shall be payable in United States dollars and shall be paid by bank wire transfer in immediately available funds to such bank account in the state and country in which such principal office is located as is designated in writing by such party from time to time.

ARTICLE 7. CONFIDENTIALITY

7.1 Nondisclosure Obligations.

7.1.1 General. Except as otherwise provided in this Article 7, during the term of this Agreement and for a period of five (5) years thereafter, each Receiving Party shall maintain the Confidential Information of each Disclosing Party in confidence and use it only for purposes specifically authorized under this Agreement.

Limitations. To the extent it is reasonably necessary 7.1.2 or appropriate to fulfill its obligations or exercise its rights under this Agreement and subject to advance written notification to the Disclosing Party: (a) a party may disclose to Third Parties Confidential Information it is otherwise obligated not to disclose under this Section 7.1, to its Affiliates, (sub)licensees, consultants, outside contractors and clinical investigators, on a strict need-to-know basis for the purposes contemplated by this Agreement and on condition that such entities or persons agree to keep the Confidential Information confidential for the same time periods and to the same extent as such party is required to keep the Confidential Information confidential hereunder; and (b) a party or its (sub)licensees may disclose, using appropriate measures to preserve confidentiality, such Confidential Information to government or other regulatory authorities to the extent that such disclosure is reasonably necessary to obtain authorizations to conduct clinical trials of, and to commercially market, the Licensed Product pursuant to this Agreement. Furthermore, a Receiving Party may request permission from the Disclosing Party to disclose such Confidential Information to the extent that such disclosure is reasonably necessary [**].

7.1.3 Required Disclosure. A Receiving Party may disclose Confidential Information pursuant to interrogatories, requests for information or documents, subpoena, civil investigative demand issued by a court or governmental agency or as otherwise required by law; provided, however, that the Receiving Party shall notify the Disclosing Party promptly upon receipt thereof, giving [**] the Disclosing Party sufficient advance notice to permit it to oppose, limit or seek confidential treatment for such disclosure; and provided, further, that the Receiving Party shall furnish only that portion of the Confidential Information which it is advised by counsel is legally required whether or not a protective order or other similar order is obtained by the Disclosing Party.

7.2 Injunctive Relief. The parties hereto understand and agree that remedies at law may be inadequate to protect against any breach of any of the provisions of this Article 7 by either party or their employees, agents, officers or directors or any other person acting in concert with it or on its behalf. Accordingly, each party shall be entitled to the granting of injunctive relief by a court of competent jurisdiction against any action that constitutes any such breach of this Article 7.

Publication. Organon and Lexicon (each, a "Submitting Party") 7.3 may each publish or present data and results relating to the Licensed Product, subject to the prior review of the proposed disclosure by the other party (each, a "Reviewing Party"), solely to determine (a) whether the proposed disclosure contains the Confidential Information of the Reviewing Party or (b) whether the information contained in the proposed disclosure should be the subject of a patent application to be filed by the Reviewing Party prior to such disclosure. Each Submitting Party shall provide the Reviewing Party with the opportunity to review any proposed abstract, manuscript or presentation by delivering a copy thereof to the Reviewing Party no less than [**] before its intended submission for publication or presentation. The Reviewing Party shall have [**] from its receipt of any such abstract, manuscript or presentation in which to notify the Submitting Party in writing of any specific objections to the disclosure, based on either the need to seek patent protection or concern regarding the specific disclosure of the Confidential Information of the Reviewing Party. In the event the Reviewing Party objects to the disclosure, the Submitting Party agrees not to submit the publication or abstract or make the presentation containing the objected-to information until the Reviewing Party is given a reasonable additional period of time (not to exceed an additional [**]) to seek patent protection for any material in the disclosure which the Reviewing Party believes is patentable (subject, in all events, to Section 7.2) or, in the case of Confidential Information, to allow the Submitting Party to delete any Confidential Information of the Reviewing Party from the proposed disclosure. The Submitting Party agrees to delete from the proposed disclosure any Confidential Information of the Reviewing Party upon request.

ARTICLE 8. REPRESENTATIONS AND WARRANTIES

8.1 Representations, Warranties and Covenants of Lexicon. Lexicon represents and warrants to and covenants with Organon that:

8.1.1 Lexicon is a corporation duly organized, validly existing and in corporate good standing under the laws of the State of Delaware, United States of America;

8.1.2 Lexicon has the corporate and legal right, authority and power to enter into this Agreement, and to extend the rights and licenses granted to Organon in this Agreement;

8.1.3 Lexicon has taken all necessary action to authorize the execution, delivery and performance of this Agreement;

8.1.4 upon the execution and delivery of this Agreement, this Agreement shall constitute a valid and binding obligation of Lexicon, enforceable in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' and contracting parties' rights generally and except as enforceability may be subject to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law);

8.1.5 the performance of Lexicon's obligations under this Agreement will not conflict with its charter documents or result in a breach of any agreements, contracts or other arrangements to which it is a party; and

8.1.6 Lexicon will not during the term of this Agreement enter into any agreements, contracts or other arrangements that would be inconsistent with its obligations under this Agreement.

8.2 Representations, Warranties and Covenants of Organon. Organon represents and warrants to and covenants with Lexicon that:

8.2.1 Organon is a corporation duly organized, validly existing and in corporate good standing under the laws of the Netherlands;

8.2.2 Organon has the corporate and legal right, authority and power to enter into this Agreement, and to extend the rights and licenses granted to Lexicon in this Agreement;

8.2.3 Organon has taken all necessary action to authorize the execution, delivery and performance of this Agreement;

8.2.4 upon the execution and delivery of this Agreement, this Agreement shall constitute a valid and binding obligation of Organon enforceable in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' and contracting parties' rights generally and except as enforceability may be subject to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law);

8.2.5 the performance of its obligations under this Agreement will not conflict with Organon's charter documents or result in a breach of any agreements, contracts or other arrangements to which it is a party; and

8.2.6 Organon will not after the Effective Date enter into any agreements, contracts or other arrangements that would be inconsistent with its obligations under this Agreement.

8.3 Warranty Disclaimer. EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN THIS AGREEMENT, NEITHER PARTY MAKES ANY WARRANTY WITH RESPECT TO ANY PRODUCT, PATENT RIGHTS, GOODS, SERVICES, BACKGROUND MATERIALS OR ANY OTHER SUBJECT MATTER OF THIS AGREEMENT, AND EACH PARTY HEREBY DISCLAIMS WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NON-INFRINGEMENT WITH RESPECT TO ANY AND ALL OF THE FOREGOING. IN ADDITION, THE PARTIES ACKNOWLEDGE THAT THE GENERATION OR USE OF BACKGROUND MATERIALS MAY BE COVERED BY ONE OR MORE VALID PATENTS OF THIRD PARTIES. EACH PARTY ACKNOWLEDGES THAT EXERCISE BY IT OF THE RIGHTS AND LICENSES GRANTED TO IT PURSUANT TO ARTICLE 5 HEREOF MAY BE COVERED BY ONE OR MORE VALID PATENTS OF THIRD PARTIES.

8.4 Limited Liability. EXCEPT AS SPECIFICALLY SET FORTH IN THIS AGREEMENT, NEITHER LEXICON NOR ORGANON WILL BE LIABLE WITH RESPECT TO ANY MATTER ARISING UNDER THIS AGREEMENT UNDER ANY CONTRACT, NEGLIGENCE, STRICT LIABILITY OR OTHER LEGAL OR EQUITABLE THEORY FOR ANY PUNITIVE, EXEMPLARY, INCIDENTAL OR CONSEQUENTIAL DAMAGES OR LOST PROFITS.

9.1 Lexicon Indemnity Obligations. Lexicon agrees to defend, indemnify and hold Organon, its Affiliates and their respective employees and agents harmless from all claims, losses, damages or expenses (including reasonable attorneys' fees and costs of litigation) arising as a result of Lexicon's breach of any of its representations, warranties or covenants hereunder.

9.2 Organon Indemnity Obligations. Organon agrees to defend, indemnify and hold Lexicon, its Affiliates and their respective employees and agents harmless from all claims, losses, damages or expenses (including reasonable attorneys' fees and costs of litigation) arising as a result of Organon's breach of any of its representations, warranties or covenants hereunder.

9.3 Limitation on Indemnity Obligations. Neither party, its Affiliates or their respective employees and agents shall be entitled to the indemnities set forth in Sections 9.1 or 9.2, respectively, to the comparative extent the claim, loss, damage or expense for which indemnification is sought was caused by a grossly negligent, reckless or intentional act or omission by such party, its directors, officers, employees or authorized agents.

Procedure. If a party or any of its Affiliates or their 9.4 respective employees or agents (collectively, the "Indemnitee") intends to claim indemnification under this Article 9, the Indemnitee shall promptly notify the other party (the "Indemnitor") of any loss, claim, damage, liability or action in respect of which the Indemnitee intends to claim such indemnification, and the Indemnitor shall assume the defense thereof with counsel selected by the Indemnitor and reasonably acceptable to the Indemnitee, provided, however, that an Indemnitee shall have the right to retain its own counsel, with the fees and expenses to be paid by the Indemnitee, if representation of such Indemnitee by the counsel retained by the Indemnitor would be inappropriate due to actual or potential differing interests between such Indemnitee and any other party represented by such counsel in such proceedings. The Indemnitor shall have the right to settle or compromise any claims for which it is providing indemnification under this Article 15, provided that the consent of the Indemnitee (which shall not be unreasonably withheld or delayed) shall be required in the event any such settlement or compromise would adversely affect the interests of the Indemnitee. The indemnity agreement in this Article 9 shall not apply to amounts paid in settlement of any loss, claim, damage, liability or action if such settlement is effected without the consent of the Indemnitor. The failure to deliver notice to the Indemnitor within a reasonable time after the commencement of any such action, if prejudicial to the Indemnitor's ability to defend such action, shall relieve such Indemnitor of any liability to the Indemnitee under this Article 9, but the omission so to deliver notice to the Indemnitor will not relieve it of any liability that it may have to any Indemnitee otherwise than under this Article 9. The Indemnitee under this Article 9, its employees and agents, shall cooperate fully with the Indemnitor and its legal representatives in the investigation of any action, claim or liability covered by this indemnification.

9.5 Insurance. Each party shall maintain appropriate product liability insurance with respect to Commercialization of the Licensed Product by such party in such amount as such party customarily maintains with respect to sales of its other products. Each party shall maintain such insurance for so long as it continues to manufacture or sell the Licensed Product, and thereafter for so long as such party customarily maintains insurance with respect to sales of its other products.

ARTICLE 10. EXPIRATION AND TERMINATION

10.1 Term of Agreement. The term of this Agreement shall commence on the Effective Date and shall continue until the earliest to occur of (a) a party Opting Out of the Licensed Product, (b) the parties' grant of Commercialization rights with respect to the Licensed Product to a Joint Marketing/Development Collaborator in all countries within the Co-Commercialization Territory or (c) the agreement of the parties to cease all Development and Commercialization of the Licensed Product.

10.2 Events of Default. An Event of Default by either party shall have occurred upon (a) the occurrence of a material breach of this Agreement if such party fails to remedy such breach within [**] after written notice thereof by the non-breaching party ([**] in the event of a party's failure to make a payment required hereunder) or, if remediation of such breach in [**] is not practicable, if such party fails to commence and diligently pursue such remediation during such [**] period, or (b) the commencement of any proceeding in or for bankruptcy, insolvency, dissolution or winding up by or against such party that is not dismissed or otherwise disposed of within [**] thereafter.

10.3 Effect of an Event of Default. In the event of an Event of Default, the non-defaulting party shall have the right, at its option exercisable in its sole discretion, in addition to any other rights or remedies available to it at law or in equity and subject to the limitations set forth in Sections 3.7.2, 9.4 and 11.7 hereof, to terminate this Agreement upon [**] notice thereof to the other party, in which case (a) the defaulting party shall be deemed to have Opted Out with respect to the Licensed Product and (b) the defaulting party shall return to the non-defaulting party or, upon the non-defaulting party's written instruction, destroy all information, materials or documentation provided by the non-defaulting party pursuant to this Agreement.

10.4 Effect of Expiration or Termination of Agreement. The expiration or termination of this Agreement shall not relieve the parties of any obligation accruing prior to such expiration or termination. The provisions of Articles 7 and 9 and Sections 11.2 and 11.4 through 11.7 hereof shall survive the expiration or termination of this Agreement.

ARTICLE 11. MISCELLANEOUS

11.1 Force Majeure. Neither party shall be held liable or responsible to the other party nor be deemed to have defaulted under or breached this Agreement for failure or delay in fulfilling or performing any obligation under this Agreement when such failure or delay is caused by or results from causes beyond the reasonable control of the affected party, including but not limited to fire, floods, embargoes, war, acts of war (whether war is declared or not), insurrections, riots, civil commotions, strikes, lockouts or other labor disturbances, acts of God or acts, omissions or delays in acting by any governmental authority; provided, however, that the party so affected shall use reasonable commercial efforts to avoid or remove such causes of nonperformance, and shall continue performance hereunder with reasonable dispatch whenever such causes are removed. Either party shall provide the other party with prompt written notice of any delay or failure to perform that occurs by reason of force majeure. The parties shall mutually seek a resolution of the delay or the failure to perform as noted above.

11.2 Assignment. This Agreement may not be assigned or otherwise transferred, in whole or in part, by either party without the consent of the other party; provided, however, that either Lexicon or Organon may, without such consent, assign its rights and obligations under this Agreement (i) to any Affiliate, or (ii) in connection with a merger, consolidation or sale of such portion of a party's assets that includes rights under this Agreement to an unrelated Third Party; provided, further, that such party's rights and obligations under this Agreement to be transferred separate from all or substantially all of its other business assets, including those business assets that are the subject of this Agreement. Any purported assignment in violation of the preceding sentence shall be void. Any permitted assignee shall assume all obligations of its assignor under this Agreement, unless the parties otherwise agree.

11.3.1 Notice of and Reasonable Assurances upon a Change in Control. Within [**] after any Change in Control of a party, the surviving party to such Change in Control (the "Acquiring Party") (a) will notify the other party of the occurrence of such Change in Control and (b) will provide the other party with reasonable assurances of its commitment to fulfill its obligations under this Agreement and the applicable Commercialization Plan(s) then in effect.

11.3.2 Effect of Failure to Provide Assurances upon a Change in Control. If the Acquiring Party fails to provide the assurances contemplated by Section 11.3.1 within the [**] period set forth therein, and does not cure such failure within [**] after notice from the other party, the Acquiring Party shall be deemed to have Opted Out of the Licensed Product, with the same effect as if the Acquiring Party had Opted Out under Section 10.1.1.1 of the Collaboration Agreement; provided that the Acquiring Party shall not [**] with respect to (a) [**] or (b) [**].

11.3.3 Option to Negotiate for Rights to Collaboration Product. Without limiting the foregoing, if requested by the other party at any time within the [**] period following the Acquiring Party's notice under Section 11.3.1 of a Change in Control, the Acquiring Party and the other party shall enter into good faith negotiations, for a period of [**] following such request, for the other party to obtain from the Acquiring Party the exclusive rights to Commercialize the Licensed Product. In the event the Acquiring Party and the other party do not enter into a definitive agreement with respect to such exclusive rights within such [**] period, except to the extent set forth in Section 11.3.4 below, the Acquiring Party and the other party shall each maintain their respective rights with respect to the Commercialization of the Licensed Product as otherwise described in this Agreement.

11.3.4 [**] upon a Change in Control. In the event that (a) the Acquiring Party and the other party enter into negotiations in accordance with Section 11.3.3 with respect to the Licensed Product but do not enter into a definitive agreement within the [**] period contemplated thereby and (b) the Acquiring Party (i) [**] and (ii) has not [**], the other party shall have the right, by delivering written notice thereof within [**] after the expiration of such [**] period, to [**] consisting of (i) [**] and (ii) [**]. [**]at the time such notice is delivered or as soon thereafter as [**] in accordance with the terms of this Section 11.3.4. Unless otherwise agreed by the Acquiring Party and the other party, [**], shall be determined by binding arbitration in accordance with the following provisions. The parties shall attempt to mutually agree upon a single independent Third Party arbitrator (who shall be a professional with appropriate experience in [**]) within [**] after the other party's delivery of the written notice in accordance with this Section 11.3.4 of [**]. If the parties are unable to mutually agree upon one such person, then each party shall appoint one independent Third Party professional with appropriate experience in [**] prior to the expiration of such [**] period, and within [**] after the initial notice, such person(s) shall select a single independent Third Party arbitrator, who shall be a professional with appropriate experience in [**]. Each party shall present all information as such party reasonably desires regarding [**]. Within [**] after the initial notice, the arbitrator shall provide written notice to the parties regarding his or her determination regarding [**].

11.4 Severability. Each party hereby agrees that it does not intend to violate any public policy, statutory or common laws, rules, regulations, treaty or decision of any government agency or executive body thereof of any country or community or association of countries. Should one or more provisions of this Agreement be or become invalid, the parties hereto shall substitute, by mutual consent, valid provisions for such invalid provisions which valid provisions in their economic effect are sufficiently similar to the invalid provisions that it can be reasonably assumed that the parties would have entered into this Agreement with such valid provisions in lieu of such invalid provisions. In case such valid provisions cannot be agreed upon, the invalidity of one or several provisions of this Agreement shall not affect the validity of this Agreement as a whole, unless the invalid provisions are of such essential importance to this Agreement that it is to be reasonably assumed that the parties would not have entered into this Agreement without the invalid provisions.

11.5 Notices. Any consent, notice or report required or permitted to be given or made under this Agreement by one of the notification parties hereto to the other shall be in writing, delivered personally or by facsimile (and promptly confirmed by telephone, personal delivery or courier) or courier, postage prepaid (where applicable), addressed to such other party at its address indicated below, or to such other address as the addressee shall have last furnished in writing to the addressor and shall be effective upon receipt by the addressee.

If to Lexicon:	Lexicon Genetics Incorporated 8800 Technology Forest Place The Woodlands, Texas 77381 Attention: President and Chief Executive Officer Telephone: (281) 863-3000 Facsimile: (281) 863-8095
With copies to:	Lexicon Genetics Incorporated 8800 Technology Forest Place The Woodlands, Texas 77381 Attention: Chief Financial Officer Telephone: (281) 863-3000 Facsimile: (281) 863-8095
	Lexicon Genetics Incorporated 8800 Technology Forest Place The Woodlands, Texas 77381 Attention: General Counsel Telephone: (281) 863-3000 Facsimile: (281) 863-8010
If to Organon:	N.V. Organon Kloosterstraat 6 5349 AB Oss The Netherlands Attention: General Manager Telephone: 31 412 661 397 Facsimile: 31 412 663 529
With a copy to:	Director Legal Affairs Akzo Nobel Pharma B.V. Wethouder van Eschstraat 1 P.O. Box 20 5340 BH Oss The Netherlands

All such communications shall be effective upon receipt.

11.6 Applicable Law. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, United States of America, without reference to the conflicts of law principles thereof.

11.7 Dispute Resolution. Subject to Section 3.7.2, the parties hereby agree that they will first attempt in good faith to resolve any controversy or claim arising out of or relating to this Agreement promptly by negotiations. If a controversy or claim should arise hereunder, the matter shall be referred to an individual designated by the Chief Executive Officer (or the equivalent position) of Lexicon and by the Chief Executive Officer (or the equivalent position) of Organon (the "Representatives"). If the matter has not been resolved within [**] of the first meeting of the Representatives of the parties (which period may be extended by mutual agreement) concerning such matter, the parties shall be free to pursue all available recourse both at law and in equity.

11.8 Entire Agreement. This Agreement, together with the exhibits and appendices hereto and any confidentiality agreement(s) executed in contemplation of this Agreement, contains the entire understanding of the parties with respect to the subject matter hereof. All express or implied agreements and understandings, either oral or written, heretofore made are expressly merged in and made a part of this Agreement. This Agreement may be amended, or any term hereof modified, only by a written instrument duly executed by both parties hereto.

Publicity. Lexicon and Organon each agree not to disclose any 11.9 terms or conditions of this Agreement to any Third Party without consulting the other party prior to such disclosure. Notwithstanding the foregoing, prior to execution of this Agreement, Lexicon and Organon shall agree upon the substance of information that can be used as a routine reference in the usual course of business to describe the existence and general nature of this transaction, and Lexicon and Organon may disclose such information without consulting the other party. The parties may thereafter from time to time mutually agree on revisions to material to be used as a routine reference, which revisions shall be submitted by one party for the review and approval of the other party at least [**] prior to the anticipated use or disclosure of the revised material, such approval not to be unreasonably withheld. The terms of this Agreement shall be treated as the Confidential Information of Lexicon and Organon, and, except to the extent required by applicable law, shall not be disclosed to anyone (except for the parties' respective employees, consultants, agents and attorneys assisting in the review and negotiation of this Agreement who have a need to know the terms of this Agreement) without the written permission of Organon or Lexicon. If either party desires to release a separate announcement relating to this Agreement, it shall first allow the other party to approve in writing such proposed announcement; provided that such approval shall not be unreasonably withheld or delayed.

11.10 Headings. The captions to the several Articles and Sections hereof are not a part of this Agreement, but are merely guides or labels to assist in locating and reading the several Articles and Sections hereof.

11.11 No Partnership. It is expressly agreed that the relationship between Lexicon and Organon shall not constitute a partnership, joint venture or agency. Neither Lexicon nor Organon shall have the authority to make any statements, representations or commitments of any kind, or to take any action, which shall be binding on the other, without the prior consent of the other party to do so.

11.12 Exports. The parties acknowledge that the export of technical data, materials or products is subject to the exporting party receiving any necessary export licenses and that the parties cannot be responsible for any delays attributable to export controls which are beyond the reasonable control of either party. Lexicon and Organon agree not to export or re-export, directly or indirectly, any information, technical data, the direct product of such data, samples or equipment received or generated under this Agreement in violation of any applicable export control laws or governmental regulations. Lexicon and Organon agree to obtain similar covenants from their licensees, (sub)licensees, or corporate partners, as the case may be, and contractors with respect to the subject matter of this Section 11.12.

11.13 Waiver. The waiver by either party hereto of any right hereunder or the failure to perform or of a breach by the other party shall not be deemed a waiver of any other right hereunder or of any other breach or failure by said other party whether of a similar nature or otherwise.

11.14 Counterparts. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

IN WITNESS WHEREOF, the parties have ca officers to execute and deliver this Agreement a					
LEXICON GENETICS INCORPORATED					
Ву:	Date:				
Name:					
Title:					
N.V. ORGANON					
By:	Date:				
Name:					
Title:					
Ву:	Date:				
Name:					
Title:					

EXHIBIT 1.40

ALLOCATION OF NET SALES IN BUNDLED TRANSACTION

With respect to Licensed Product sold in a Bundled Transaction in which Organon, Lexicon or any of their respective Affiliates or (sub)licensees discounts the sales price of the Licensed Product to a greater degree than such seller generally discounts the price of its other products to such customer, the amount to be included in Net Sales of the Licensed Product shall be calculated in accordance with the following formula:

NS-P =	ASP-P X N-P	x BTF
-	(SIGMA)=1 ASP-pi x N-pi	

Where:

	NS-P	=	Amount allocated to Net Sales of the Licensed Product
	ASP-P	=	Average Selling Price (as defined below) per unit, during the applicable period, of the Licensed Product when sold alone
	ASP-pi	=	Average Selling Price per unit, during the applicable period, of the Licensed Product or each product other than a Licensed Product in the Bundled Transaction when sold alone
	N - P	=	Total number of units of Licensed Product included in the Bundled Transaction during the applicable period
	N-pi	=	Total number of units (i.e., corresponding to the same ASP-pi) of the Licensed Product or product other than a Licensed Product included in the Bundled Transaction during the applicable period
	(SIGMA)=1	=	The sum of the products of the formula ASP-pi ? N-pi for each and every Licensed Product or product other than a Licensed Product included in the Bundled Transaction during the applicable period
	BTF	=	The aggregate amounts paid to the seller for the Bundled Transaction during the applicable period
r	age Selling	Price	shall be based on the actual average selling

The Average Selling Price shall be based on the actual average selling price of the Licensed Product or product other than a Licensed Product, as the case may be, determined for the applicable period.

If the Licensed Product or other product is not sold separately and no bona fide list price exists for the Licensed Product or other product, the parties shall agree upon an imputed bona fide list price for the Licensed Product or other product, and Net Sales with respect thereto shall be based on such imputed list price. I, Arthur T. Sands, certify that:

- I have reviewed this Quarterly Report on Form 10-Q of Lexicon Genetics Incorporated;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: July 29, 2005

/s/ Arthur T. Sands

Arthur T. Sands, M.D., Ph.D. President and Chief Executive Officer I, Julia P. Gregory, certify that:

- I have reviewed this Quarterly Report on Form 10-Q of Lexicon Genetics Incorporated;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: July 29, 2005

/s/ Julia P. Gregory

Julia P. Gregory

Executive Vice President, Corporate Development and Chief Financial Officer

CERTIFICATION

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (18 U.S.C. 1350, as adopted), Arthur T. Sands, M.D., Ph.D., Chief Executive Officer of Lexicon Genetics Incorporated ("Lexicon"), and Julia P. Gregory, Chief Financial Officer of Lexicon, each hereby certify that:

- Lexicon's Quarterly Report on Form 10-Q for the period ended June 30, 2005, and to which this Certification is attached as Exhibit 32.1 (the "Periodic Report"), fully complies with the requirements of section 13(a) or section 15(d) of the Securities Exchange Act of 1934, and
- 2. The information contained in the Periodic Report fairly presents, in all material respects, the financial condition and results of operations of Lexicon.

IN WITNESS WHEREOF, the undersigned have set their hands hereto as of the 29th day of July, 2005.

By: /s/ Arthur T. Sands Arthur T. Sands, M.D., Ph.D. President and Chief Executive Officer

By: /s/ Julia P. Gregory Julia P. Gregory Executive Vice President, Corporate Development and

Chief Financial Officer